## BERGISCHE UNIVERSITÄT WUPPERTAL

DOCTORAL THESIS

## General Extrapolation Spaces and Perturbations of Bi-Continuous Semigroups

Author: Christian BUDDE Supervisor: Prof. Dr. Bálint FARKAS

Second Reader: Prof. Dr. Abdelaziz Rhandi

A thesis submitted in fulfillment of the requirements for the degree of Doctor of Science (Dr. rer. nat.)

in the

Workgroup Functional Analysis School of Mathematics and Natural Sciences The PhD thesis can be quoted as follows:

urn:nbn:de:hbz:468-20191205-110300-4 [http://nbn-resolving.de/urn/resolver.pl?urn=urn%3Anbn%3Ade%3Ahbz%3A468-20191205-110300-4]

DOI: 10.25926/7ss7-bt33 [https://doi.org/10.25926/7ss7-bt33]

Date of oral examination: Dean: Members of the examining board:

19. September 2019
Prof.Dr. Michael Günther
Prof.Dr. Bálint Farkas
Prof.Dr. Abdelaziz Rhandi
Prof.Dr. Birgit Jacob
Prof.Dr. Barbara Rüdiger-Mastandrea

Dedicated to my family and the love of my life

## Acknowledgements

First and foremost I would like to express my sincere gratitude to my advisor Prof.Dr. Bálint Farkas for the continuous support of my Ph.D study and related research, for his patience, motivation, and immense knowledge. It has been an honor to be his Ph.D student. His guidance helped me in all the time of research and writing of this thesis. I could not have imagined having a better advisor and mentor for my Ph.D study. The joy and enthusiasm he has for his research was contagious and motivational for me. I am also thankful for the excellent example he has provided as a successful mathematician and professor.

My sincere thanks also goes to Prof.Dr. Abdelaziz Rhandi, Dr. Cristian Tacelli, Prof.Dr. Marjeta Kramar Fijavž, Dr. Retha Heyman and Dr. Ronalda Benjamin who provided me an opportunity to join their team as intern in Salerno (Italy), Ljubljana (Slovenia) and Stellenbosch (South Africa). Without their precious support it would not be possible to reach my research goals. Especially, I want to thank the *Team Outgoing* of the International Office of the University Wuppertal which supported my research visits with the ERASMUS+ program.

I am indebted to Prof.Dr. Moritz Weber und Viktoria Kühner for the additional proofreading of this thesis.

I would like to thank my family for supporting me spiritually throughout writing this thesis and my life in general. Last but not least I would like to thank Lena, who was always there to help me.

Christian Budde Bergische Universität Wuppertal July 2019

## Contents

A	Acknowledgements V															
In	Introduction															
1	Bi-Continuous Semigroups															
	1.1	Bi-admissible spaces														
		1.1.1 The space $C_b(\Omega)$ of bounded continuous functions	2													
		1.1.2 The dual space $X'$	2													
	1.2	Bi-continuous semigroups	3													
		1.2.1 The left-translation semigroup	4													
		1.2.2 The adjoint semigroup	4													
	1.3		5													
		1.3.1 The left-translation semigroup on $C_b(\mathbb{R})$	6													
		1.3.2 The adjoint semigroup	7													
	1.4	The space of strong continuity	8													
		1.4.1 The left-translation semigroup on $C_b(\mathbb{R})$	8													
		1.4.2 The adjoint semigroup	9													
	1.5	Well-posedness	9													
	1.6	Adjoint bi-continuous semigroups	10													
	1.7	Approximation theorems	11													
	1.8	Notes	12													
<b>2</b>	Intermediate and Extrapolated Spaces															
	2.1	Sobolev and extrapolation spaces for invertible operators	17													
		2.1.1 Abstract Sobolev spaces	17													
		2.1.2 Extrapolation spaces	20													
	2.2	Intermediate spaces for operators with rays of minimal growth	25													
	2.3	Intermediate and extrapolation spaces for semigroup generators	28													
		2.3.1 Intermezzo on real interpolation spaces	31													
	2.4	Intermediate and extrapolation spaces for bi-continuous semigroups	32													
		2.4.1 The Hille–Yosida Theorem	32													
		2.4.2 Extrapolated semigroups	34													
		2.4.3 Hölder spaces of bi-continuous semigroups	36													
		2.4.4 Representation of Hölder spaces by generators	41													
	2.5	Examples	43													
		2.5.1 The left-translation semigroup	43													

		2.5.2 The multiplication semigroup	:5
		2.5.3 The Gauss–Weierstrass semigroup	6
	2.6	Notes	7
3	$\mathbf{Fib}$	erwise Multiplication 4	9
	3.1	$L^p$ -fiber spaces	0
	3.2	Unbounded operator-valued multiplication operators	<b>1</b>
	3.3	Extrapolation of unbounded multiplication operators	3
	3.4	Notes	6
4	Des	ch–Schappacher Perturbations 5	7
	4.1	An abstract Desch–Schappacher perturbation result 5	9
		4.1.1 Abstract Favard Spaces and comparison of semigroups	<b>52</b>
	4.2	Admissible operators	53
	4.3	Perturbations of the translation semigroup 6	6
	44	Notes	'1
	1.1		1
<b>5</b>	Pos	tive Mivadera–Voigt perturbations 7	3
	5.1	Preliminaries 7	'4
	0.1	5.1.1 Positivity and bi-AL-spaces 7	·4
		5.1.2 Bounded perturbations	 '8
		5.1.2 Dounded perturbations	0 ,0
	F 9	The Dreef	0 )1
	0.Z	Ine Proof         8           D         1	1. . 4
	5.3	Examples	•4
		5.3.1 Rank-one perturbations	•4
		5.3.2 Gauss–Weierstrass semigroup on $M(\mathbb{R})$	55
	5.4	Notes	;9
c	т		-
0	Imp	Plemented Semigroups 9	T
	0.1	Preliminaries	12
		6.1.1 The space of bounded linear operators $\mathscr{L}(E)$	2
		6.1.2 The implemented semigroup	3
	6.2	Intermediate and extrapolation spaces	14
		6.2.1 Favard- and Hölder spaces	)4
		$6.2.2  \text{Extrapolation}  \dots  \dots  \dots  \dots  \dots  \dots  \dots  \dots  \dots  $	15
	6.3	Perturbations	6
		6.3.1 Ideals in $\mathscr{L}(E)$ and module homomorphisms $\ldots \ldots \ldots \ldots \ldots $ 9	6
		6.3.2 A one-to-one correspondence	18
		6.3.3 Comparisons	0
	6.4	Notes	)2
	-		
<b>7</b>	Flo	vs on Networks 10	3
	7.1	Preliminaries	14
	7.2	Transport problems on (in)finite metric graphs	15
	_	7.2.1 The simple case: constant velocities	)6
		7.2.2 The rationally dependent case	1
		7.2.3 The general case for finite networks 11	- ?
			. 4

	7.3	Notes	 • •	•••	 •••	•••	 	 	•	 	 •	 	 	 	114
Ε	Bibliog	graphy													115

## Introduction

A lot of well-known partial differential equations modeling physical systems, such as the heat equation, the Schrödinger equation or the wave equation, use temporal change of states. *Evolution equation* is an umbrella term for such equations that can be interpreted as differential laws describing the development of a system or as a mathematical treatment of motion in time. In 1921 Albert Einstein said about physical models:

As far as the laws of mathematics refer to reality, they are not certain; and as far as they are certain, they do not refer to reality.

Certainly, this quote encourages a discussion on the benefit of evolution equations in connection with the applications to physics, i.e., it exacerbates the discussion on the connection between mathematics and reality. However, one can say that the science provides models and we process and work them out. Hence, we are, as it seems, not responsible for the relation to reality. Of course, this is a short and crisp consideration of this topic, which normally requires much more discussion. This is for example done by G. Nickel [52, pp. 531-554]. Since this topic is beyond the goals of this thesis, we now leave this philosophic area and turn to mathematics. To quote Henri Poincaré:

Mathematics has a threefold purpose. It must provide an instrument for the study of nature. But this is not all: it has a philosophical purpose, and, I daresay, an aesthetic purpose.

With a solution of an evolution equation, one can predict the future of the corresponding physical system which makes it deterministic. One can find various books on the theory of evolution equations. At the same time, there are also monographs consisting only of mathematical applications of evolution equations in physics and life sciences, cf. [84], which in fact emphasizes the strength of the theory. Evolution equations can be treated by an operator theoretical approach. They can be rewritten as so-called abstract Cauchy problems. We illustrate this by an example. Let  $\Omega \subseteq \mathbb{R}^n$  be a bounded domain with smooth boundary  $\partial \Omega$ . Let  $\Delta := \sum_{i=1}^{n} \frac{\partial^2}{\partial x_i^2}$  denote the Laplacian and consider the following problem

$$\begin{cases} \frac{\partial}{\partial t}w(t,x) = \Delta w(t,x), & (t,x) \in [0,\infty) \times \Omega, \\ w(0,x) = f(x), & x \in \Omega, \\ w(t,x) = 0, & x \in \partial\Omega, \ t \ge 0, \end{cases}$$
(PDE)

where  $f \in L^2(\Omega)$  is given. Now consider the Banach space  $X := L^2(\Omega)$  and define  $u(t) := w(t, \cdot)$  to be a function with variable x. Furthermore, define a linear operator by

$$Au := \Delta u, \quad D(A) := \mathrm{H}_0^2(\Omega)$$

where  $H_0^2(\Omega)$  denotes the Sobolev space of functions with zero trace on the boundary, which is in fact the closure of  $C_0^{\infty}(\Omega)$  with respect to the Sobolev norm. Observe that the boundary conditions in (PDE) are now incorporated into the domain of the operator, so that (PDE) can be rewritten as

$$\begin{cases} \dot{u}(t) = Au(t), & t \ge 0, \\ u(0) = x \in X, \end{cases}$$
(ACP)

which is a Banach space valued initial value problem, also called an *abstract Cauchy problem*. The task is to analyse whether such an abstract Cauchy problem has a solution. Generally speaking, given an unbounded operator (A, D(A)) on a Banach space X, a (classical) solution of the corresponding abstract Cauchy problem (ACP) is by definition a function  $u : \mathbb{R}_{\geq 0} \to X$  such that u is continuously differentiable,  $u(t) \in D(A)$  for all  $t \geq 0$  and (ACP) is satisfied. In fact, this leads to the generic term *well-posedness*, including the existence of a unique solution of (ACP). By definition the equation is well-posed if D(A) is dense in X, for every  $x \in D(A)$  there exists a unique solution  $u(\cdot, x)$  of (ACP) and for every sequence  $(x_n)_{n \in \mathbb{N}}$  in D(A) with  $\lim_{n\to\infty} x_n = 0$  one has  $\lim_{n\to\infty} u(t, x_n) = 0$  uniformly for t on compact intervals  $[0, t_0]$  for each  $t_0 > 0$ .

Now operator semigroups, the generalization of the exponential function, come into the picture. By definition, a family  $(T(t))_{t\geq 0}$  of bounded linear operators on a Banach space is called a *strongly continuous one-parameter semigroup*, or  $C_0$ -semigroups for short, if T(0) = I, T(t+s) = T(t)T(s) for all  $s, t \geq 0$  and  $||T(t)x - x|| \to 0$  as  $t \to 0$  for each  $x \in X$ . To each  $C_0$ -semigroup  $(T(t))_{t\geq 0}$  we can assign a linear operator (A, D(A)) by setting

$$Ax := \lim_{t \to 0} \frac{T(t)x - x}{t}, \quad D(A) := \left\{ x \in X : \lim_{t \to 0} \frac{T(t)x - x}{t} \text{ exists} \right\},$$

which is in some sense the derivative of  $(T(t))_{t\geq 0}$  in t = 0. This operator, which enjoys nice properties like closedness or having a dense domain, is called the generator of  $(T(t))_{t\geq 0}$ . The converse question, which linear operators (A, D(A)) are generators of a  $C_0$ -semigroup, is answered by the *Hille-Yosida theorem*, cf. [52, Chapter II, Sect. 3, Thm. 3.8].

Solutions of (ACP) and semigroups are strongly connected. In particular, for a closed operator (A, D(A)) the problem (ACP) is well-posed if and only if A generates a  $C_0$ -semigroup. In this case for every  $x \in D(A)$  the function  $u : \mathbb{R}_{\geq 0} \to X$  defined by u(t) := T(t)x is the unique classical solution.

Stochastic differential equations, Ornstein–Uhlenbeck processes or Feller processes, give rise to transition semigroups which are in general not strongly continuous. An example of such a semigroup is the one coming from the differential operator on  $C_b(\mathbb{R}^n)$ ,  $n \in \mathbb{N}$ , given by

$$\mathcal{A}u(x) := \sum_{i,j=1}^{n} q_{ij}(x) \frac{\partial^2}{\partial x_i \partial x_j} u(x) + \sum_{i=1}^{n} b_i(x) \frac{\partial}{\partial x_i} u(x),$$

on the domain

$$D(\mathcal{A}) := \left\{ u \in \bigcap_{1$$

where  $q_{ij}$  and  $b_i$  are sufficiently regular functions. In which way stochastic differential equa-

tions are associated with this differential operator is discussed in [82, Sect. 2.5]. Now consider the corresponding parabolic problem

$$\begin{cases} \frac{\partial}{\partial t} w(t, x) = \mathcal{A}w(t, x), & t \ge 0, \\ w(0, x) = f(x) \in \mathcal{C}_{\mathbf{b}}(\mathbb{R}^n). \end{cases}$$

By [82, Thm. 2.2.1], for every  $f \in C_b(\mathbb{R}^n)$  there exists a solution  $u \in C(\mathbb{R}_{\geq 0} \times \mathbb{R}^n) \cap C^{1+\alpha/2,2+\alpha}_{loc}(\mathbb{R}_{\geq 0} \times \mathbb{R}^n)$  of this problem. Furthermore, by [82, Thm. 2.2.5] this solution can be represented by a semigroup  $(T(t))_{t\geq 0}$  on  $C_b(\mathbb{R}^n)$ , i.e.,  $u(t,x) = (T(t)f)(x), t \geq 0, x \in \mathbb{R}^n$ . However, by [82, Thm. 9.2.6] this semigroup of bounded linear operators fails to be strongly continuous on  $C_b(\mathbb{R}^n)$  in general. Here bi-continuous semigroups come into play, which form in fact the key subject of this thesis. Let us consider our topics and results in more mathematical detail and introduce the main themes and explain the structure of this thesis.

The research on bi-continuous semigroups was motivated by the work of F. Kühnemund. In fact, she was the initiator for the development of the theory of bi-continuous semigroups, cf. [78, 79]. The main idea is to equip the Banach space X, on which the semigroup  $(T(t))_{t>0}$  fails to be strongly continuous with respect to the norm, with an additional locally convex topology  $\tau$ , which is compatible with the norm topology, such that the semigroup becomes strongly continuous with respect to  $\tau$ . We will investigate some explicit examples of Banach spaces and bi-continuous semigroups in Chapter 1. Moreover, we will recall in Section 1.3 that, as in the case of  $C_0$ -semigroups, one can assign a generator to each bi-continuous semigroup which has properties similar to the generators of strongly continuous semigroups. As might have been expected, the semigroup  $(T(t))_{t>0}$ , corresponding to the differential operator from above, yields, under an additional spectral assumption, a generator which coincides with  $(\mathcal{A}, D(\mathcal{A}))$ , cf. [82, Prop. 2.3.6]. Likewise, we also discuss abstract Cauchy problems for bicontinuous semigroups. Especially, there is also a notion of well-posedness which is related to the generator as well, see Section 1.5. Moreover, F. Kühnemund proved a Hille-Yosida generation type theorem [79, Thm. 16] which we will address from a new point of view in Section 2.4.1, cf. Theorem 2.38.

The work of F. Kühnemund was followed by research by B. Farkas. He investigated perturbation theory for bi-continuous semigroups. The general idea for perturbations of semigroups is the following: think of an explicit partial differential equation and the corresponding abstract Cauchy problem. In order to show that our problem has a unique solution we want to apply the Hille–Yosida generation theorem. Verifying the conditions of this theorem can be really involved in specific situations. The idea is to split the operator of the given abstract Cauchy problem into a sum of simpler operators. Quite often, it is the case that one of theses operators generates a semigroup. In the abstract one can formulate the question for bi-continuous semigroups as follows: consider a generator (A, D(A)) of a bi-continuous semigroup and (B, D(B)) a second operator. The task is to find conditions on the operator (B, D(B)) such that the following abstract Cauchy problem

$$\begin{cases} \dot{u}(t) = Au(t) + Bu(t), \quad t \ge 0, \\ u(0) = x \in X. \end{cases}$$

is well-posed, i.e., the sum A + B together with an appropriate domain generates a semigroup.

There is no universal theory concerning the operator (B, D(B)) to achieve this. Therefore, one regards several classes of operators. In case that B is a bounded operator, we talk about bounded perturbations. If the domain D(B) coincides with D(A) we consider the so-called Miyadera–Voigt perturbations. Both perturbation types for bi-continuous semigroups were treated by Farkas [56], [55] and [54, Chapter 3]. We will recall the corresponding results in Section 5.1.2 and Section 5.1.3 and take positivity of the Miyadera–Voigt perturbations in Chapter 5 into account. The corresponding perturbation result in this thesis is Theorem 5.19. As remarked by B. Farkas in [54, Sect. 3.2], a third type of semigroup perturbation, the Desch–Schappacher perturbation, was left open. In Chapter 4 we discuss this type of perturbation in detail which culminate in Theorem 4.4 and Theorem 4.7. As a matter of fact, the idea of this perturbation type is to enlarge the Banach space by a technique called extrapolation. In the case of a strongly continuous semigroups  $(T(t))_{t>0}$  with generator (A, D(A)) one introduces a new norm  $\|\cdot\|_{-1}$  on X by  $\|x\|_{-1} := \|A^{-1}x\|, x \in X$  (observe that we may assume without loss of generality that (A, D(A)) is invertible). The completion of X with respect to  $\|\cdot\|_{-1}$  is then denoted by  $X_{-1}$  and is called the first extrapolation space. In general, the resulting spaces are abstract objects which can not easily be identified as wellknown spaces. Some special examples where this is possible are discussed by Nagel, Nickel and Romanelli [93, Sect. 2] or Engel and Nagel [52, Chapter II, Sect. 5(a)]. For non-strongly continuous semigroups, and especially for bi-continuous semigroups one has to take a closer look. As a matter of fact, we discuss these spaces in detail in Chapter 2 and identify some of these spaces for the specific examples as well-known function spaces, cf. Section 2.5. The semigroup  $(T(t))_{t>0}$  can be continuously extended to a semigroup on the extrapolation space. The corresponding generator is denoted by  $(A_{-1}, D(A_{-1}))$ . One combines this extrapolation procedure with perturbations by considering operators  $B: X \to X_{-1}$  that are admissible in the sense that  $(A_{-1} + B)|_X$ , with an appropriate domain, is the generator of a semigroup.

Having the abstract theory of bi-continuous semigroups at hand, one can consider applications. In particular, we consider dynamical processes on graphs. In the modern time of the world wide web, everyone and everything is connected electronically or socially. Nowadays the transport of information, goods or passengers is of great interest if these are carried between a large number of customers. These transport processes between the consumers are described by the transport equation. The connections can be modelled by graphs. A topological structure turns these graphs into networks. Precursors in the operator theoretical approach of dynamics on networks are M. Kramar Fijavž and E. Sikolya and their coauthors [76, 77, 50, 43] as well as B. Dorn [41, 42] at a later date, who even treated flows on infinite networks. Briefly, they considered the transport equation on each edge together with Kirchhoff law boundary conditions in the vertices. From this one obtains an abstract Cauchy problem. For the simple case where the velocities on the edges are the same and equal to 1, one can even give an explicit expression for the  $C_0$ -semigroup solving the abstract Cauchy problem, cf. [41, Prop. 3.3]. In particular, one can even recover well-posedness, spectral and asymptotic properties. To do so, one consideres  $L^{1}([0,1],\mathbb{C}^{m})$  or  $L^{1}([0,1],\ell^{1})$  as the state space for the dynamics on the edges of the network. Together with M. Kramar Fijavž we regarded the state space  $L^{\infty}([0,1],\ell^1)$ . Here adjoint semigroups and hence bi-continuous semigroups play an important role. By Theorem 7.9 and Corollary 7.16 we show that the transport problem is well-posed on our new state space  $L^{\infty}([0,1],\ell^1)$  in the simple case where all velocities are equal as well as for the general case.

#### Overview of this thesis

In **Chapter 1** we introduce the concept of bi-continuous semigroups and the underlying structure. The Banach spaces we are looking at are all equipped with a locally convex topology which interacts with the norm topology in a certain way. Typical examples are the compactopen topology on the space of bounded continuous functions and the weak\*-topology on the dual of a Banach space. Beside that we give illustrations of bi-continuous semigroups on these spaces, for instance, the left-translation semigroup and the adjoint semigroup. Another interesting example, the implemented semigroup on the space of bounded linear operators, is reviewed in Chapter 6. These semigroups belong to the guideline of this chapter, and of this thesis in general. In fact, we determine their generators and describe the corresponding space of strong continuity. The last two points are only treated for two of the major examples, because the implemented semigroup needs more attention, hence the discussion of their properties will take place later in Chapter 6.

**Chapter 2** consists of the construction and study of extrapolation- and intermediate spaces. For  $C_0$ -semigroups these spaces are well-known and can be applied for example to maximal regularity problems. However, the case of bi-continuous semigroups, or more generally nondensely defined operators, is new. One only has to assume some density condition which is in most of the examples, and especially for bi-continuous semigroups, fulfilled. At the end of Section 2.1.2 we give a universal construction for extrapolation spaces which allows us to determine these spaces explicitly. Additionally, in Section 2.4.1 we give a direct proof of the Hille–Yosida generation theorem for bi-continuous semigroups by utilizing these new techniques. This chapter is the starting point for various applications as they come up in Chapter 3, Chapter 4 and Chapter 6.

In Chapter 3 we consider an explicit application of extrapolation for unbounded operatorvalued multiplication operators on Bochner  $L^p$ -spaces. In particular, we show that this extrapolation procedure gives rise to the study of fiber  $L^p$ -spaces as they are introduced by R. Heymann [65]. In the same manner, T. Graser studied multiplication operators and their extrapolation spaces on the space  $C_0(\mathbb{R}, X)$  in [59] and obtained comparable results. This topic is ongoing research with R. Heymann.

Our main objective in **Chapter 4** is a Desch–Schappacher type perturbation result for bicontinuous semigroups. For this we use the theory from Chapter 2. In this context we consider operators  $B \in \mathscr{L}(X, X_{-1})$  such that the map  $B : X \to X_{-1}$  is continuous with respect to the corresponding locally convex topologies as described in Section 2.4.2. We show that under certain conditions the operator  $(A_{-1}+B)_{|X}$  generates again a bi-continuous semigroup on the Banach space X. In addition, we combine these perturbation results with the intermediate spaces from Chapter 2.

We carry on perturbation theory for bi-continuous semigroups in **Chapter 5**. In particular, we consider Miyadera–Voigt perturbations as they were already treated by Farkas [54, Sect. 3.2] and [55], but with the extra assumption of positivity. The corresponding result for strongly continuous semigroups is due to J. Voigt, cf. [119]. To proceed we consider an order on the underlying Banach space and say that a semigroup is positive if and only if the semigroup operators maps positive elements to positive elements. One observes from [54, Thm. 1.4.2] that positivity of a semigroup is equivalent to the fact that the resolvent operator is also positive. At the end of this chapter, we prove that every unbounded positive rank-one perturbation gives rise to a positive Miyadera–Voigt perturbation. As a second example we consider a semigroup on the space of bounded Borel measures on the real line. As a matter of fact, this specific semigroup is the adjoint of the Gauss–Weierstrass semigroup, also known as diffusion semigroup or heat semigroup.

As already mentioned above, **Chapter 6** is devoted to implemented semigroups which are bi-continuous with respect to the strong operator topology on the space of bounded linear operators. In particular, we discuss generators and the space of strong continuity for such semigroups. At the end of this chapter we consider also Desch–Schappacher perturbations for this class of semigroups. We show that there is a one-to-one correspondence between the implemented semigroups and the underlying strongly continuous semigroups. For this connection we study ideals in the space of bounded linear operators and corresponding module homomorphisms.

Last but not least we consider flows on (infinite) networks in **Chapter 7**. For that we first recall some notations from graph theory, and go on with rephrasing the corresponding partial differential equations of the transport process on the network to an abstract Cauchy problem. Assuming the same transport velocities on each edge of the graph we can give an explicit formula of the appropriate semigroup solving this abstract Cauchy problem. For the case where we have different speeds on the edges, we have to restrict ourselves to finite networks since we apply a Trotter–Kato approximation theorem to handle this case. The conditions in this theorem forces the networks to be finite.

#### Concerning originality

We remark that not all results in this thesis are to be considered original. Moreover, not all new results use "new techniques". This is the reason that we want to separate here the wheat from the chaff. Firstly, we notice that some of the chapters of this thesis are based on papers:

- Chapter 2 is based on the paper Intermediate and extrapolated spaces for bi-continuous operator semigroups [28] (with minor modification and without the part about the implemented semigroup), which is joint work with B. Farkas and has been published in Journal of Evolution Equations (DOI:10.1007/s00028-018-0477-8).
- Chapter 4 is based on the paper A Desch-Schappacher perturbation theorem for bicontinuous semigroups [27] (with minor modifications and without the part about the implemented semigroup), which is joint work with B. Farkas and accepted for publication in Mathematische Nachrichten, and is available online as ArXiv preprint (ArXiv:1811.08455).
- Chapter 7 is based on the paper *Bi-continuous semigroups for flows in infinite networks* [29] (with some more preliminaries on networks), which is joint work with M. Kramar Fijavž, which is submitted to *Operators and Matrices*, and is available online as ArXiv preprint (ArXiv:1901.10292).

As already mentioned above, Chapter 1 has an introducing character. This means among others, that the first chapter consists of already known results. However, we choose to include this chapter to stay as self-contained as possible. In this chapter we present two important examples of bi-continuous semigroups in detail. These examples already occured in [78, Sect. 3.5.1] and [54, Rem. 2.5.3], but to be self-contained we added some details concerning

the underlying Banach spaces and locally convex topologies. To do so, some of the arguments are taken from [103, Chapter 2]. Particularly, in Section 1.2.1 we use the arguments from [103, Exam. 2.2.4] whereas we use [103, Thm. 2.1.6] at a later time for the discussion of implemented semigroups in Section 6.1.1.

The novelty of Chapter 2 starts with the the construction of extrapolation spaces for nondensely defined operator due to Definition 2.9. A way more general consideration of extrapolation spaces is treated by Theorem 2.16 using a new approach which is connected with universal extrapolation spaces. Until here we do not use the assumption that the operators we consider are generators of (bi-continuous) semigroups. We connect the extrapolation spaces with semigroups in a general context in Section 2.3. Moreover, we consider the Hille– Yosida generation theorem for bi-continuous semigroups, cf. Theorem 2.38. Even though the statement of this theorem is already known, the techniques for the proof using extrapolation are new. The extrapolation of bi-continuous semigroups is a result of Proposition 2.40. The results in Section 2.4.3 and Section 2.4.4, treating Hölder spaces for bi-continuous semigroups, are new. The proofs uses techniques similar to theses in [52, Chapter II, Sect. 5(b)]. The concluding examples in Section 2.5 generalize the examples of extrapolation spaces in the strongly continuous case, cf. [52, Chapter II, Exam. 5.8 & Ex. 5.23(5)].

The definition of fiber  $L^p$ -spaces in Section 3.2 is originally due to R. Heymann. However, Theorem 3.14 regarding extrapolation spaces of unbounded operator-valued multiplication operators, by means of these fiber  $L^p$ -spaces, is new and the key result in this chapter. The results from Section 3.3 help us to prove our main result. In fact, the main result gives an insight how to describe the domain of evolution semigroups coming from the solutions of non-autonomous abstract Cauchy problem.

The main results of Chapter 4 are Theorem 4.4 and Theorem 4.7. These are new results since they discuss Desch–Schappacher perturbations for bi-continuous semigroups. Here we use the fact from Chapter 2 that we can extrapolate bi-continuous semigroups. The proofs of the main results are at some places verbatime the same as for the strongly continuous case [52, Chapter III, Thm. 3.1 & Cor. 3.3]. However, we always have to take care of the locally convex topology in the bi-continuous case which leads to some technical subtleties. The example in Section 4.3 is based on [52, Chapter III, Exam. 3.5], however, we consider a different underlying function space which fits in the bi-continuous framework. A novel application of the Desch–Schappacher theorem is later on given in Chapter 6.

Chapter 5 is devoted to Theorem 5.19 as main result. As a matter of fact, the result is based on [119, Thm. 0.1]. We introduce the new notion of bi-AL-spaces in Definition 5.9, which is related to the classical AL-spaces. In order to prove the main result, we use Lemma 5.20 as an auxiliary tool. For the proofs we proceed similar to the original proofs [119, Lemma 2.1] and [119, Thm. 0.1]. However, the argumentation differs, since we work with an additional locally convex topology. The first example of Section 5.3.1 is based on [13, Thm. 2.2] whereas Section 5.3.2 covers a new example on the space of bounded Borel measures, which is in fact the adjoint semigroup of the Gauss–Weierstrass semigroup.

The novelty of Chapter 6 is the consideration of extrapolation and perturbations of implemented semigroups. Although, J. Alber deals with extrapolation spaces [6], our approach is new since we do not only consider extrapolation on the space of strong continuity but in the new setting of bi-continuous semigroups as dicussed in Chapter 2. The highlights of this chapter is the extrapolation procedure of the left implemented semigroup, cf. Section 6.2.2, Lemma 6.4 where we discuss the properties of the domain of the generator of bi-continuous semigroups on the space  $\mathscr{L}(E)$  by means of ideals as well as Theorem 6.6, Theorem 6.7 and Theorem 6.14 where we relate perturbations of implemented semigroups with these of strongly continuous semigroups.

The study of large networks by means of operator semigroups, as it is discussed in Chapter 7. in not new, cf. [41, 43, 42]. However, an approach via bi-continuous semigroups considered on the phase space  $L^{\infty}([0,1], \ell^1)$  is novel. In this context, Theorem 7.9 is a main result. Added to that we consider also Corollary 7.13 and Theorem 7.15 as highlight of this chapter. The novelty of the approach is in the first instance the use of bi-continuous semigroups and moreover that we can not make use of the Lumer–Phillips generation theorem as it is used in the strongly continuous case, cf. [21, Cor. 18.15]. For this reason we use the Trotter–Kato approximation theorem for bi-continuous semigroups, cf. [78, Thm. 2.3 & Thm. 2.6], [54, Thm. 1.2.10] and [4].

# Chapter 1 Bi-Continuous Semigroups

#### § 1.1 Bi-admissible spaces

In this first chapter we concentrate on the basic theory of *bi-continuous semigroups*. This class of semigroups was first introduced by F. Kühnemund in [78]. The following assumptions, as proposed by F. Kühnemund, cf. [78, Assum. 1.1], will be made during the whole thesis.

**Assumption 1.1.** Consider a triple  $(X_0, \|\cdot\|, \tau)$  where  $X_0$  is a Banach space, and

- 1.  $\tau$  is a locally convex Hausdorff topology coarser than the norm-topology on  $X_0$ , i.e., the identity map  $(X_0, \|\cdot\|) \to (X_0, \tau)$  is continuous;
- 2.  $\tau$  is sequentially complete on the  $\|\cdot\|$ -closed unit ball, i.e., every  $\|\cdot\|$ -bounded  $\tau$ -Cauchy sequence is  $\tau$ -convergent;
- 3. The dual space of  $(X_0, \tau)$  is norming for  $X_0$ , i.e.,

$$\|x\| = \sup_{\substack{\varphi \in (X_0, \tau)' \\ \|\varphi\| \le 1}} |\varphi(x)|.$$
(1.1.1)

- **Remark 1.2.** (i) There is the related notion of so-called Saks spaces, see [34]. By definition a *Saks space* is a triple  $(X_0, \|\cdot\|, \tau)$  such that  $X_0$  is a vector space with a norm  $\|\cdot\|$  and locally convex topology  $\tau$  in such a way that  $\tau$  is coarser than the  $\|\cdot\|$ -topology, but the closed unit ball is  $\tau$ -complete. In particular,  $X_0$  is a Banach space.
  - (ii) There is also a connection to norming dual pairs discussed in [80]. More information about this can also be found in Section 2.3. In particular,  $(X_0, Y)$  with  $Y = (X_0, \tau)'$  is a norming dual pair.
- (iii) R. Kraaij puts this setting in a more general framework of locally convex spaces with mixed topologies, see [75, Sec. 4], and also [54, App. A]
- (iv) Recall from [101, Thm. 1.36 & 1.37] that every locally convex topology gives rise to a family of seminorms and vice versa. In this regard Assumption (1.1.1) is equivalent to the following: There is a set  $\mathcal{P}$  of  $\tau$ -continuous seminorms defining the topology  $\tau$ , such that

$$||x|| = \sup_{p \in \mathcal{P}} p(x).$$
 (1.1.2)

This description is also used by R. Kraaij in [75], cf. his Lemma 4.4. Note also that by this remark and by Lemma 3.1 in [34] we see that a Saks space satisfies Assumption 1.1. Indeed, assume (1.1.1) and let  $\mathcal{P}$  be the collection of all  $\tau$ -continuous seminorms psuch that  $p(x) \leq ||x||$ . Then  $|\varphi(\cdot)| \in \mathcal{P}$  for each  $\varphi \in (X_0, \tau)'$  with  $||\varphi|| \leq 1$ , and (1.1.2) is trivially satisfied. If q is any  $\tau$ -continuous seminorm, then  $q(x) \leq M||x||$  for some constant M and for all  $x \in X_0$ . So that  $q/M \in \mathcal{P}$ , proving that  $\mathcal{P}$  defines precisely the topology  $\tau$ . For the converse implication suppose that (1.1.2) holds. Then by the application of the Hahn–Banach theorem we obtain (1.1.1).

For the sake of completeness and to be self-contained we discuss two typical examples of spaces which satisfy Assumption 1.1 in the following section. For related examples we refer to [54, Sect. 1.2].

#### 1.1.1 The space $C_b(\Omega)$ of bounded continuous functions

Consider an arbitrary topological space  $\Omega$  with a Hausdorff topology  $\kappa$  on it. With  $C_b(\Omega)$  we denote the space of all bounded continuous functions  $f : \Omega \to \mathbb{R}$ . Equipped with the supremum-norm  $\|\cdot\|_{\infty}$ , defined by

$$||f||_{\infty} := \sup_{x \in \Omega} |f(x)|, \quad f \in \mathcal{C}_{\mathbf{b}}(\Omega),$$

this becomes a Banach space. Another topology on  $C_b(\Omega)$  is the *compact-open topology*  $\tau_{co}$ . To describe this topology, we define for a compact subset  $K \subseteq \Omega$  and an open subset  $U \subseteq \mathbb{R}$  the set

$$\mathcal{V}(K,U) := \{ f \in \mathcal{C}_{\mathbf{b}}(\Omega) : f(K) \subseteq U \}$$

The collection of all such sets forms a subbase for  $\tau_{co}$ , i.e., every proper open set of the compact-open topology can be written as a union of finite intersections of elements of the form  $\mathcal{V}(K, U)$ . Another description of this topology uses seminorms. In fact, a generating family of seminorms  $\mathcal{P}$  is given by

$$\mathcal{P} = \{ p_K : K \subseteq \Omega \text{ compact} \}, \quad p_K(f) := \sup_{x \in K} |f(x)|.$$

If the topological space  $(\Omega, \kappa)$  is a Tychonoff space, i.e., every singleton subset of  $\Omega$  is closed and for every closed subset  $C \subseteq \Omega$  and each  $x \in \Omega \setminus C$  there exists a continuous function  $f: \Omega \to [0, 1]$  such that f(x) = 0 and  $f_{|C} \equiv 1$ , then  $\tau_{co}$  is a Hausdorff topology. If moreover  $(\Omega, \kappa)$  is a  $k_f$ -space, i.e., the continuity of a function  $f: \Omega \to \mathbb{R}$  depends only on the continuity of  $f_{|K}$  for all compact  $K \subseteq \Omega$ , then  $C_b(\Omega)$  is  $\tau_{co}$ -complete on norm-bounded sets. Notice that every locally compact or metrizable space has this property. Observe that the point measures are  $\tau_{co}$ -continuous and hence we conclude that  $(C_b(\Omega), \tau_{co})$  is norming for  $(C_b(\Omega), \|\cdot\|_{\infty})$ . Even for a Banach space X, the space  $C_b(\Omega, X)$  of vector-valued, continuous, bounded functions fits into the framework of Assumption 1.1.

#### 1.1.2 The dual space X'

The second example makes use of duality in Banach spaces. Let X be a Banach space and X' the norm dual of X, consisting of all linear functionals  $\varphi : X \to \mathbb{C}$  which are continuous with respect to the Banach space norm. Then X' becomes a Banach space with respect to

the norm defined by

$$\|\varphi\|_{X'} := \sup_{\substack{x \in X \\ \|x\| \le 1}} |\langle x, \varphi \rangle|, \quad \varphi \in X',$$

where  $\langle \cdot, \cdot \rangle : X \times X' \to \mathbb{C}$  is the (canonical) dual pairing between X and X' defined by  $\langle x, \varphi \rangle := \varphi(x)$  for  $x \in X$  and  $\varphi \in X'$ . In addition X' becomes a locally convex space, with the family of generating seminorms  $\mathcal{P}$  defined by

$$\mathcal{P} = \{ p_x : x \in X \}, \quad p_x(\varphi) := |\varphi(x)| = |\langle x, \varphi \rangle|$$

The corresponding topology is called the *weak*<sup>\*</sup>-topology  $\tau_{w^*}$  and is the coarsest topology making the evaluation maps  $\varphi \mapsto \varphi(x)$  continuous. We notice that a sequence  $(\varphi_n)_{n \in \mathbb{N}}$  in X' converges to  $\varphi$  with respect to the weak<sup>\*</sup>-topology if and only if  $\varphi_n(x) \to \varphi(x)$  for each  $x \in X$ . This topology is sometimes also denoted by  $\sigma(X', X)$ . We now show that  $\sigma(X', X)$ satisfies Assumption 1.1. To do so, notice that  $\sigma(X', X)$  is not first countable in general, so we have to argue by nets instead of sequences. To show that  $\sigma(X', X)$  is weaker than the norm-topology observe that  $|\langle x, \varphi \rangle| \leq ||x|| \, ||\varphi||$  for each  $x \in X$  and  $\varphi \in X'$ . Hence, if  $(\varphi_i)_{i \in I}$  is a net converging in the norm of X', then  $(\varphi_i)_{i \in I}$  converges also with respect to the weak<sup>\*</sup>-topology on X', meaning that  $\sigma(X', X)$  is weaker than the norm topology. That  $\sigma(X', X)$  is a Hausdorff topology is trivial, i.e., let  $\varphi, \psi \in X'$  such that  $\varphi \neq \psi$ , then there exists  $x \in X$  with  $p_x(\varphi) = |\varphi(x)| \neq |\psi(x)| = p_x(\psi)$ . The norming property follows from the definition of the norm

$$\|\varphi\|_{X'} = \sup_{\|x\| \le 1} |\varphi(x)| = \sup_{\|x\| \le 1} p_x(\varphi).$$

Now let  $(\varphi_n)_{n\in\mathbb{N}}$  be a norm-bounded weak\*-Cauchy sequence in X'. Then there exists  $M \ge 0$ such that  $\|\varphi_n\| \le M$  for each  $n \in \mathbb{N}$  and  $(\varphi_n(x))_{n\in\mathbb{N}}$  is a Cauchy sequence in  $\mathbb{R}$  for each  $x \in X$ and hence  $\lim_{n\to\infty} \varphi_n(x)$  exists for each  $x \in X$ . Define  $\varphi \in X'$  by  $\varphi(x) := \lim_{n\to\infty} \varphi_n(x)$ and observe that dy definition  $\varphi_n \to \varphi$  with respect to the weak\*-topology. In addition, we obtain that also the limit  $\varphi$  is bounded with  $\|\varphi\| \le M$ .

#### § 1.2 Bi-continuous semigroups

Now we formulate the definition of a bi-continuous semigroup which is originally due to F. Kühnemund, cf. [79, Def. 3].

**Definition 1.3.** Let  $X_0$  be a Banach space with norm  $\|\cdot\|$  together with a locally convex topology  $\tau$ , such that the conditions in Assumption 1.1 are satisfied. We call a family of bounded linear operators  $(T(t))_{t>0}$  a *bi-continuous semigroup* if it has the following properties.

- 1. T(t+s) = T(t)T(s) and T(0) = I for all  $s, t \ge 0$ .
- 2.  $(T(t))_{t\geq 0}$  is strongly  $\tau$ -continuous, i.e., the map  $\varphi_x : [0,\infty) \to (X_0,\tau)$  defined by  $\varphi_x(t) = T(t)x$  is continuous for every  $x \in X_0$ .
- 3.  $(T(t))_{t\geq 0}$  is exponentially bounded, i.e., there exist  $M \geq 1$  and  $\omega \in \mathbb{R}$  such that  $||T(t)|| \leq Me^{\omega t}$  for each  $t \geq 0$ .
- 4.  $(T(t))_{t\geq 0}$  is locally-bi-equicontinuous, i.e., if  $(x_n)_{n\in\mathbb{N}}$  is a norm-bounded sequence in  $X_0$ which is  $\tau$ -convergent to 0, then  $(T(s)x_n)_{n\in\mathbb{N}}$  is  $\tau$ -convergent to 0 uniformly for  $s \in [0, t_0]$ for each fixed  $t_0 \geq 0$ .

The following definition will occur frequently in the forthcoming chapters and originates from the theory of strongly continuous semigroups, cf. [52, Chapter I, Def. 5.6].

**Definition 1.4.** For an exponentially bounded semigroup of bounded linear operators  $(T(t))_{t>0}$  we define the *growth bound* to be

$$\omega_0(T) := \inf \left\{ \omega \in \mathbb{R} : \exists M \ge 1 \; \forall t \ge 0 : \|T(t)\| \le M e^{\omega t} \right\}.$$

We now continue with examples of bi-continuous semigroups on the space  $C_b(\mathbb{R})$  and X' which we already discussed before in Section 1.1.1 and Section 1.1.2.

#### 1.2.1 The left-translation semigroup

As we already saw the space  $X_0 = C_b(\mathbb{R})$  equipped with the supremum-norm and the compactopen topology  $\tau_{co}$  satisfies Assumption 1.1. Now consider the *left-translation semigroup*  $(T(t))_{t\geq 0}$  on  $C_b(\mathbb{R})$  defined by

$$(T(t)f)(x) := f(x+t), \quad t \ge 0, \ f \in \mathcal{C}_{\mathcal{b}}(\mathbb{R}), \ x \in \mathbb{R}.$$

This semigroup actually fails to be strongly continuous with respect to the supremum-norm. In particular, since T(t)f(x) - f(x) = f(x+t) - f(x) the continuity of  $t \mapsto T(t)f$  entails that the function f has to be uniformly continuous. By taking the function  $f(x) := \sin(x^2)$ for  $x \in \mathbb{R}$ , which is bounded continuous and not uniformly continuous, we conclude that  $||T(t)f - f||_{\infty}$  does not vanish for  $t \to 0$ . Although  $(T(t))_{t\geq 0}$  is not a  $C_0$ -semigroup, we can show that it is bi-continuous with respect to the compact-open topology. Since  $||T(t)f||_{\infty} =$  $||f||_{\infty}$  for each  $f \in C_b(\mathbb{R})$ ,  $(T(t))_{t\geq 0}$  becomes a contraction semigroup and therefore it is exponentially bounded. Observe that for a compact subset  $K \subseteq \mathbb{R}$  holds that

$$p_K(T(t)f - f) = \sup_{x \in K} |f(x + t) - f(x)|, \quad t \ge 0, \ f \in \mathcal{C}_{\mathbf{b}}(\mathbb{R}).$$

Since f is continuous and K is compact,  $f_{|K}$  is uniformly continuous, hence for each  $\varepsilon > 0$ there exists  $\delta > 0$  such that

$$\sup_{x \in K} |f(x+t) - f(x)| < \varepsilon$$

whenever  $|t| < \delta$  and we obtain the strong continuity of  $(T(t))_{t\geq 0}$  with respect to  $\tau_{co}$ . For the local bi-equicontinuity let  $(f_n)_{n\in\mathbb{N}}$  be a  $\|\cdot\|_{\infty}$ -bounded  $\tau_{co}$ -null sequence. For  $t_0 > 0$  and  $t \in [0, t_0]$  we obtain

$$p_K(T(t)f_n) = \sup_{x \in K} |f_n(x+t)| \le \sup_{x \in H} |f_n(x)| = p_H(f_n),$$

where  $H := \bigcup_{t \in [0,t_0]} K + t$  is compact. This shows that  $T(t)f_n \xrightarrow{\tau_{co}} 0$  uniformly for  $t \in [0,t_0]$ . We conclude that the left-translation semigroup is indeed a bi-continuous semigroup on  $C_b(\mathbb{R})$ .

#### 1.2.2 The adjoint semigroup

Let  $(T(t))_{t\geq 0}$  be a  $C_0$ -semigroup on a Banach space X. Recall that the adjoint of a bounded linear operator  $T \in \mathscr{L}(X)$  is defined to be the unique bounded linear operator  $T' \in \mathscr{L}(X')$  such that

$$\langle Tx, \varphi \rangle = \langle x, T'\varphi \rangle$$

for all  $x \in X$  and  $\varphi \in X'$ . The adjoint semigroup  $(T(t)')_{t\geq 0}$  on the dual Banach space X'consist of all adjoint operators T(t)' on X'. To show that in general  $(T(t)')_{t\geq 0}$  is not strongly continuous with respect to the norm on X' consider the strongly continuous left-translation semigroup  $(T(t))_{t\geq 0}$  on  $L^1(\mathbb{R})$  and have a look at its adjoint, the right-translation semigroup on  $L^{\infty}(\mathbb{R})$ . By taking the the characteristic function on the interval [0, 1], i.e.,  $f(x) := \mathbf{1}_{[0,1]}(x)$ for  $x \in \mathbb{R}$ , one observes that ess  $\sup_{x\in\mathbb{R}} |f(x-t) - f(x)| = 1$  for t > 0. Hence this semigroup is not strongly continuous. Since we already saw that X' equipped with the weak\*-topology  $\tau_{w^*}$  satisfies Assumption 1.1 we show that the adjoint semigroup is bi-continuous on X' with respect to  $\tau_{w^*}$  (see also [78, Prop. 3.18]). The exponential boundedness is clear, since  $(T(t))_{t\geq 0}$ was exponentially bounded. By the strong continuity of  $(T(t))_{t\geq 0}$  we obtain for  $x' \in X'$  and  $y \in X$ 

$$\left|\left\langle y,T(t)'x'-x'\right\rangle\right|=\left|\left\langle T(t)y-y,x'\right\rangle\right|\leq\left\|x'\right\|\cdot\left\|T(t)y-y\right\|$$

and hence the strong continuity of  $(T(t)')_{t\geq 0}$  with respect to the weak\*-topology. By using the same argument and the fact that  $\{T(t)x : t \in [0, t_0]\}$  is compact, we also conclude that  $(T(t)')_{t\geq 0}$  is locally bi-equicontinuous.

**Remark 1.5.** Other important examples are evolution semigroups on  $C_b(\mathbb{R}, X)$ , semigroups induced by flows, see [78, Sect. 3.2], and the Ornstein–Uhlenbeck semigroup on  $C_b(\mathbb{R})$ . For more details we refer to [78, Sect. 3.3] and [54, Sect. 2.3]. In Chapter 6 we consider another interesting example, the implemented semigroups.

#### § 1.3 The generator

As in the case of  $C_0$ -semigroups, [52, Chapter II, Def. 1.2], we can define a generator for a bi-continuous semigroup in the following way, cf. [54, Def. 1.2.6].

**Definition 1.6.** Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup on  $X_0$ . The generator A is defined by

$$Ax := \tau \lim_{t \to 0} \frac{T(t)x - x}{t}$$

with the domain

$$D(A) := \Big\{ x \in X_0 : \ \tau \lim_{t \to 0} \frac{T(t)x - x}{t} \text{ exists and } \sup_{t \in (0,1]} \frac{\|T(t)x - x\|}{t} < \infty \Big\}.$$

This definition of the generator leads to a couple of important properties. Recall that for an linear operator (A, D(A)) on a Banach space X the resolvent set  $\rho(A)$  consists of  $\lambda \in \mathbb{C}$  such that  $\lambda - A$  is invertible, i.e., there exists a bounded operator B with  $Bx \in D(A)$  for each  $x \in X$  such that  $(\lambda - A)Bx = x$  for each  $x \in X$  and  $B(\lambda - A)x = x$  for each  $x \in D(A)$ . The inverse of  $\lambda - A$  is often denoted by  $R(\lambda, A) := (\lambda - A)^{-1}$ . The complement  $\sigma(A) := \mathbb{C} \setminus \rho(A)$  is called the *spectrum*. The following theorem summarizes some properties the generator of a bi-continuous semigroup, see [79, Sect. 1.2] and [54, Thm. 1.2.7]:

**Theorem 1.7.** Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup on a Banach space  $X_0$  with respect to  $\tau$  and with generator (A, D(A)). Then the following hold:

- (a) The operator A is bi-closed, i.e., whenever  $(x_n)_{n \in \mathbb{N}}$  is a sequence in D(A) such that  $x_n \xrightarrow{\tau} x$  and  $Ax_n \xrightarrow{\tau} y$  and both sequences are norm-bounded, then  $x \in D(A)$  and Ax = y.
- (b) The domain D(A) is bi-dense in  $X_0$ , i.e., for each  $x \in X_0$  there exists a norm-bounded sequence  $(x_n)_{n \in \mathbb{N}}$  in D(A) such that  $x_n \xrightarrow{\tau} x$ .
- (c) For  $x \in D(A)$  we have  $T(t)x \in D(A)$  and T(t)Ax = AT(t)x for all  $t \ge 0$ .
- (d) For t > 0 and  $x \in X_0$  one has

$$\int_{0}^{t} T(s)x \, \mathrm{d}s \in D(A) \quad and \quad A \int_{0}^{t} T(s)x \, \mathrm{d}s = T(t)x - x.$$
(1.3.1)

(e) For  $\lambda > \omega$  one has  $\lambda \in \rho(A)$  (thus A is closed) and for  $x \in X_0$  holds:

$$R(\lambda, A)x = \int_0^\infty e^{-\lambda s} T(s)x \, \mathrm{d}s \tag{1.3.2}$$

where the integral is a  $\tau$ -improper integral.

From Definition 1.6 and Theorem 1.7 we conclude that every bi-continuous semigroup gives rise to a (bi)-closed and bi-densely defined operator (A, D(A)). The question under what conditions an operator (A, D(A)) is the generator of a bi-continuous semigroup is given by the *Hille–Yosida generation theorem*. For  $C_0$ -semigroups this theorem was originally proven by K. Yosida [124] and E. Hille [66, Thm. 12.2.1] and can for example also be found in [52, Chapter II, Thm. 3.8]. Kühnemund also proved a similar generation theorem for bi-continuous semigroups in [78, Thm. 1.28]. There integrated semigroups are used for the proof of the theorem. In Chapter 2 we give a new direct proof of this theorem. Since we use extrapolation methods for this, we talk about the Hille–Yosida theorem for bi-continuous semigroups later on in Section 2.4.1. As a matter of fact, this will be Theorem 2.38 in this thesis. The Hille– Yosida Theorem is closely related to the problem of the existence and uniqueness of solutions of a evolution equations. This correspondence will be discussed in Section 1.5.

In what follows we determine the generator of the translation semigroup from Section 1.2.1 and of the adjoint semigroup from Section 1.2.2.

#### 1.3.1 The left-translation semigroup on $C_b(\mathbb{R})$

Let us start with the generator of the translation semigroup on  $C_b(\mathbb{R})$  from Section 1.2.1. The following result identifies the generator explicitly.

**Proposition 1.8.** Let  $(T(t))_{t\geq 0}$  be the left-translation semigroup on  $C_b(\mathbb{R})$ . The generator is given by Af := f' with domain  $D(A) := C_b^1(\mathbb{R})$ , the space of differentiable functions with bounded and continuous derivative.

*Proof.* If  $f \in C^1_{\rm b}(\mathbb{R})$ , then we obtain by the fundamental theorem of calculus

$$\frac{T(t)f(x) - f(x)}{t} = \frac{f(x+t) - f(x)}{t} = \frac{1}{t} \int_0^t f'(x+s) \, \mathrm{d}s, \quad x \in \mathbb{R}, \ t > 0.$$

Now let  $K \subseteq \mathbb{R}$  be an arbitrary compact subset. Since by assumption the derivative of f is continuous, the restriction of f' to K is uniformly continuous, i.e., for arbitrary  $\varepsilon > 0$  there

exists  $\delta > 0$  such that  $|f'(x) - f'(y)| < \varepsilon$  for every  $x, y \in K$  whenever  $|x - y| < \delta$ . Hence for  $0 < t < \delta$  we obtain

$$\sup_{x \in K} \left| \frac{f(x+t) - f(x)}{t} - f'(x) \right| \le \frac{1}{t} \int_0^t \sup_{x \in K} \left| f'(x+s) - f'(x) \right| \, \mathrm{d}s < \varepsilon.$$

Hence  $\frac{1}{t}(T(t)f - f)$  converges uniformly on compact sets to f' as  $t \to 0$ . Since  $f \in C^1_b(\mathbb{R})$  one also has  $f \in Lip(\mathbb{R})$  which means that there exists  $K \ge 0$  such that  $|f(x) - f(y)| \le K |x - y|$  for all  $x, y \in \mathbb{R}$ . In particular

$$\sup_{t \in (0,1]} \frac{\|T(t)f - f\|_{\infty}}{t} \le \sup_{t \in (0,1]} \sup_{x \in \mathbb{R}} \frac{Kt}{t} = K.$$

We conclude that  $C^1_b(\mathbb{R}) \subseteq D(A)$ .

Conversely, suppose that  $f \in D(A)$ . Then  $\frac{1}{t}(f(\cdot + t) - f(\cdot))$  converges uniformly on compact intervals to Af in  $C_b(\mathbb{R})$  as t tends to 0. In particular, for each compact subset  $K \subseteq \mathbb{R}$  and each  $x \in K$  the difference quotient  $\frac{1}{t}(f(\cdot + t) - f(\cdot))$  converges to (Af)(x). We conclude that f is differentiable in  $\mathbb{R}$  and  $f' = Af \in C_b(\mathbb{R})$ . Hence  $D(A) \subseteq C_b^1(\mathbb{R})$ .

#### 1.3.2 The adjoint semigroup

Consider a  $C_0$ -semigroup  $(T(t))_{t\geq 0}$  with generator (A, D(A)) on a Banach space X and the corresponding adjoint semigroup  $(T'(t))_{t\geq 0}$  on X'. We already saw in Section 1.2.2 that this leads to a bi-continuous semigroup with respect to the weak\*-topology on X'. Recall the following definition [52, Appendix, Def. B.8].

**Definition 1.9.** For a densely defined operator (A, D(A)) on X, the *adjoint operator* (A', D(A')) on X' is defined by

$$D(A') := \left\{ x' \in X' : \exists y' \in X' \; \forall x \in D(A) : \langle Ax, x' \rangle = \langle x, y' \rangle \right\}, \quad A'x' := y'.$$

We prove the following assertion which is already mentioned as an exercise in [52, Chapter I, Sect. 2.5] and can also be found in [116, Thm. 1.2.3].

**Proposition 1.10.** Let  $(T(t))_{t\geq 0}$  be a strongly continuous semigroup on X. The generator (B, D(B)) of the adjoint semigroup  $(T'(t))_{t\geq 0}$  on X' is given by D(B) = D(A') and Bx' := A'x'.

*Proof.* Firstly, fix  $x' \in D(A')$ . By an application of [116, Prop. 1.2.2] we obtain that for arbitrary  $x \in X$  the following holds.

$$\lim_{t \to 0} \frac{1}{t} \left\langle x, T'(t)x' - x' \right\rangle = \lim_{t \to 0} \frac{1}{t} \left\langle x, A' \int_0^t T'(s)x' \, \mathrm{d}s \right\rangle = \lim_{t \to 0} \frac{1}{t} \int_0^t \left\langle x, T'(s)A'x' \right\rangle \, \mathrm{d}s = \left\langle x, A'x' \right\rangle.$$

Hence the limit  $\lim_{t\to 0} \frac{1}{t}(T'(t)x' - x')$  exists with respect to the weak\*-topology and is equal to A'x'. We conclude that  $x' \in D(B)$  and Bx' = A'x'. This shows that  $A' \subseteq B$ . For the converse fix  $x' \in D(B)$ , let  $x \in D(A)$  be arbitrary and take notice of the following equality

$$\langle x, Bx' \rangle = \lim_{t \to 0} \frac{1}{t} \langle x, T'(t)x' - x' \rangle = \langle Ax, x' \rangle$$

This shows that  $x' \in D(A')$  and A'x' = Bx', showing that  $B \subseteq A'$ .

#### § 1.4 The space of strong continuity

The following statement relates bi-continuous semigroups and  $C_0$ -semigroups. In particular, by [78, Thm. 1.18(d)] we can always restrict a given bi-continuous semigroup to a large subspace of  $X_0$  which is invariant under the semigroup and cause by restriction a strongly continuous semigroup on this subspace.

**Proposition 1.11.** Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup on  $X_0$  with generator (A, D(A)). Then  $\underline{X}_0 := \overline{D(A)}^{\|\cdot\|}$  is invariant under  $(T(t))_{t\geq 0}$ . Moreover, the restriction  $(\underline{T}(t))_{t\geq 0} := (T(t)_{|\underline{X}_0})_{t\geq 0}$  is a strongly continuous semigroup on  $\underline{X}_0$  generated by  $(\underline{A}, D(\underline{A}))$ , the part of A in  $\underline{X}_0$ .

This property will be very useful in Chapter 2 when we discuss extrapolation spaces. There we also show that the space  $\underline{X}_0$  from Proposition 1.11 coincides with the space of strong continuity, i.e.,  $\underline{X}_0$  consists of all  $x \in X_0$  such that the map  $t \mapsto T(t)x$  is strongly continuous with respect to the norm topology on  $X_0$ , cf. Lemma 2.31. We will take this result here for granted, since we prove this result later on in Chapter 2 in a more general setting, and provide the spaces of strong continuity  $\underline{X}_0$  for our reccurrent examples from this chapter.

#### 1.4.1 The left-translation semigroup on $C_b(\mathbb{R})$

The following result determines the space of strong continuity for this semigroup.

**Lemma 1.12.** The space of strong continuity for the left-translation semigroup is given by  $\underline{X}_0 = \mathrm{UC}_{\mathrm{b}}(\mathbb{R})$ , the space of bounded uniformly continuous functions.

Proof. It is not hard to see that the left-translation semigroup is strongly continuous for  $f \in \mathrm{UC}_{\mathrm{b}}(\mathbb{R})$ . Indeed T(t)f converges to f with respect to the supremum-norm for  $t \to 0$  if and only if T(t)f converges uniformly to f in  $\mathbb{R}$  for  $t \to 0$  and this happens if and only if  $||T(t)f - f||_{\infty} = \sup_{x \in \mathbb{R}} |f(x+t) - f(x)|$  vanishes for  $t \to 0$ . By rewriting the definition of uniform continuity we see that this condition holds. Hence  $\mathrm{UC}_{\mathrm{b}}(\mathbb{R}) \subseteq \underline{X}_0$ . Conversely, let  $f \in \underline{X}_0$ , i.e., there exists a sequence  $(f_n)_{n \in \mathbb{N}}$  in  $\mathrm{C}^1_{\mathrm{b}}(\mathbb{R})$  such that  $f_n \to f$  with respect to the supremum-norm. This means that f is the uniform limit of the sequence  $(f_n)_{n \in \mathbb{N}}$ . Since  $f_n \in \mathrm{C}^1_{\mathrm{b}}(\mathbb{R})$  we know that  $f_n \in \mathrm{UC}_{\mathrm{b}}(\mathbb{R})$ . Since uniform convergence of functions perserves uniform continuity and boundedness we conclude that for the limit  $f \in \mathrm{UC}_{\mathrm{b}}(\mathbb{R})$  must hold, showing that  $\underline{X}_0 \subseteq \mathrm{UC}_{\mathrm{b}}(\mathbb{R})$ .

**Remark 1.13.** There is another elementary way, which avoids Lemma 2.31, to show that the norm-closure of  $C_b^1(\mathbb{R})$  coincides with  $UC_b(\mathbb{R})$ , namely by using mollifiers. As shown above, it thus remains to approximate every  $f \in UC_b(\mathbb{R})$  by elements of  $C_b^1(\mathbb{R})$ , since  $UC_b(\mathbb{R})$  is norm-closed. To proceed define the function  $\eta : \mathbb{R} \to \mathbb{R}$  by

$$\eta(x) := \begin{cases} C e^{\frac{1}{|x|^2 - 1}}, & |x| < 1, \\ 0, & |x| \ge 1, \end{cases}$$

where C > 0 is a constant such that  $\int_{\mathbb{R}} \eta(x) \, dx = 1$ . Then  $\eta \in C^{\infty}(\mathbb{R})$  and is called the *(standard) mollifier*. Moreover, for each  $\varepsilon > 0$  the function

$$\eta_{\varepsilon}(x) := \frac{1}{\varepsilon} \eta\left(\frac{x}{\varepsilon}\right)$$

is again a C<sup>∞</sup>-function and satisfies  $\operatorname{supp}(\eta_{\varepsilon}) \subseteq (-\varepsilon, \varepsilon)$  as well as  $\int_{\mathbb{R}} \eta_{\varepsilon}(x) \, dx = 1$ . For  $n \in \mathbb{N}$  consider the function  $f_n := \eta_{\frac{1}{n}} * f$ . By the observation that for each  $n \in \mathbb{N}$  one has  $\frac{\mathrm{d}}{\mathrm{d}x} \left(\eta_{\frac{1}{n}} * f\right) = \left(\frac{\mathrm{d}}{\mathrm{d}x} \eta_{\frac{1}{n}}\right) * f$  one concludes that  $\eta_{\frac{1}{n}} \in \mathrm{C}^1_{\mathrm{b}}(\mathbb{R})$  for each  $n \in \mathbb{N}$ . Moreover one shows that  $\|f_n - f\|_{\infty} \to 0$  as n tends to infinity.

#### 1.4.2 The adjoint semigroup

We know that the adjoint semigroup  $(T'(t))_{t\geq 0}$  on X' of a  $C_0$ -semigroup is not strongly continuous with respect to the norm but is bi-continuous with respect to the weak\*-topology. The space of strong continuity is handled by the so-called *sun dual* defined by

$$X^{\odot} := \left\{ x' \in X' : \lim_{t \to 0} \|T'(t)x' - x'\| = 0 \right\}.$$

This concept is e.g. discussed in [116, Sect. 1.3] and [52, Chapter II, Sect. 2.6]. It was shown that the generator  $(A^{\odot}, D(A^{\odot}))$  of the corresponding strongly continuous semigroup  $(T(t)^{\odot})_{t\geq 0}$  on  $X^{\odot}$  is given by the part of the generator (A', D(A')) on  $(T'(t))_{t\geq 0}$  in  $X^{\odot}$ , i.e.,

$$A^{\odot}x' = A'x', \quad D(A^{\odot}) = \left\{ x' \in D(A'): A'x' \in X^{\odot} \right\}.$$

Since  $D(A^{\odot})$ , as a generator of a  $C_0$ -semigroup, is dense in  $X^{\odot}$  one directly concludes that  $X^{\odot} = \overline{D(A')}^{\|\cdot\|}$  which by Lemma 2.31 coincides with the space of strong continuity.

#### § 1.5 Well-posedness

There is a common technique to connect strongly continuous semigroups on Banach spaces, their generators and well-posedness of so-called (autonomous) abstract Cauchy problems arsing from partial differential equations. These abstract Cauchy problems are studied for example by Engel and Nagel [52, Chapter II, Sec. 6] or also by Melnikova and Filinkov [87]. For bi-continuous semigroups this concept was also studied by Farkas in [54, Sect. 4.1]. The abstract Cauchy problem for a linear operator A is given by

$$\begin{cases} \dot{u}(t) = Au(t), & t \ge 0, \\ u(0) = x \in D(A). \end{cases}$$
(ACP)

As an example of a abstract Cauchy problem, coming from a partial differential equation, we consider the following equation

$$\begin{cases} \frac{\partial w(t,x)}{\partial t} - \frac{\partial w(t,x)}{\partial x} = 0, \quad t \ge 0, \ x \in \mathbb{R}, \\ w(x,0) = f(x). \end{cases}$$
(1.5.1)

on the space  $C_b(\mathbb{R})$  of bounded continuous functions on  $\mathbb{R}$ . This equation can be rewritten in the form of (ACP) as already described in the introduction, i.e., consider the Banach space  $X := C_b(\mathbb{R})$  and define  $u(t) := w(t, \cdot)$  to be a function with variable x. Furthermore, let (A, D(A)) be the operator defined by

$$A := \frac{\mathrm{d}}{\mathrm{d}x}, \quad D(A) := \mathrm{C}^{1}_{\mathrm{b}}(\mathbb{R}).$$
(1.5.2)

Then the partial differential equation (1.5.1) is of the form (ACP). We now recall the definition of well-posedness for bi-continuous semigroups, cf. [54, Def. 4.1.1], as already promised in Section 1.3.

**Definition 1.14.** The abstract Cauchy problem (ACP) is called *well-posed* if the following three assertions hold

- 1. For all  $x \in D(A)$  there exists a function u(t) := u(t, x) which solves (ACP) and such that  $u \in B_{loc}(\mathbb{R}_{\geq 0}, X_0) \cap C^1(\mathbb{R}_{\geq 0}, (X_0, \tau))$  and  $u' \in B_{loc}(\mathbb{R}_{\geq 0}, X_0)$ , where the differentiation is understood in the vector-valued sense with respect to  $\tau$ .
- 2. The solution of (ACP) is unique.
- 3. The solution u of (ACP) depends continuously on  $x \in D(A)$ , i.e. if  $(x_n)_{n \in \mathbb{N}}$  is a normbounded  $\tau$ -null sequence in D(A), then the solutions  $u_n(t) := u(t, x_n)$  are  $\tau$ -convergent to zero uniformly on compact intervals of  $\mathbb{R}_{>0}$ .

The following result, due to B. Farkas [54, Thm. 4.1.2], now relates generators of bi-continuous semigroups to well-posedness of abstract Cauchy problems. Observe that the previous definition and the following theorem already contains the well-posedness result (at least in one direction) for  $C_0$ -semigroups, cf. [52, Chapter II, Thm. 6.7].

**Theorem 1.15.** If (A, D(A)) generates a bi-continuous semigroup  $(T(t))_{t\geq 0}$ , then the abstract Cauchy problem (ACP) is well-posed. In this case the function u(t) := T(t)x is the solution of (ACP).

In the case of our example the corresponding abstract Cauchy problem is well-posed since the operator (A, D(A)) from (1.5.2) is the generator of the left-translation semigroup on  $C_b(\mathbb{R})$ , see Section 1.3.1.

#### § 1.6 Adjoint bi-continuous semigroups

This section is distinguished from the previous sections even if we consider again adjoint semigroups. The reason is that we now investigate adjoints of bi-continuous semigroups, instead of strongly continuous ones. In fact, the results we present in this Section are due to B. Farkas. For more details and the proofs we refer to the corresponding work [57].

Let us start with a Banach space X and a locally convex topology  $\tau$  satisfying Assumption 1.1. Moreover let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup on X with respect to  $\tau$ . As in the case of adjoints of strongly continuous semigroups, where we considered the sun dual  $X^{\odot}$ , we now examine the subspace  $X^{\circ}$  of the dual space X' consisting of all norm-bounded linear functionals which are  $\tau$ -sequentially continuous on norm-bounded sets of X. As a matter of fact  $X^{\circ}$  is a closed linear subspace of X' and hence a Banach space if equipped with the inherited norm of X'. Furthermore,  $X^{\circ}$  can be equipped with the topology  $\tau^{\circ} := \sigma(X^{\circ}, X)$ . In order to show that  $\tau^{\circ}$  satisfies Assumption 1.1, we have to postulate that  $X^{\circ} \cap \overline{B(0,1)}$  is sequentially complete with respect to  $\sigma(X^{\circ}, X)$ . We remark that this assumption in general does not follow from the general Assumptions 1.1.

**Proposition 1.16.** [57, Prop. 2.3] Let  $B \in \mathscr{L}(X)$  be a norm-bounded linear operator which is  $\tau$ -sequentially continuous on norm-bounded sets. Then the adjoint  $B' \in \mathscr{L}(X')$  leaves  $X^{\circ}$ invariant.

From the previous result we conclude that we can restrict  $(T'(t))_{t\geq 0}$  to the space  $X^{\circ}$ . We denote this restricted semigroup by  $(T^{\circ}(t))_{t\geq 0}$ . To conclude that  $(T^{\circ}(t))_{t\geq 0}$  is bi-continuous on  $X^{\circ}$  with respect to  $\tau^{\circ}$  we again have to impose a additional hypothesis. Especially, we have to assume that every norm-bounded  $\tau^{\circ}$ -null sequence  $(\varphi_n)_{n\in\mathbb{N}}$  in  $X^{\circ}$  is  $\tau$ -equicontinuous on norm bounded sets.

Let us continue with some examples. Actually, we consider semigroups  $(T(t))_{t\geq 0}$  on the Banach space  $X := C_b(\Omega)$ , where  $\Omega$  is a Polish space, which are bi-continuous with respect to the compact-open topology  $\tau_{co}$ . In [57, Sect. 3] B. Farkas illustrated that  $C_b(\Omega)^\circ$  coincides with  $M(\Omega)$ , the space of bounded Baire measures. The following results connect bi-continuous semigroups on  $C_b(\Omega)$  with these on  $M(\Omega)$ .

**Theorem 1.17.** [57, Thm. 3.5] Let  $\Omega$  be a Polish space and  $(T(t))_{t\geq 0}$  bi-continuous on  $C_b(\mathbb{R})$ with respect to  $\tau_{co}$ . Then the semigroup  $(T^{\circ}(t))_{t\geq 0}$  defined as  $T^{\circ}(t) := T'(t)_{|M(\Omega)}, t \geq 0$ , is a bi-continuous semigroup on  $M(\Omega)$  with respect to  $\tau^{\circ}$ .

Counter-intuitively the converse of the previous theorem also holds true.

**Theorem 1.18.** [57, Thm. 3.6] Let  $\Omega$  be a Polish space. Let  $(S(t))_{t\geq 0}$  be a bi-continuous semigroup on  $M(\Omega)$  with respect to  $\tau^{\circ}$ . Then there exists a semigroup  $(T(t))_{t\geq 0}$  on  $C_{b}(\mathbb{R})$  which is bi-continuous with respect to  $\tau_{co}$  and such that  $T^{\circ}(t) = S(t)$  for all  $t \geq 0$ 

The following result yields a characterization for the generator of adjoints of bi-continuous semigroups. The result is related to Proposition 1.10. In particular, the proof of the result is the same.

**Lemma 1.19.** Let  $(T(t))_{t\geq 0}$  be bi-continuous on X with respect to  $\tau$  and assume that the additional hypothesis on  $X^{\circ}$  and  $\tau^{\circ}$  hold. Let us denote the generator of  $(T^{\circ}(t))_{t\geq 0}$  by  $(A^{\circ}, D(A^{\circ}))$ . Then

 $D(A^{\circ}) = \left\{ x' \in X^{\circ} : \exists y' \in X^{\circ} \ \forall x \in D(A) : \langle Ax, x' \rangle = \langle x, y' \rangle \right\}, \quad A^{\circ}x' := y'.$ 

#### § 1.7 Approximation theorems

In this section we recall the convergence of sequences of bi-continuous semigroups  $(T_n(t))_{t\geq 0}$ , since we need them later in Chapter 7. For  $C_0$ -semigroup this topic is for example treated in [52, Chapter III, Sect. 4]. The original results trace back to the papers by H. Trotter [114] and T. Kato [71]. For the bi-continuous case we need the following definition. **Definition 1.20.** For each  $n \in \mathbb{N}$  let  $(T_n(t))_{t\geq 0}$  be a bi-continuous semigroup on a Banach space  $X_0$  with respect to  $\tau$ . We say that a sequence  $(T_n(t))_{t\geq 0}$  is uniformly bi-continuous (of type  $\omega$ ) if the following conditions hold:

- 1. There exists  $M \ge 1$  and  $\omega \in \mathbb{R}$  such that  $||T_n(t)|| \le M e^{t\omega}$  for each  $t \ge 0$  and  $n \in \mathbb{N}$ .
- 2. For every  $t_0 > 0$  and for every  $\|\cdot\|$ -bounded  $\tau$ -null sequence  $(x_n)_{n \in \mathbb{N}}$  in  $X_0$  we have

$$\tau \lim_{n \to \infty} T_k(t) x_n = 0,$$

uniformly for  $t \in [0, t_0]$  and  $k \in \mathbb{N}$ .

Next we formulate the Trotter–Kato Approximation Theorems for bi-continuous semigroups, see [78, Thm. 2.3 & Thm. 2.6], [54, Thm. 1.2.10] and [4]. Even if only Theorem 1.22 is needed in what follows we stay self-contained and also recall Theorem 1.21.

**Theorem 1.21** (First Trotter–Kato Approximation Theorem). Let  $(T_n(t))_{t\geq 0}$ ,  $n \in \mathbb{N}$ , and  $(T(t))_{t\geq 0}$  be uniformly bi-continuous semigroups (of type  $\omega$ ) with generator  $(A_n, D(A_n))$  and (A, D(A)), respectively and let D be a  $\|\cdot\|$ -dense subset of  $\overline{D(A)}^{\|\cdot\|}$ . If

$$R(\lambda, A_n) x \xrightarrow{\|\cdot\|} R(\lambda, A) x, \quad n \to \infty,$$

for each  $x \in D$  and some  $\lambda > \omega$ , then

$$T_n(t)x \xrightarrow{\tau} T(t)x, \quad n \to \infty,$$

for all  $x \in X_0$  uniformly for t in compact intervals of  $\mathbb{R}_{\geq 0}$ .

**Theorem 1.22** (Second Trotter–Kato Approximation Theorem). For  $n \in \mathbb{N}$  let  $(T_n(t))_{t\geq 0}$  be uniformly bi-continuous semigroups (of type  $\omega$ ) on  $X_0$  with generators  $(A_n, D(A_n))$ . Consider the following assertions.

- (a) There exists a bi-densely defined operator (A, D(A)) such that  $A_n x \to Ax$  for all x in a bi-core of A and such that  $\operatorname{Ran}(\lambda_0 A)$  is bi-dense in  $X_0$ .
- (b) There exists an operator  $R \in \mathscr{L}(X_0)$  such that  $R(\lambda_0, A_n) x \xrightarrow{\|\cdot\|} Rx$  for all x in a subset of  $\operatorname{Ran}(R)$  which is bi-dense in  $X_0$ .
- (c) There exists a bi-continuous semigroups  $(T(t))_{t\geq 0}$  with generator (B, D(B)) such that  $T_n(t)x \xrightarrow{\tau} T(t)x$  for all  $x \in X_0$  uniformly for t in compact intervals

Then the implications (a)  $\Rightarrow$  (b)  $\Rightarrow$  (c) hold. In particular, if (a) holds, then  $B = \overline{A}^{\tau}$  (the bi-closure of A).

**Remark 1.23.** In the proof of [78, Thm. 2.6] one observes that operator R in assertion (b) gives rise to a pseudo-resolvent that is used to define operator (B, D(B)) in assertion (c).

#### § 1.8 Notes

The (left)-translation semigroup on  $C_b(\mathbb{R})$  is going to accompany us through out this thesis. Actually, one can find more information in combination with different topics of this thesis in Section 2.5.1 and Section 4.3. The results of Section 1.6 about adjoints of bi-continuous semigroups will play an important role in Section 5.3.2 when we consider the Gauss–Weierstrass semigroup on the space of bounded Borel measures on  $\mathbb{R}$ . Likewise we use the approximation theory from Section 1.7 in Chapter 7 in connection with flows on networks.

The use of bi-continuous semigroups can also be motivated by the work of H. Lotz [83]. He proved that on certain Banach spaces, every strongly continuous semigroup is uniformly continuous, i.e., the generator of the semigroups are bounded operators. Typical examples are  $L^{\infty}(\Omega)$  where  $(\Omega, \Sigma, \mu)$  is a measure space, and C(X) where X is either a compact  $\sigma$ -Stonian space or a compact F-space, just to mention a few. For this reason, strong continuity with respect to the norm is not a fruitful property on these spaces and hence another variant of strong continuity needs to be considered.

## Chapter 2

# Intermediate and Extrapolated Spaces

#### Introduction

Extrapolation spaces for generators of  $C_0$ -semigroups (used here synonymously to "strongly continuous, one-parameter semigroups of bounded linear operators") on Banach spaces, or for more general operators, have been designed to study e.g., maximal regularity questions by Da Prato and Grisvard [37]; see also Walter [122], Amann [7], van Neerven [116], Nagel, Sinestrari [94], Nagel [92], Sinestrari [104], Magal, Ruan [86, Ch. 3]. These spaces (and the corresponding extrapolated operators) play a central role in recent abstract perturbation results, most prominently in boundary-type or domain perturbations, see e.g., Desch, Schappacher [38], Greiner [60], Staffans, Weiss [109], Adler, Bombieri, Engel [3], Hadd, Manzo, Rhandi [63]. Extrapolation spaces are also important in the theory of coupled operator matrices, see Engel [46].

In this chapter, we concentrate on the construction of extrapolation spaces for linear operators having a non-empty resolvent set on a Banach space, but we do not assume the operator to fulfill the Hille–Yosida conditions or to be densely defined. In case the operator is densely defined such a construction is known from the seminal papers of Da Prato, Grisvard, [36], Amann [7] and Nagel, Sinestrari [94]. In the case of non-densely defined, sectorial operators there is a very general—almost purely algebraic—construction due to Haase [62] leading also to universal extrapolation spaces. Here, we present a slightly different construction of extrapolation and extrapolated Favard spaces, allowing the construction of extrapolated semigroups in the absence of strong continuity with respect to the norm. For a non-densely defined Hille–Yosida operator A on the Banach space  $X_0$  such a construction is possible by taking the part of A in  $\underline{X}_0 := D(A)$ , so that the restricted operator becomes the generator of a  $C_0$ semigroup on  $\underline{X}_0$ , thus leading to an extrapolated semigroup on the extrapolation space  $\underline{X}_{-1}$ , see Nagel, Sinestrari [95]. But this semigroup will usually not leave the original Banach space  $X_0$  invariant. This is why we restrict our attention to the situation where strong continuity of the semigroup is guaranteed with respect to some coarser locally convex topology  $\tau$  on  $X_0$ . Here the framework of bi-continuous semigroups, or that of Saks spaces, (see Kühnemund [79] and Section 2.4 below) appears to be adequate. However, most of the results presented here are valid also for generators of other classes of semigroups: integrable semigroups of Kunze [80], "C-class" semigroups of Kraaij [75],  $\pi$ -semigroups of Priola [100], weakly continuous semigroups of Cerrai [32], to mention a few.

Given a Banach space  $X_0$  and a Hausdorff locally convex topology  $\tau$  on  $X_0$  (with certain properties described in Section 2.4), and a bi-continuous semigroup  $(T(t))_{t\geq 0}$  with generator A, we construct the full scale of abstract Sobolev (or Hölder) and Favard spaces  $X_{\alpha}, \underline{X}_{\alpha}, F_{\alpha}$  for  $\alpha \in \mathbb{R}$ , and the corresponding extrapolated semigroups  $(T_{\alpha}(t))_{t\geq 0}$ . (If  $\tau$  is the norm topology, there is nothing new here, and everything can be found in [52, Section II.5].) These constructions, along with some applications, form the main content of this chapter. Here we illustrate the results on the following well-known example (see also Nagel, Nickel, Romanelli [93] and Section 2.5 for details): Consider the Banach space  $X_0 := C_b(\mathbb{R})$  of bounded, continuous functions and the (left)-translation semigroup  $(S(t))_{t\geq 0}$  thereon, defined by (S(t)f)(x) = f(x+t),  $x \in \mathbb{R}, t \geq 0, f \in X_0$ . For  $\alpha \in (0, 1)$  we have the continuous embeddings

$$C^1_b(\mathbb{R}) \hookrightarrow \operatorname{Lip}_b(\mathbb{R}) \hookrightarrow h^{\alpha}_b(\mathbb{R}) \hookrightarrow h^{\alpha}_{b, \operatorname{loc}}(\mathbb{R}) \hookrightarrow C^{\alpha}_b(\mathbb{R}) \hookrightarrow \operatorname{UC}_b(\mathbb{R}) \hookrightarrow \operatorname{C}_b(\mathbb{R}) \hookrightarrow \operatorname{L}^{\infty}(\mathbb{R}),$$

where  $C_b^1(\mathbb{R})$  is the space of differentiable functions with derivative in  $C_b(\mathbb{R})$ ,  $Lip_b(\mathbb{R})$  is the space of bounded, Lipschitz functions,  $h_b^{\alpha}(\mathbb{R})$  is the space of bounded, little-Hölder continuous functions,  $h_{b,loc}^{\alpha}(\mathbb{R})$  is the space of bounded, locally little-Hölder continuous functions,  $C_b^{\alpha}(\mathbb{R})$  is the space of bounded, Hölder continuous functions,  $UC_b(\mathbb{R})$  is the space of bounded, Hölder continuous functions,  $UC_b(\mathbb{R})$  is the space of bounded, uniformly continuous functions. In the abstract perspective and using the notation in this chapter, this corresponds to the inclusions of Banach spaces:

$$X_1 \hookrightarrow F_1 \hookrightarrow \underline{X}_{\alpha} \hookrightarrow X_{\alpha} \hookrightarrow F_{\alpha} \hookrightarrow \underline{X}_0 \hookrightarrow X_0 \hookrightarrow F_0.$$

The extension of the previous diagram for the full scale  $\alpha \in \mathbb{R}$  is possible by extrapolation. The (abstract) spaces  $\underline{X}_{\alpha}$  and  $F_{\alpha}$  ( $\alpha \in (0, 1)$ ) are well studied and we refer to the books by Lunardi [85] and Engel, Nagel [52, Section II.5] for a systematic treatment. However, the definition of  $X_{\alpha}$  is new, and requires a recollection of results concerning the other spaces,  $\underline{X}_{\alpha}$ and  $F_{\alpha}$ .

Extrapolated Favard spaces are not only important for perturbation theory. They help to reduce problems concerning semigroups being not strongly continuous to the study of an underlying  $C_0$ -semigroup. This perspective is propagated by Nagel and Sinestrari in [95]: To any Hille–Yosida operator on  $X_0$  one can construct a Banach space  $F_0$  (the Favard class) containing  $X_0$  as a closed subspace, and a semigroup  $(T(t))_{t\geq 0}$  on  $F_0$ . (Note, however, that the semigroup  $(T(t))_{t\geq 0}$  defined on  $F_0$  may not leave  $X_0$  invariant.) We adapt this point of view also in this chapter. In particular, we provide an alternative (and short) proof of the Hille–Yosida type generation theorem for bi-continuous semigroups (due to Kühnemund [79]) by employing solely the  $C_0$ -theory.

Applications of the Sobolev (Hölder) scale, as presented here, to perturbation theory, in the spirit of the results of Desch, Schappacher [38], or of Jacob, Wegner, Wintermayr [70], will be presented in Chapter 4.

This chapter is organized as follows: In Section 2.1 we recall the standard constructions and results for extrapolation spaces for densely defined (invertible) operators. Moreover, we construct extrapolation spaces for not densely defined operators A with  $D(A^2)$  dense in D(A)for the norm of  $X_0$ . Our argument differs form the one in Haase [62] in that we build the space  $X_{-1}$  based on  $\underline{X}_{-2}$  (which, in turn, arises from  $\underline{X}_0$  and  $\underline{X}_{-1}$ ), i.e., in a bottom-to-top and then back-to-bottom manner, resulting in the continuous inclusions

$$\underline{X}_0 \hookrightarrow X_0 \hookrightarrow \underline{X}_{-1} \hookrightarrow X_{-1} \hookrightarrow \underline{X}_{-2}.$$

(None of these inclusions is surjective in general.) This approach becomes convenient when we compare the arising extrapolation spaces  $X_{-1}$  and  $\underline{X}_{-1}$  and construct the extrapolated semigroups thereon. In Section 2.2 we turn to intermediate spaces; the results there are classical, but are put in the general perspective of this chapter. We also present a method for a "concrete" representation of extrapolation spaces (see Theorem 2.16). Section 2.3 discusses the Sobolev (Hölder) scale for semigroup generators. In Section 2.4 we recall the concept of bi-continuous semigroups, construct the corresponding extrapolated semigroups and give a direct proof of the Hille–Yosida generation theorem (due to Kühnemund, see [79]) using extrapolation techniques. We conclude this chapter with some examples in Section 2.5, where we determine the extrapolation spaces of concrete semigroup generators. In particular, we discuss the previously mentioned example of the translation semigroup (complementing results of Nagel, Nickel, Romanelli [93, Sec. 3.1, 3.2]) and then *left implemented semigroups* (cf. Alber [6]).

#### § 2.1 Sobolev and extrapolation spaces for invertible operators

In this section we construct abstract Sobolev (Hölder) and extrapolation spaces (the so-called Sobolev scale) for a boundedly invertible linear operator defined on a Banach space. Some of the results are well-known and even standard, but we chose to include them here for the sake of completeness, and because they are needed for the construction of spaces when we deal with not densely defined operators. The emphasis will be, however, on this latter case, when the construction is new, see Section 2.1.2 below. We also note that everything what follows is also valid for operators on Fréchet spaces.

The following is a standing assumption in this chapter.

Assumption A. We suppose that  $A : D(A) \to X_0$  is a (not necessarily densely defined) linear operator on a Banach space  $X_0$  with 0 in the resolvent set  $\rho(A)$ .

As a matter of fact, it is only for convenience to suppose  $0 \in \rho(A)$  instead of  $\rho(A) \neq \emptyset$ . Indeed, if  $\lambda \in \rho(A)$  we may consider  $A - \lambda$  and carry out the constructions for this new operator satisfying  $0 \in \rho(A - \lambda)$ . The arising spaces will not depend on  $\lambda \in \rho(A)$  (up to isomorphism).

#### 2.1.1 Abstract Sobolev spaces

The material presented here is standard, see Nagel [91], Nagel, Nickel, Romanelli [93] or Engel, Nagel [52, Section II.5], and some parts are valid even for operators on locally convex spaces, when one has to argue with a family of generating seminorms instead of one norm. We set  $X_1 := D(A)$  which becomes a Banach space if endowed with the graph norm

$$|x||_A := ||x|| + ||Ax||$$

An equivalent norm is given by  $||x||_{X_1} := ||Ax||$  since we have assumed  $0 \in \rho(A)$ . Then we have the isometric isomorphism

$$A: X_1 \to X_0$$
 with inverse  $A^{-1}: X_0 \to X_1$ .

**Definition 2.1.** Recall the assumption that  $0 \in \rho(A)$ , and take  $n \in \mathbb{N}$ ,  $n \ge 1$ .

1. We define

$$X_n := D(A^n)$$
 and  $||x||_{X_n} := ||A^n x||$  for  $x \in X_n$ 

If we want to stress the dependence on A, then we write  $X_n(A)$  and  $\|\cdot\|_{X_n(A)}$ .

2. Let

$$X_{\infty}(A) := \bigcap_{n \in \mathbb{N}} X_n,$$

often abbreviated as  $X_{\infty}$ .

3. We further set

$$\underline{X}_0 := \overline{D(A)}, \quad \underline{A} := A|_{\underline{X}_0},$$

the part of A in  $\underline{X}_0$ , i.e.,

$$D(\underline{A}) = \{ x \in D(A) : Ax \in \underline{X}_0 \}.$$

Moreover, we let

$$\underline{X}_n := D(\underline{A}^n), \quad \|x\|_{\underline{X}_n} := \|\underline{A}^n x\|.$$

To be specific about the underlying operator A we write  $\underline{X}_n(A)$  and  $||x||_{\underline{X}_n(A)}$ .

- 4. For  $n \in \mathbb{N}$  we set  $A_n := A|_{X_n}$ , the part of A in  $X_n$ , in particular  $A_0 = A$ . Similarly, we let  $\underline{A}_n := \underline{A}|_{\underline{X}_n}$ , for example  $\underline{A}_0 = \underline{A}$ . By this notation we also understand implicitly that the surrounding space is  $X_n(A)$  respectively  $\underline{X}_n(A)$  with its norm, see Remark 2.2.
- **Remark 2.2.** (i) By "underlining" we always indicate an object which is in some sense smaller than the one without underlining. The space  $\underline{X}_0(A)$  is connected with the domain of D(A), and the whole issue of distinguishing between  $X_0$  and  $\underline{X}_0$  becomes relevant only if A is not densely defined but its part <u>A</u> is (cf. Remark 2.6). We keep to the notation <u>A</u> for the part of the operator A instead of  $A|_{X_0}$ .
- (ii) If A is densely defined, then  $X_n(A) = \underline{X}_n(A)$  for each  $n \in \mathbb{N}$ . In particular, if  $\underline{X}_1(A) = D(\underline{A})$  is dense in  $\underline{X}_0(A)$ , then  $\underline{X}_n(A) = \underline{X}_n(\underline{A})$  for each  $n \in \mathbb{N}$ .
- (iii) For  $n \in \mathbb{N}$  we evidently have  $X_1(A^n) = X_n(A)$ . Also  $\underline{X}_1(A^n) = \underline{X}_n(A)$  holds, because  $D(\underline{A}^n) = D(\underline{A}^n)$ . Indeed, the inclusion " $D(\underline{A}^n) \subseteq D(\underline{A}^n)$ " is trivial. While for  $x \in D(\underline{A}^n)$  we have  $x \in \underline{X}_0$  and  $A^n x \in \underline{X}_0$ , implying  $A^{n-1}x \in D(\underline{A})$ , and then recursively  $x \in D(\underline{A}^n)$ .
- (iv) For  $x \in D(A_n) = D(A^{n+1})$  we have  $||x||_{X_1(A_n)} = ||A_nx||_{X_n(A)} = ||A^{n+1}x|| = ||x||_{X_{n+1}(A)}$ . Similarly  $D(\underline{A}_n) = D(\underline{A}^{n+1})$ .

In order to prove our next result we recall that a projective system  $(E_n, \theta_n)_{n \in \mathbb{N}}$  consists of a countable family of sets  $(E_n)_{n \in \mathbb{N}}$  and maps  $\theta_n : E_{n+1} \to E_n$ . The corresponding projective
limit, denoted by  $\varprojlim_{n \in \mathbb{N}} (E_n, \theta_n)$ , is the set of all elements  $x = (x_n)_{n \in \mathbb{N}}$  of the cartesian product  $\prod_{n \in \mathbb{N}} E_n$  satisfying  $x_n = \theta_n(x_{n+1})$  for each  $n \in \mathbb{N}$ . With  $\pi_m$  we denote the *m*-th coordinate projection  $\pi_m : \prod_{n \in \mathbb{N}} E_n \to E_m$ . The following Mittag–Leffler type result is due to J. Esterle.

**Theorem 2.3.** [53, Cor. 2.2] Let  $(E_n, \theta_n)_{n \in \mathbb{N}}$  be a projective system, where  $E_n$  is a complete metric space and  $\theta_n : E_{n+1} \to E_n$  continuous for each  $n \in \mathbb{N}$ . If  $\theta_n(E_{n+1})$  is dense in  $E_n$  for each  $n \in \mathbb{N}$ , then  $\pi_m\left(\lim_{m \to \infty} (E_n, \theta_n)\right)$  is dense in  $E_m$  for each  $m \in \mathbb{N}$ 

One can the statement of this theorem reformulate for our purposes as follows: For all  $x \in E_1$ and all  $\varepsilon > 0$  there exists  $x_n \in E_n$   $(n \in \mathbb{N})$  such that  $x_n = \theta_n(x_{n+1})$  and  $d_1(x, x_1) < \varepsilon$ , where  $d_1$  denotes the metric on  $E_1$ .

**Proposition 2.4.** Suppose  $\underline{A}$  is densely defined in  $\underline{X}_0$ .

- (a) For  $n \in \mathbb{N}$  the mappings  $A^n : X_n \to X_0$  and  $\underline{A}^n : \underline{X}_n \to \underline{X}_0$  are isometric isomorphisms.
- (b) For  $n \in \mathbb{N}$  the operators  $A_n : X_{n+1} \to X_n$  and  $\underline{A}_n : \underline{X}_{n+1} \to \underline{X}_n$  are isometric isomorphisms that intertwine  $A_{n+1}$  and  $A_n$ , respectively,  $\underline{A}_{n+1}$  and  $\underline{A}_n$ .
- (c) If  $D(\underline{A})$  is dense in  $\underline{X}_0$ , then  $X_\infty$  is dense in  $\underline{X}_n$  for each  $n \in \mathbb{N}$ . As a consequence,  $\underline{X}_m$  is dense in  $\underline{X}_n$  for each  $m, n \in \mathbb{N}$  with  $m \ge n$ .

Part (a) and (b) of Proposition 2.4 are trivial by construction. In order to prove the third statement We use the statement of Theorem 2.3 to prove part (c) of Proposition 2.4, which was originally proven by Arendt, El-Mennaoui and Kéyantuo [9, Thm. 6.2].

**Proof of Proposition 2.4**(c). Owing to the fact that  $\underline{A}$  is densely defined in  $\underline{X}_0$  we are able to use Theorem 2.3. The space  $D(\underline{A}^n)$  equipped with the norm  $\|\cdot\|_{\underline{X}_n}$  is a Banach space. Let  $\theta_n: D(\underline{A}^{n+1}) \hookrightarrow D(\underline{A}^n)$  be the natural (continuous) inclusion. For  $\lambda \in \rho(\underline{A})$  is  $(\lambda - \underline{A})^n$  an isomorphism from  $D(\underline{A}^n)$  to  $\underline{X}_0$  which maps  $D(\underline{A}^{n+1})$  onto  $D(\underline{A})$ . Since  $D(\underline{A})$  was supposed to be dense in  $\underline{X}_0$  also  $D(\underline{A}^{n+1})$  is dense in  $D(\underline{A}^n)$ . Let  $x \in \underline{X}_0$  and  $\varepsilon > 0$  be arbitrary. By Theorem 2.3 there exists  $x_n \in D(\underline{A}^n)$  for each  $n \in \mathbb{N}$  such that  $x_n = \theta_n(x_{n+1})$  and  $\|x - x_1\| < \varepsilon$ . Thus  $x_n = x_1$  for each  $n \in \mathbb{N}$  and hence  $x_1 \in X_\infty$ . The last part of the statement follows by replacing X by  $D(\underline{A}^m)$  and  $\underline{A}$  by the part of  $\underline{A}$  in  $D(\underline{A}^m)$ .

**Remark 2.5.** We note that the Mittag–Leffler type result Theorem 2.3 is valid in complete metric spaces. Hence the statements (a), (b) and (c) are all remain true for Fréchet spaces with verbatim the same proof as in [9].

Henceforth, another standing assumption will be the following (though not everywhere needed).

Assumption B. The operator  $\underline{A} := A|_{\underline{X}_0} : D(\underline{A}) \to \underline{X}_0$  is densely defined, i.e.,

$$\overline{D(\underline{A})} = \underline{X}_0$$

**Remark 2.6.** The condition of  $D(\underline{A})$  being dense in  $\underline{X}_0$  holds for example if there are  $M, \omega > 0$  such that  $(\omega, \infty) \subseteq \rho(A)$  and

$$\|\lambda R(\lambda, A)\| \le M \quad \text{for all } \lambda > \omega. \tag{2.1.1}$$

Indeed, in this case we have for  $x \in D(A)$ 

$$\|\lambda R(\lambda, A)x - x\| = \|R(\lambda, A)Ax\| \le \frac{M\|Ax\|}{\lambda} \to 0 \quad \text{for } \lambda \to \infty.$$

Hence  $D(A^2) \subseteq D(\underline{A})$  is dense in D(A) for the norm of  $X_0$ , and this implies the density of  $D(\underline{A})$  in  $\underline{X}_0$ . An operator A satisfying (2.1.1) is often said to have a ray of minimal growth, see, e.g., [85, Chapter 3], and also Section 2.2 below. Another term used is "weak Hille–Yosida operator".

**Proposition 2.7.** If  $T \in \mathscr{L}(X_0)$  is a linear operator commuting with  $A^{-1}$ , then the spaces  $X_n$  and  $\underline{X}_n$  are T-invariant, and  $T \in \mathscr{L}(X_n)$  for  $n \in \mathbb{N}$ .

*Proof.* The condition means that  $Tx \in D(A)$  for each  $x \in D(A)$  and for such x we have ATx = TAx. This implies the invariance of  $X_1$  and that  $||Tx||_{X_1(A)} \leq ||T|| ||x||_{X_1(A)}$ . Using the boundedness assumption we see that  $\underline{X}_1$  remains invariant under T. For general  $n \in \mathbb{N}$  we may argue by recursion, or simply invoke Remark 2.2.

#### 2.1.2 Extrapolation spaces

The construction for the extrapolation spaces here is standard if A is densely defined, or if A is a Hille–Yosida operator, see, e.g., [95]. For  $x \in X_0$  we define  $||x||_{\underline{X}_{-1}(A)} := ||A^{-1}x||$ . Then the surjective mapping

$$A: (D(A), \|\cdot\|) \to (X_0, \|\cdot\|_{X_{-1}(A)})$$

becomes isometric, and hence has a uniquely continuous extension

$$\underline{A}_{-1}: (\underline{X}_0, \|\cdot\|) \to (\underline{X}_{-1}, \|\cdot\|_{\underline{X}_{-1}(\underline{A})}),$$

which is an isometric isomorphism, where  $(\underline{X}_{-1}, \|\cdot\|_{\underline{X}_{-1}(\underline{A})})$  denotes the completion of  $(\underline{X}_0, \|\cdot\|_{\underline{X}_{-1}(A)})$ . By construction we obtain immediately:

**Proposition 2.8.** The space  $X_0$  is continuously and densely embedded in  $\underline{X}_{-1}$ . If  $\underline{A}$  is densely defined in  $\underline{X}_0$ , then also  $X_\infty$  is dense in  $\underline{X}_{-1}$ . As a consequence  $(\underline{X}_{-1}, \|\cdot\|_{\underline{X}_{-1}(\underline{A})})$  is the completion of  $(\underline{X}_0, \|\underline{A}^{-1} \cdot \|)$ .

*Proof.* The space  $X_0$  is dense in  $\underline{X}_{-1}$  by construction. For  $x \in X_0$  we have

 $\|x\|_{\underline{X}_{-1}(\underline{A})} = \|AA^{-1}x\|_{\underline{X}_{-1}(\underline{A})} = \|\underline{A}_{-1}A^{-1}x\|_{\underline{X}_{-1}(A)} \le \|\underline{A}_{-1}\| \cdot \|A^{-1}x\| \le \|\underline{A}_{-1}\| \cdot \|A^{-1}\| \cdot \|x\|,$ 

showing the continuity of the embedding. The last assertion follows since  $X_{\infty}$  is dense in D(A) with respect to  $\|\cdot\|$ .

Of course one can iterate the whole procedure and obtain the following chain of dense and continuous embeddings

$$\underline{X}_0 \hookrightarrow \underline{X}_{-1} \hookrightarrow \underline{X}_{-2} \hookrightarrow \dots \hookrightarrow \underline{X}_{-n} \quad \text{for } n \in \mathbb{N},$$

where for  $n \ge 1$  the space  $\underline{X}_{-n}$  is a completion of  $\underline{X}_{-n+1}$  with respect to the norm  $\|\cdot\|_{\underline{X}_{-n}(\underline{A})}$  defined by  $\|x\|_{\underline{X}_{-n}(\underline{A})} = \|\underline{A}_{-n+1}^{-1}x\|_{\underline{X}_{-n+1}(\underline{A})}$  and

$$\underline{A}_{-n}: \underline{X}_{-n+1} \to \underline{X}_{-n}$$

is a unique continuous extension of  $\underline{A}_{-n+1}: D(\underline{A}_{-n+1}) \to \underline{X}_{-n+1}$  to  $\underline{X}_{-n}$ .

These spaces, just as well the ones in the next definition, are called *extrapolation spaces* for the operator A, see, e.g., [95] or [52, Section II.5] for the case of semigroup generators. The spaces  $\underline{X}_{-1}, \underline{X}_{-2}$  and the operator  $\underline{A}_{-2}$  will be used to define the extrapolation space  $X_{-1}(A)$ . To this purpose we identify  $X_0$  with a subspace of  $\underline{X}_{-1}$  and of  $\underline{X}_{-2}$ .

**Definition 2.9.** Consider  $X_0$  as a subspace of  $X_{-2}$ , and define

$$X_{-1} := \underline{A}_{-2}(X_0) := \{ \underline{A}_{-2}x : x \in X_0 \} \text{ and } \|x\|_{X_{-1}} := \|\underline{A}_{-2}^{-1}x\|.$$

Furthermore, we set  $D(A_{-1}) := X_0$  and for  $x \in X_0$  we define  $A_{-1}x := \underline{A}_{-2}x$ . To note the dependence on the operator A we write  $X_{-1}(A)$  and  $\|\cdot\|_{X_{-1}(A)}$ .

**Remark 2.10.** It is easy to see that the operator  $A_{-1}$  is the part of  $\underline{A}_{-2}$  in  $X_{-1}$ .

In what follows, we will define higher order extrapolation spaces and prove that all these spaces line up in a scale, where one can switch between the levels with the help of (a version) of the operator A (or  $A_{-1}$ ).

**Proposition 2.11.** The operator  $A_{-1}$  is an extension of  $\underline{A}_{-1}$ ,  $(X_{-1}, \|\cdot\|_{X_{-1}})$  is a Banach space, the norms of  $\underline{X}_{-1}$  and  $X_{-1}$  coincide on  $\underline{X}_{-1}$ , and  $\underline{X}_{-1}$  is a closed subspace of  $X_{-1}$ . The mapping  $A_{-1}: X_0 \to X_{-1}$  is an isometric isomorphism.

*Proof.* The first assertion is true because  $\underline{A}_{-2}$  is an extension of  $\underline{A}_{-1}$ . That  $X_{-1}$  is a Banach space is immediate from the definition. Since  $\underline{A}_{-2}^{-1}\underline{A}_{-1} = I$  on  $\underline{X}_0$ , we have  $\underline{A}_{-1}^{-1}x \in \underline{X}_0 \subseteq X_0$  for  $x \in \underline{X}_{-1}$ , so that  $\|\underline{A}_{-2}^{-1}x\| = \|\underline{A}_{-2}^{-1}\underline{A}_{-1}\underline{A}_{-1}^{-1}x\| = \|\underline{A}_{-1}^{-1}x\| = \|x\|_{\underline{X}_{-1}}$ . This establishes that the norms coincide. Since  $\underline{X}_{-1}$  is a Banach space (with its own norm), it is a closed subspace of  $X_{-1}$ . That  $A_{-1}$  is an isometric isomorphism follows from the definition.

**Remark 2.12.** By construction we have  $\underline{X}_{-1}(\underline{A}_{-n}) = \underline{X}_{-(n+1)}(\underline{A})$  as well as  $X_{-1}(A_{-n}) = X_{-(n+1)}(A)$  for each  $n \in \mathbb{N}$ .

**Proposition 2.13.** For  $n \in \mathbb{Z}$  the operators  $A_n : X_{n+1} \to X_n$  and  $\underline{A}_n : \underline{X}_{n+1} \to \underline{X}_n$  are isometric isomorphisms that intertwine  $A_{n+1}$  and  $A_n$ , respectively,  $\underline{A}_{n+1}$  and  $\underline{A}_n$ .

*Proof.* For  $n \in \mathbb{N}$  this is Proposition 2.13. So we assume n < 0. For n = -1 the statement about isometric isomorphisms is just the definition, and the intertwining property is also evident. By recursion we obtain the validity of the assertion for general  $n \leq -1$  and for the operator  $\underline{A}_n$ . By Remark 2.12 it suffices to prove that  $A_{-1}$  intertwines  $A_{-1}$  and  $A_0 = A$ . For  $x \in D(A_0) = D(A)$  we have  $A_{-1}x \in X_0 = D(A_{-1})$  and  $Ax = A_{-1}^{-1}A_{-1}A_{-1}x$ .

Thus for  $n \in \mathbb{N}$  we have the following chain of embeddings (continuous and dense, denoted by  $\hookrightarrow$ ) and inclusions as closed subspaces (denoted by  $\subseteq$ ):

$$\cdots \hookrightarrow \underline{X}_n \subseteq X_n \hookrightarrow \underline{X}_0 \subseteq X_0 \hookrightarrow \underline{X}_{-1} \subseteq X_{-1} \hookrightarrow \underline{X}_{-2} \subseteq X_{-2} \hookrightarrow \cdots \underline{X}_{-n} \subseteq X_{-n} \hookrightarrow \cdots,$$

where in general the inclusions are strict (see the examples in Section 2.5). We also have the following chain of isometric isomorphisms

$$\cdots \longrightarrow \underline{X}_{n+1} \xrightarrow{\underline{A}_n^{-1}} \underline{X}_n \longrightarrow \cdots \longrightarrow \underline{X}_1 \xrightarrow{\underline{A}_0^{-1}} \underline{X}_0 \xrightarrow{\underline{A}_{-1}^{-1}} \underline{X}_{-1} \longrightarrow \cdots \longrightarrow \underline{X}_{-n+1} \xrightarrow{\underline{A}_{-n}^{-1}} \underline{X}_{-n} \longrightarrow \cdots$$

and

$$\cdots \longrightarrow X_{n+1} \xrightarrow{A_n^{-1}} X_n \longrightarrow \cdots \longrightarrow X_1 \xrightarrow{A_0^{-1}} X_0 \xrightarrow{A_{-1}^{-1}} X_{-1} \longrightarrow \cdots \longrightarrow X_{-n+1} \xrightarrow{A_{-n}^{-1}} X_{-n} \longrightarrow \cdots$$

**Proposition 2.14.** (a)  $\underline{X}_1(\underline{A}_{-1}) = \underline{X}_0$  and  $X_1(A_{-1}) = X_0$  with the same norms.

- (b)  $\underline{X}_{-1}(\underline{A}_1) = \underline{X}_0$  with the same norms.
- (c)  $(\underline{A}_1)_{-1} = \underline{A}$ .
- (d)  $X_{-1}(A_1) = X_0$  with the same norms, and  $(A_1)_{-1} = A$ .

*Proof.* (a) By definition  $X_1(A_{-1}) = D(A_{-1}) = X_0$  with the graph norm of  $A_{-1}$ . Since  $A_{-1}$  extends A, for  $x \in X_0$  we have  $||A_{-1}x||_{X_{-1}(A)} = ||Ax||_{X_{-1}} = ||A^{-1}Ax|| = ||x||$ . The first statement then follows, because  $\underline{X}_1(\underline{A}_{-1}) = \overline{X}_1(\underline{A}_{-1}) = \overline{D(\underline{A})} = \underline{X}_0$  with the same norms.

(b) For  $x \in \underline{X}_1(\underline{A}) = D(\underline{A}^2)$  we have

$$\|x\|_{\underline{X}_{-1}(\underline{A}_1)} = \|\underline{A}_1^{-1}x\|_{\underline{X}_1(\underline{A})} = \|\underline{A}\,\underline{A}_1^{-1}x\| = \|x\|,$$

which can be extended by density for all  $x \in \underline{X}_0$ , showing the equality of the spaces  $\underline{X}_{-1}(\underline{A}_1) = \underline{X}_0$  (with the same norm).

(c) By construction the operator  $(\underline{A}_1)_{-1} : \underline{X}_1(A) \to \underline{X}_{-1}(\underline{A}_1)$  is the unique continuous extension of

$$\underline{A}_1: D(\underline{A}_1) = D(\underline{A}^2) \to \underline{X}_1(A),$$

and  $(\underline{A}_1)_{-1}$  is an isometric isomorphism. For  $x \in \underline{X}_1(A)$  we have  $||x||_{\underline{X}_{-1}(A_1)} = ||\underline{A}_1^{-1}x||_{\underline{X}_1(A)} = ||x||$ . But then it follows that  $(\underline{A}_1)_{-1} = \underline{A} : D(\underline{A}) \to \underline{X}_0$ .

(d) The space  $X_{-1}(A_1)$  is defined by

$$X_{-1}(A_1) := (\underline{A_1})_{-2}(X_1(A)) = ((\underline{A_1})_{-1})_{-1}(X_1(A)) = \underline{A}_{-1}(X_1(A)) = AX_1(A) = X_0,$$

by part (c). For the norm equality let  $x \in X_0$ . Then

$$||x|| = ||AA^{-1}x|| = ||A^{-1}x||_{X_1(A)} = ||\underline{A}_{-1}^{-1}x||_{X_1(A)} = ||(\underline{A}_1)_{-2}^{-1}x||_{X_1(A)} = ||x||_{X_{-1}(A_1)}.$$

For the last assertion we note:  $(A_1)_{-1} = (\underline{A}_1)_{-2}|_{X_1(A)} = A$ .

Recall the standing assumption that  $\underline{A} = A|_{\underline{X}_0}$  is densely defined in  $\underline{X}_0 = \overline{D(A)}$ . The following proposition plays the key role for the extension of operators to the extrapolation spaces, particularly for the construction of extrapolated semigroups in Section 2.3.

**Proposition 2.15.** (a) Let  $n \in \mathbb{N}$ . If  $T \in \mathscr{L}(X_0)$  is a linear operator commuting with  $A^{-1}$ , then the operator T has a unique continuous extension to  $\underline{X}_{-n}$  denoted by  $\underline{T}_{-n}$ . The

operator  $\underline{T}_{-n}$  is the restriction of  $\underline{T}_{-n-1}$ . The space  $X_{-n}$  is invariant under  $\underline{T}_{-n-1}$ , whose restriction is denoted by  $T_{-n}$ , for which  $T_{-n} \in \mathscr{L}(X_{-n})$ . For  $k, n \in -\mathbb{N}$  the operators  $\underline{T}_n$ ,  $\underline{T}_k$  are all similar; the same holds for  $T_n$  and  $T_k$ .

(b) Let  $\underline{T} \in \mathscr{L}(\underline{X}_0)$  such that it leaves D(A) invariant and commutes with  $\underline{A}^{-1} = A^{-1}|_{X_0}$ . Then  $\underline{T}_{-1}x = A\underline{T}A^{-1}x$  for each  $x \in X_0$ , and as a consequence,  $\underline{T}_{-1} : \underline{X}_{-1} \to \underline{X}_{-1}$  leaves  $X_0$  invariant (and, of course, extends  $\underline{T}$ ).

*Proof.* (a) For  $x \in X_0$  we have

$$||Tx||_{X_{-1}(A)} = ||A^{-1}Tx|| = ||TA^{-1}x|| \le ||T|| \cdot ||A^{-1}x|| = ||T|| \cdot ||x||_{X_{-1}(A)}.$$

Therefore  $T: (X_0, \|\cdot\|_{X_{-1}(A)}) \to (X_0, \|\cdot\|_{X_{-1}(A)})$  is continuous, and hence has a unique continuous extension  $\underline{T}_{-1}$  to  $\underline{X}_{-1}$ . This extension commutes with  $\underline{A}_{-1}^{-1}$ , because T commutes with  $A^{-1}$  and  $\underline{A}_{-1}^{-1}$  is the unique continuous extension of  $A^{-1}$ . By iteration we obtain the continuous extensions  $\underline{T}_{-n}$  onto  $\underline{X}_{-n}$ , which then all commute with the corresponding  $\underline{A}_{-n}^{-1}$ . By construction  $\underline{T}_{-n}$  is a restriction of  $\underline{T}_{-n-1}$ . We prove that  $X_{-1}$  is invariant under  $\underline{T}_{-2}$ . Let  $x \in X_{-1}$ , hence  $x = \underline{A}_{-2}y$  for some  $y \in X_0$ . Then  $Ty = \underline{T}_{-2}y = \underline{T}_{-2}\underline{A}_{-2}^{-1}x = \underline{A}_{-2}^{-1}\underline{T}_{-2}x$ , hence  $\underline{T}_{-2}x = \underline{A}_{-2}Ty \in X_{-1}$ , i.e., the invariance of  $X_{-1}$  is proved. We have for  $x \in X_{-1}$  that  $\|T_{-1}x\|_{X_{-1}} = \|A_{-2}^{-1}T_{-1}x\| = \|A_{-2}^{-1}\underline{T}_{-2}x\| = \|\underline{T}_{-2}A_{-2}^{-1}x\| \le \|\underline{T}_{2}\| \cdot \|\underline{A}_{-2}^{-1}x\| = \|\underline{T}_{2}\| \cdot \|x\|_{X_{-1}}$ , therefore  $T_{-1} \in \mathscr{L}(X_{-1})$ . The assertion about  $T_{-n}$  follows by recursion.

It is enough to prove the similarity of  $T_0 = T$  and  $T_{-1}$ , and the similarity of  $\underline{T}_0$  and  $\underline{T}_{-1}$ . The latter assertions can be proved as follows: For  $x \in D(A)$  we have

$$\underline{A}_{-1}^{-1}\underline{T}_{-1}\underline{A}_{-1}x = \underline{A}_{-1}^{-1}\underline{T}_{-1}Ax = \underline{A}_{-1}^{-1}TAx = \underline{A}_{-1}^{-1}ATx = \underline{A}_{-1}^{-1}A_{-1}Tx = \underline{T}x,$$

then by continuity and denseness the equality follows even for  $x \in \underline{X}_0$ . For the similarity of T and  $T_{-1}$  take  $x \in X_0$ . Then

$$A_{-1}^{-1}T_{-1}A_{-1}x = \underline{A}_{-2}^{-1}\underline{T}_{-2}\underline{A}_{-2}x = \underline{T}_{-1}x = Tx.$$

(b) Let  $x \in X_0 \subseteq \underline{X}_{-1}$ . Then there is a sequence  $(x_n)$  in  $\underline{X}_0$  with  $x_n \to x$  in  $\underline{X}_{-1}$  (see Proposition 2.8). But then  $A^{-1}x_n \to A^{-1}x$  in  $\underline{X}_0$  and  $\underline{T}x_n \to \underline{T}_{-1}x$  in  $\underline{X}_{-1}$  by part (a). These imply  $\underline{T}A^{-1}x_n = A^{-1}\underline{T}x_n \to \underline{A}_{-1}^{-1}\underline{T}_{-1}x$ . Hence we conclude  $\underline{T}A^{-1}x = \underline{A}_{-1}^{-1}\underline{T}_{-1}x$  and  $A\underline{T}A^{-1}x = \underline{T}_{-1}x$  for  $x \in X_0$ .

M. Haase in [62] and S.-A. Wegner in [123] have constructed the so-called universal extrapolation space  $X_{-\infty}$  as follows: Suppose A is densely defined (this assumption is not made by M. Haase), then  $X_n = \underline{X}_n$  for each  $n \in \mathbb{Z}$  and let  $X_{-\infty}$  to be the inductive limit of the sequence of Banach spaces  $(X_{-n})_{n \in \mathbb{N}}$  (algebraic inductive limit in [62]). In particular, the space  $X_{-\infty}$  is algebraically defined by

$$X_{-\infty} := \varinjlim_{n \in \mathbb{N}} X_{-n} = \bigcup_{n \in \mathbb{N}} X_{-n}$$

A topology on this space is defined by means of nets as follows: A net  $(x_{\iota})_{\iota \in I}$  in  $X_{-\infty}$ converges to  $x \in X_{-\infty}$  if and only if there exists  $n \in \mathbb{N}$  and  $\iota_0 \in I$  such that  $x, x_{\iota} \in X_{-n}$ for each  $\iota \geq \iota_0$  and  $||x - x_{\iota}||_{X_{-n}} \to 0$ . In other words, the topology on  $X_{-\infty}$  is the finest topology which makes all inclusions  $X_{-n} \hookrightarrow X_{-\infty}$  continuous. One can extend the operator A to an operator  $A_{-\infty} : X_{-\infty} \to X_{-\infty}$  such that

$$A_{-\infty}|_{X_n} = A_n, \quad n \in \mathbb{Z}.$$

Observe that this operator is continuous with respect to the notion of convergence defined above. We now look at a converse situation, and our starting point is the following: Let  $\mathscr{E}$  be a locally convex space such that we can embed the Banach space  $X_0$  continuously in  $\mathscr{E}$ , i.e., there is a continuous injective map  $i : X_0 \to \mathscr{E}$ , and so we can identify  $X_0$  with a subspace of  $\mathscr{E}$ . We also assume that we have a continuous operator  $\mathcal{A} : \mathscr{E} \to \mathscr{E}$  such that  $\lambda - \mathcal{A} : i(X_0) \to \mathscr{E}$  is injective and that

$$D(A) = \{ x \in X_0 : \mathcal{A} \circ i(x) \in i(X_0) \},\$$

and

$$i \circ A = \mathcal{A} \circ i|_{D(A)}.$$

In the next theorem we use this setting to describe the extrapolation spaces  $X_{-n}$ ,  $X_{-n}$ . Notice that we do not assume that A is a Hille–Yosida operator or densely defined.

**Theorem 2.16.** Let  $X_0$  be a Banach space with a continuous embedding  $i : X_0 \to \mathscr{E}$  into a locally convex space  $\mathscr{E}$ , let  $A : D(A) \to X_0$  be a linear operator with  $\lambda \in \rho(A)$  such that  $A = \mathcal{A}|_{X_0}$  (after identifying  $X_0$  with a subspace of  $\mathscr{E}$  as described above). We suppose furthermore that  $\lambda - \mathcal{A}$  is injective on  $X_0$ . Then there is a continuous embedding  $i_{-1} : X_{-1} \to \mathscr{E}$  which extends i. After identifying  $X_{-1}$  with a subspace of  $\mathscr{E}$  (under  $i_{-1}$ ) we have

$$X_{-1} = \{ (\lambda - \mathcal{A})x : x \in X_0 \}, \quad \underline{X}_{-1} = \{ (\lambda - \mathcal{A})x : x \in \underline{X}_0 \} \quad and \quad A_{-1} = \mathcal{A}|_{X_{-1}}.$$

*Proof.* Without lost of generality we may assume that  $\lambda = 0$ . Recall that  $A_{-1}|_{X_0} = A$  and  $A_{-1}$  is an isometric isomorphism  $A_{-1}: X_0 \to X_{-1}$ . We now define the embedding  $i_{-1}: X_{-1} \to \mathscr{E}$  by

$$i_{-1} := \mathcal{A} \circ i \circ A_{-1}^{-1},$$

which is indeed injective and continuous by assumption. Of course,  $i_{-1}$  extends i since we have  $i = \mathcal{A} \circ i \circ \mathcal{A}^{-1}$ . We can write

$$i_{-1} \circ A_{-1} = \mathcal{A} \circ i \circ A_{-1}^{-1} \circ A_{-1} = \mathcal{A} \circ i,$$

which yields the following commutative diagram:



Now all assertions follow easily.

The last corollary in this section can be proved by induction based on the previous facts.

**Corollary 2.17.** Let  $\mathcal{A}$ ,  $X_0$ ,  $\mathscr{E}$  and i be as in Theorem 2.16. Then  $X_n \subseteq \mathscr{E}$  and  $A_n = \mathcal{A}|_{X_n}$  for each  $n \in \mathbb{Z}$  (after identifying  $X_n$  with a subspace of  $\mathscr{E}$  under an embedding  $i_n$  compatible with i).

# § 2.2 Intermediate spaces for operators with rays of minimal growth

The following definition of intermediate, and as a matter of fact interpolation spaces, just as well many results in this section are standard, and we refer, e.g., to the book by Lunardi [85, Chapter 3], and to Engel, Nagel [52, Section II.5] for the case of semigroup generators. In this section we suppose the following.

Assumption C. The operator A on the Banach space  $X_0$  has a ray of minimal growth, i.e.,  $(0, \infty) \subseteq \rho(A)$  and for some  $M \ge 0$ 

$$\|\lambda R(\lambda, A)\| \le M \quad \text{for all } \lambda > 0. \tag{2.2.1}$$

**Definition 2.18.** For  $\alpha \in (0, 1]$  and  $x \in X_0$  we define

$$||x||_{F_{\alpha}(A)} := \sup_{\lambda > 0} ||\lambda^{\alpha} AR(\lambda, A)x||,$$

and the *abstract Favard space* of order  $\alpha$  by

$$F_{\alpha}(A) := \{ x \in X_0 : \|x\|_{F_{\alpha}(A)} < \infty \}.$$

In the literature the notation  $D_A(\alpha, \infty)$  is also used, see, e.g., [85]. We further set

$$F_0(A) := F_1(A_{-1})$$

see [52, Section II.5(b)] for the case of semigroup generators.

- **Proposition 2.19.** (a) The Favard space  $F_{\alpha}(A)$  becomes a Banach space if endowed with the norm  $\|\cdot\|_{F_{\alpha}(A)}$ .
- (b) The space  $X_0$  is isomorphic to a closed subspace of  $F_0(A)$ .

The statement that  $X_0$  is a closed subspace of  $F_0(A)$  when A is a Hille–Yosida operator is due to Nagel and Sinestrari [95, Proof of Prop. 2.7].

*Proof.* (a) is trivial.

(b) For  $x \in X_0$  we have

$$\|\lambda A_{-1}R(\lambda, A_{-1})x\|_{X_{-1}(A)} = \|\lambda AR(\lambda, A)x\|_{X_{-1}(A)} = \|\lambda A^{-1}AR(\lambda, A)x\| \le M\|x\|_{X_{-1}(A)}$$

yielding

$$||x||_{F_0(A)} = ||x||_{F_1(A_{-1})} \le M ||x||.$$

On the other hand, since A and  $A_{-1}$  are similar, we have  $\sup_{\lambda>0} \|\lambda R(\lambda, A_{-1})\|_{X_{-1}(A)} \leq M'$ for some  $M' \geq 0$  and for all  $\lambda > 0$ . In particular, by Remark 2.6,  $\lambda R(\lambda, A_{-1})x \to x$  for each  $x \in X_{-1}$ . From this we obtain for  $x \in X_0$  that

$$\begin{aligned} \|x\| &= \|A_{-1}x\|_{X_{-1}(A)} = \left\|\lim_{\lambda \to 0} \lambda R(\lambda, A_{-1})A_{-1}x\right\|_{X_{-1}(A)} \le \sup_{\lambda > 0} \left\|\lambda A_{-1}R(\lambda, A_{-1})x\right\|_{X_{-1}(A)} \\ &= \|x\|_{F_1(A_{-1})} = \|x\|_{F_0(A)}, \end{aligned}$$

showing the equivalence of the norms  $\|\cdot\|$  and  $\|x\|_{F_0(A)}$  on  $X_0$ .

We also need the following well-known result, see, e.g., [85, Chapters 1 and 3], for which we give a short proof.

**Proposition 2.20.** For  $\alpha \in (0,1]$  we have  $F_{\alpha}(A) \subseteq \overline{D(A)} = \underline{X}_0$ .

*Proof.* We have

$$AR(\lambda, A)x = \lambda R(\lambda, A)x - x,$$

so that

$$\|\lambda R(\lambda, A)x - x\| \le \frac{\|x\|_{F_{\alpha}(A)}}{\lambda^{\alpha}} \to 0 \text{ as } \lambda \to \infty.$$

**Definition 2.21.** Let A be a linear operator on the Banach space  $X_0$  satisfying (2.2.1). For  $\alpha \in (0, 1)$  we set

$$\underline{X}_{\alpha}(A) := \Big\{ x \in F_{\alpha}(A) : \lim_{\lambda \to \infty} \lambda^{\alpha} AR(\lambda, A) x = 0 \Big\},$$

and we recall from Section 2.1 that

$$\underline{X}_0(A) := \overline{D(A)}, \quad \underline{X}_1(A) = D(A|_{\underline{X}_0(A)}).$$

The proof of the next proposition is straightforward and well-known.

**Proposition 2.22.** For  $\alpha, \beta \in (0, 1)$  with  $\alpha > \beta$  we have

$$\underline{X}_1(A) \hookrightarrow \underline{X}_{\alpha}(A) \subseteq F_{\alpha}(A) \hookrightarrow \underline{X}_{\beta}(A) \subseteq F_{\beta}(A) \hookrightarrow \underline{X}_0(A) \subseteq X_0(A)$$

with  $\hookrightarrow$  denoting continuous and dense embeddings of Banach spaces, and  $\subseteq$  denoting inclusion of closed subspaces.

*Proof.* For  $x \in F_{\alpha}(A)$  we have

$$\|\lambda^{\beta} A R(\lambda, A) x\| = \lambda^{\beta - \alpha} \|\lambda^{\alpha} A R(\lambda, A) x\| \le \lambda^{\beta - \alpha} \|x\|_{\alpha} \to 0 \quad \text{as } \lambda \to \infty,$$

which also proves the continuity of  $F_{\alpha}(A) \hookrightarrow \underline{X}_{\beta}(A)$ . The other statements can be proved by similar reasonings.

- **Proposition 2.23.** (a) The spaces  $F_{\alpha}(A)$  and  $\underline{X}_{\alpha}(A)$  are invariant under each  $T \in \mathscr{L}(X_0)$  which commutes with  $A^{-1}$ .
- (b) If  $T \in \mathscr{L}(X_0)$  commutes with  $A^{-1}$ , then the space  $F_0(A)$  is invariant under  $T_{-1}$ .

*Proof.* (a) Suppose that  $T \in \mathscr{L}(X_0)$  commutes with  $R(\cdot, A)$  and let  $x \in \underline{X}_{\alpha}(A)$ . We have to show that  $Tx \in \underline{X}_{\alpha}(A)$ . Since T is assumed to be bounded, we obtain:

$$\|\lambda^{\alpha} AR(\lambda, A)Tx\| = \|\lambda^{\alpha} ATR(\lambda, A)x\| \le \|T\| \cdot \|\lambda^{\alpha} AR(\lambda, A)x\|.$$

This implies both assertions.

(b) Follows from part (a) applied to  $T_{-1}$  on the space  $X_{-1}$ .

**Definition 2.24.** For  $\alpha \in \mathbb{R}$  we write  $\alpha = m + \beta$  with  $m \in \mathbb{Z}$  and  $\beta \in (0, 1]$ , and define

$$F_{\alpha}(A) := F_{\beta}(A_m),$$

with the corresponding norms. For  $\alpha \notin \mathbb{Z}$  we define

$$\underline{X}_{\alpha}(A) := \underline{X}_{\beta}(A_m),$$

also with the corresponding norms.

In particular we have for  $\alpha \in (0, 1)$  that

$$\underline{X}_{-\alpha}(A) = \underline{X}_{1-\alpha}(A_{-1}) \quad \text{and} \quad F_{-\alpha}(A) = F_{1-\alpha}(A_{-1}).$$

This definition is consistent with Definitions 2.18 and 2.21. The following property of these spaces can be directly deduced from the definitions and the previous assertions (by induction):

**Proposition 2.25.** For any  $\alpha, \beta \in \mathbb{R}$  with  $\alpha > \beta$  we have

$$\underline{X}_{\alpha}(A) \subseteq F_{\alpha}(A) \hookrightarrow \underline{X}_{\beta}(A) \subseteq F_{\beta}(A)$$

with  $\hookrightarrow$  denoting continuous and dense embeddings of Banach spaces, and  $\subseteq$  denoting inclusion of closed subspaces.

Now we put these spaces in the context presented at the end of Section 2.1.

**Proposition 2.26.** (a) For  $\alpha \in (0, 1]$  we have  $A_{-1}F_{\alpha} = F_{\alpha-1}$  and  $A_{-1}\underline{X}_{\alpha} = \underline{X}_{\alpha-1}$ .

(b) For  $\alpha \in (0,1]$  and  $\mathcal{A}$ ,  $\lambda$  and  $\mathcal{E}$  as in Theorem 2.16 we have

$$F_{-\alpha} = \Big\{ (\lambda - \mathcal{A}) y \in X_{-1} : y \in F_{1-\alpha} \Big\}.$$

If  $\alpha \in (0,1)$ , then

$$\underline{X}_{-\alpha} = \Big\{ (\lambda - \mathcal{A}) y \in X_{-1} : \ y \in \underline{X}_{1-\alpha} \Big\}.$$

# § 2.3 Intermediate and extrapolation spaces for semigroup generators

In this section we consider intermediate and extrapolation spaces when the linear operator  $A: D(A) \to X_0$  is the generator of a semigroup  $(T(t))_{t\geq 0}$  (meaning that  $T: [0, \infty) \to \mathscr{L}(X_0)$  is a monoid homomorphism) in the sense described in the following.

**Assumption 2.27.** 1. Let  $X_0$  be a Banach space, and let  $Y \subseteq X'_0$  be a norming subspace, i.e.,

$$||x|| = \sup_{y \in Y, ||y|| \le 1} |\langle x, y \rangle| \quad \text{for each } x \in X_0.$$

2. Let  $T: [0, \infty) \to \mathscr{L}(X_0)$  be a semigroup of contractions for which a generator  $A: D(A) \to X_0$  exists in the sense that

$$R(\lambda, A)x = \int_0^\infty e^{-\lambda s} T(s)x \, \mathrm{d}s \tag{2.3.1}$$

exists for each  $\lambda \ge 0$  as a weak integral with respect to the dual pair  $(X_0, Y)$ , i.e., for each  $y \in Y$  and  $x \in X_0$ 

$$\langle R(\lambda, A)x, y \rangle = \int_0^\infty e^{-\lambda s} \langle T(s)x, y \rangle \, \mathrm{d}s,$$

and  $R(\lambda, A) \in \mathscr{L}(X_0)$  is the resolvent of a linear operator A (see [80] by Kunze).

3. We also suppose that T(t) commutes with  $A^{-1}$  for each  $t \ge 0$ .

If the semigroup  $(T(t))_{t>0}$  is only exponentially bounded of type  $(M, \omega)$ , that is

$$||T(t)|| \le M e^{\omega t}$$
 for all  $t \ge 0$ .

then one can rescale the semigroup  $(\operatorname{consider}(e^{-(\omega+1)t}T(t))_{t\geq 0})$ , and renorm the Banach space such that the rescaled semigroup becomes a contraction semigroup. Moreover, the new semigroup has negative growth bound, meaning that  $T(t) \to 0$  in norm exponentially fast as  $t \to \infty$ . Then it also has an invertible generator.

- **Remark 2.28.** (i) There are several important classes of semigroups, satisfying Assumption 2.27, hence can be treated in a unified manner:  $\pi$ -semigroups of E. Priola [100], weakly continuous semigroups of S. Cerrai [32], bi-continuous semigroups of F. Kühnemund. We will concentrate on this latter class of semigroups in Section 2.4.
  - (ii) In this framework M. Kunze [80] introduced the notion of integrable semigroups, which we briefly describe next. Since we have

$$\|y\| = \sup_{x \in X_0, \|x\| \le 1} |\langle x, y \rangle|$$

and, by the norming assumption,

$$\|x\| = \sup_{y \in Y, \|y\| \le 1} |\langle x, y \rangle|,$$

the pair  $(X_0, Y)$  is called a norming dual pair. M. Kunze has worked out the theory of semigroups on such norming dual pairs in [80]. We recall at least the basic definitions

here: assume without loss of generality that Y is a Banach space and consider the weak topology  $\sigma = \sigma(X_0, Y)$  on  $X_0$ . An *integrable semigroup* of type  $(M, \omega)$  on the pair  $(X_0, Y)$  is a semigroup  $(T(t))_{t\geq 0}$  of  $\sigma$ -continuous linear operators satisfying:

- 1.  $(T(t))_{t\geq 0}$  is a semigroup, i.e. T(t+s) = T(t)T(s) and T(0) = I for all  $t, s \geq 0$ .
- 2. For all  $\lambda$  with  $\operatorname{Re}(\lambda) > \omega$ , there exists an  $\sigma$ -continuous linear operator  $R(\lambda)$  such that for all  $x \in X_0$  and all  $y \in Y$

$$\langle R(\lambda)x,y\rangle = \int_0^\infty e^{-\lambda t} \langle T(t)x,y\rangle \, dt$$

Kunze defines the generator A of the semigroup as the (unique) operator  $A: D(A) \to X_0$ (if it exists at all) with  $R(\lambda) = (\lambda - A)^{-1}$ , precisely as in Assumption 2.27. Note that  $\sigma$ -continuity of T(t) can be used to assure that Y is invariant under T'(t), cf. the next remark.

**Remark 2.29.** The semigroup  $(T(t))_{t\geq 0}$  commutes with the inverse of the generator if Y can be chosen such that it is invariant under T'(t) for each  $t \geq 0$ :

$$\begin{split} \langle A^{-1}T(t)x,y\rangle &= \int_0^\infty \langle T(s)T(t)x,y\rangle \,\,\mathrm{d}s = \int_0^\infty \langle T(s+t)x,y\rangle \,\,\mathrm{d}s \\ &= \Big\langle \int_0^\infty T(s)x \,\,\mathrm{d}s, T'(t)y \Big\rangle = \langle T(t)A^{-1}x,y\rangle, \end{split}$$

for each  $x \in X_0$  and  $y \in Y$ .

**Remark 2.30.** (i) From (2.3.1) it follows that for each  $x \in X_0$ 

$$T(t)x - x = A \int_0^t T(s)x \, \mathrm{d}s.$$
 (2.3.2)

Indeed, we have by (2.3.1) that

$$x = A \int_0^\infty T(s)x \, \mathrm{d}s$$
$$T(t)x = A \int_0^\infty T(s)T(t)x \, \mathrm{d}s = \int_t^\infty T(s)x \, \mathrm{d}s$$

Subtracting the first of these equation from the second one we obtain the statement.

(ii) If moreover A commutes with T(t) for each  $t \ge 0$ , then for each  $x \in D(A)$  we have

$$T(t)x - x = \int_0^t T(s)Ax \, \mathrm{d}s.$$
 (2.3.3)

Indeed, as in the above, we have by (2.3.1)

$$-x = -A^{-1}Ax = \int_0^\infty T(s)Ax \, \mathrm{d}s$$
$$-T(t)x = -A^{-1}T(t)Ax = \int_0^\infty T(s)T(t)Ax \, \mathrm{d}s = \int_t^\infty T(s)Ax \, \mathrm{d}s.$$

By a simple subtraction we obtain the statement.

The next lemma and its proof are standard for various classes of semigroups.

**Lemma 2.31.** If  $(T(t))_{t\geq 0}$  is (locally) norm bounded, then

$$X_{\text{cont}} := \{ x \in X_0 : t \mapsto T(t)x \text{ is } \| \cdot \| \text{-continuous} \}$$

is a closed a subspace of  $X_0$  invariant under the semigroup. Under Assumption 2.27 we have

$$\underline{X}_0 = \overline{D(A)} = X_{\text{cont}}.$$

*Proof.* The closedness and invariance of  $X_{\text{cont}}$  are evident. We first show  $D(A) \subseteq X_{\text{cont}}$ , which implies  $\overline{D(A)} \subseteq X_{\text{cont}}$  by closedness of  $X_{\text{cont}}$ . By (2.3.3) we conclude for  $x \in D(A)$  that  $T(t)x - x = \int_0^t T(s)Ax \, \mathrm{d}s$ . Since

$$||T(t)x - x|| = \sup_{||y|| \le 1} |\langle T(t)x - x, y\rangle| \le \sup_{||y|| \le 1} \int_0^t |\langle T(s)Ax, y\rangle| \, \mathrm{d}s \le t ||Ax|| \to 0,$$

as  $t \to 0$ , we obtain  $D(A) \subseteq X_{\text{cont}}$  and  $\overline{D(A)} \subseteq X_{\text{cont}}$ . For the converse inclusion suppose that  $x \in X_{\text{cont}}$ . Again by (2.3.2) we obtain that the sequence of vectors  $x_n := n \int_0^{\frac{1}{n}} T(s) x \, \mathrm{d}s \in D(A)$   $(n \in \mathbb{N})$  converges to x. Indeed:

$$||x_n - x|| = \sup_{||y|| \le 1} |\langle x_n - x, y \rangle| \le \sup_{||y|| \le 1} n \int_0^{\frac{1}{n}} |\langle T(s)x - x, y \rangle| \, \mathrm{d}s \le n \int_0^{\frac{1}{n}} ||T(s)x - x|| \, \mathrm{d}s.$$

By the continuity of  $s \mapsto T(s)x$  we obtain the inclusion  $X_{\text{cont}} \subseteq \overline{D(A)}$ .

Based on this lemma one can prove the following characterization of the Favard and Hölder spaces:

**Proposition 2.32.** Let  $(T(t))_{t\geq 0}$  be a semigroup satisfying Assumption 2.27 with negative growth bound and generator A. For  $\alpha \in (0,1]$  define

$$F_{\alpha}(T) := \left\{ x \in X_0 : \sup_{s>0} \frac{\|T(s)x - x\|}{s^{\alpha}} < \infty \right\} = \left\{ x \in X_0 : \sup_{s \in (0,1)} \frac{\|T(s)x - x\|}{s^{\alpha}} < \infty \right\}, \quad (2.3.4)$$

and for  $\alpha \in (0,1)$  define

$$\underline{X}_{\alpha}(T) := \left\{ x \in X_0 : \sup_{s > 0} \frac{\|T(s)x - x\|}{s^{\alpha}} < \infty \text{ and } \lim_{s \downarrow 0} \frac{\|T(s)x - x\|}{s^{\alpha}} = 0 \right\}$$

$$= \left\{ x \in X_0 : \lim_{s \downarrow 0} \frac{\|T(s)x - x\|}{s^{\alpha}} = 0 \right\},$$
(2.3.5)

which become Banach spaces if endowed with the norm

$$||x||_{F_{\alpha}(T)} := \sup_{s>0} \frac{||T(s)x - x||}{s^{\alpha}}$$

The space  $\underline{X}_{\alpha}(T)$  is a closed subspace of  $F_{\alpha}(T)$ . These spaces are invariant under the semigroup  $(T(t))_{t\geq 0}$ , and  $\underline{X}_{\alpha}(T)$  is the space of  $\|\cdot\|_{F_{\alpha}(T)}$ -strong continuity in  $F_{\alpha}(T)$ . For  $\alpha \in (0,1]$  we have  $F_{\alpha}(A) = F_{\alpha}(T)$  and for  $\alpha \in (0, 1)$  we have  $\underline{X}_{\alpha}(A) = \underline{X}_{\alpha}(T)$  with equivalent norms. Proof. For  $x \in F_{\alpha}(T)$  we have

$$\|T(t)x\|_{F_{\alpha}(T)} = \sup_{s>0} \frac{\|T(s)T(t)x - T(t)x\|}{s^{\alpha}} \le \|T(t)\| \cdot \sup_{s>0} \frac{\|T(s)x - x\|}{s^{\alpha}} \le M\|x\|_{F_{\alpha}(T)},$$

proving the invariance of  $F_{\alpha}(T)$ . Similar reasoning proves the invariance of  $\underline{X}_{\alpha}$ . Since  $F_{\alpha}(T) \subseteq X_{\text{cont}} = \underline{X}_0 = \overline{D(A)}$  and  $F_{\alpha}(A) \subseteq \underline{X}_0 = \overline{D(A)}$ , the rest of the assertions follow from the corresponding results concerning  $C_0$ -semigroups, see, e.g., [52, Sec. II.5].

We conclude with the construction of the extrapolated semigroup as a direct consequence of Proposition 2.15.

**Proposition 2.33.** Let A generate the semigroup  $(T(t))_{t\geq 0}$  of negative growth bound in the sense of Assumption 2.27. Then there is an extension  $(T_{-1}(t))_{t\geq 0}$  of the semigroup  $(T(t))_{t\geq 0}$  on the extrapolated space  $X_{-1}$ , whose generator is  $A_{-1}$ .

#### 2.3.1 Intermezzo on real interpolation spaces

To complete the picture we recall the next result [85, Prop. 5.7], which is formulated there only for  $C_0$ -semigroups as a theorem, but A. Lunardi also remarks, without stating the precise assumptions, that this result still holds if one omits the strong continuity assumption. We require here the conditions from Assumption 2.27, under which the proof is verbatim the same as for the  $C_0$ -case, and is based on formulas (2.3.2) and (2.3.3). Before we state the result, recall the following definitions from [85].

**Definition 2.34.** Let X and Y be two real or complex Banach spaces. The couple (X, Y) is an *interpolation couple* if both X and Y are continuously embedded in a Hausdorff topological vector space  $\mathcal{V}$ .

If (X, Y) is such an interpolation couple, also  $X + Y := \{x + y : x \in X, y \in Y\}$  is a linear subspace of  $\mathcal{V}$  and becomes a Banach space equipped with the norm  $\|\cdot\|_{X+Y}$  given by

$$\|z\|_{X+Y} := \inf_{\substack{x \in X, \ y \in Y \\ z = x+y}} \|x\|_X + \|y\|_Y$$

**Definition 2.35.** For each  $x \in X + Y$  and t > 0 we set

$$K(t,x) := \inf_{\substack{a \in X, \ b \in Y \\ x=a+b}} \|a\|_X + t \, \|b\|_Y.$$

**Definition 2.36.** Let  $0 < \theta < 1$ ,  $1 \le p \le \infty$ . The real interpolation spaces  $(X, Y)_{\theta,p}$  are defined by

$$(X,Y)_{\theta,p} := \left\{ x \in X + Y : t \mapsto t^{-\theta} K(t,x) \in \mathcal{L}^p_*(0,\infty) \right\}$$

and are normed by

$$\|x\|_{\theta,p}:=\left\|t^{-\theta}K(t,x)\right\|_{\mathrm{L}^p_*(0,\infty)}$$

where  $L^p_*(0,\infty)$  denotes the L<sup>*p*</sup>-space with respect to the Haar measure  $\frac{dt}{t}$  on the multiplicative group  $(0,\infty)$ .

Now we are able to formulate and to prove the promised result.

**Proposition 2.37.** Let A generate the semigroup  $(T(t))_{t>0}$  of negative growth bound in the sense describe in Assumption 2.27. Then for  $p \in [1, \infty]$  and  $\alpha \in (0, 1)$  we have for the interpolation space:

$$(X, D(A))_{\alpha, p} = \{ x \in X : t \mapsto \psi(t) := t^{-\alpha} \| T(t)x - x \| \in L^p_*(0, \infty) \},\$$

where  $L^p_*(0,\infty)$  denotes the  $L^p$ -space with respect to the Haar measure  $\frac{dt}{t}$  on the multiplicative group  $(0,\infty)$ . Moreover, the norms  $||x||_{\alpha,p}$  and

$$||x||_{\alpha,p}^{**} = ||x|| + ||\psi||_{\mathcal{L}^p_*(0,\infty)}$$

are equivalent.

*Proof.* For each  $b \in D(A)$  we have

$$T(t)b - b = \int_0^t AT(s)b \, ds = \int_0^t T(s)Ab \, ds, \quad t > 0.$$

Let  $x \in (X, D(A))_{\theta, p}$ . If x = a + b with  $a \in X$  and  $b \in D(A)$ , then

$$t^{-\theta} \|T(t)x - x\| \le t^{-\theta} \left( \|T(t)a - a\| + \|T(t)b - b\| \right) \le (M+1)t^{-\theta}K(t,x)$$

for each t > 0 which means in particular that  $\psi(t) = t^{-\theta} ||T(t)x - x|| \in L^p_*(0,\infty)$  and

$$||x||_{\theta,p}^{**} \le (M+1) ||x||_{\theta,p}$$

For the converse assume that  $\psi \in L^p_*(0,\infty)$ . We obtain

$$\lambda^{\theta} \left\| AR(\lambda, A) x \right\| \leq \int_{0}^{\infty} \lambda^{\theta+1} t^{\theta+1} \mathrm{e}^{-\lambda t} \frac{\|T(t)x - x\|}{t^{\theta}} \frac{\mathrm{d}t}{t},$$

which is in fact the convolution  $f * \psi$  for  $f(t) = t^{\theta+1} e^{-t}$  and  $\psi(t) = t^{-\theta} ||T(t)x - x||$ . Since  $f \in L^1_*(0,\infty)$  and  $\psi \in L^p_*(0,\infty)$  one gets  $f * \psi \in L^p_*(0,\infty)$  and

$$\|f * \psi\|_{\mathcal{L}^{p}_{*}(0,\infty)} \leq \|f\|_{\mathcal{L}^{1}_{*}(0,\infty)} \|\psi\|_{\mathcal{L}^{p}_{*}(0,\infty)}.$$

From that we obtain

$$||x||_{\theta,p}^* \le \Gamma(\theta+1) ||x||_{\theta,p}^{**}.$$

# § 2.4 Intermediate and extrapolation spaces for bi-continuous semigroups

#### 2.4.1 The Hille–Yosida Theorem

Recall the following result of F. Kühnemund from [79, Thm. 16], whose proof is originally based on integrated semigroups. We present here a different proof based on extrapolation spaces.

**Theorem 2.38.** Let  $(X_0, \|\cdot\|, \tau)$  be a triple satisfying Assumption 1.1, and let A be a linear operator on the Banach space  $X_0$ . The following are equivalent:

- (a) The operator A is the generator of a bi-continuous semigroup  $(T(t))_{t\geq 0}$  of type  $(M,\omega)$ .
- (b) The operator A is a Hille-Yosida operator of type  $(M, \omega)$ , i.e.,

$$||R(s,A)^k|| \le \frac{M}{(s-\omega)^k}$$

for all  $k \in \mathbb{N}$  and for all  $s > \omega$ . Moreover, A is bi-densely defined and the family

$$\{(s-\alpha)^k R(s,A)^k : k \in \mathbb{N}, s \ge \alpha\}$$
(2.4.1)

is bi-equicontinuous for each  $\alpha > \omega$ , meaning that for each norm bounded  $\tau$ -null sequence  $(x_n)$  one has  $(s-\alpha)^k R(s,A)^k x_n \to 0$  in  $\tau$  uniformly for  $k \in \mathbb{N}$  and  $s \ge \alpha$  as  $n \to \infty$ .

In this case, we have the Euler formula:

$$T(t)x := \tau \lim_{m \to \infty} \left( \frac{m}{t} R\left(\frac{m}{t}, A\right) \right)^m x \quad \text{for each } x \in X_0.$$

Moreover, the subspace  $\underline{X}_0 := D(A) \subseteq X_0$  is the space of norm strong continuity for  $(T(t))_{t\geq 0}$ , it is invariant under the semigroup, and  $(\underline{T}(t))_{t\geq 0} := (T(t)|_{\underline{X}_0})_{t\geq 0}$  is the strongly continuous semigroup on  $\underline{X}_0$  generated by the part  $\underline{A}$  of A in  $\underline{X}_0$ .

*Proof.* It follows from Lemma 2.31 that  $\underline{X}_0$  is the space of norm strong continuity for a bi-continuous semigroup  $(T(t))_{t\geq 0}$ .

We only prove the implication (b)  $\Rightarrow$  (a) and the Euler formula; the other implication is standard and the easier one and can be found here [78, Thm. 1.28]. We may suppose that  $\omega < 0$ . Since A is a Hille–Yosida operator, the part <u>A</u> of A in <u>X</u><sub>0</sub> generates a C<sub>0</sub>-semigroup  $(\underline{T}(t))_{t\geq 0}$  of type  $(M, \omega)$  on the space  $\underline{X}_0 := \overline{D(A)}$ . Define the function

$$F(s) := \begin{cases} \frac{1}{s}R(\frac{1}{s}, A) & \text{for } s > 0, \\ I & \text{for } s = 0, \end{cases}$$

which is strongly continuous on  $\underline{X}_0$  by Remark 2.6. Moreover, we have the Euler formula

$$\underline{T}_0(t)x = \lim_{m \to \infty} F(\frac{t}{m})^m x$$

for  $x \in \underline{X}_0$  with convergence being uniform for t in compact intervals  $[0, t_0]$ , see, e.g., [52, Section III.5(a)]. Since  $R(\lambda, A)|_{\underline{X}_0} = R(\lambda, \underline{A})$  and since D(A) is bi-dense in  $X_0$ , by the local bi-equicontinuity assumption in (2.4.1) we conclude that for  $x \in X_0$  and t > 0 the limit

$$S(t)x := \tau \lim_{m \to \infty} F(\frac{t}{m})^m x \tag{2.4.2}$$

exists, and the convergence is uniform for t in compact intervals  $[0, t_0]$ . It follows that  $t \mapsto S(t)x$  is  $\tau$ -strongly continuous for each  $x \in X_0$ . The operator family  $(S(t))_{t\geq 0}$  is locally bi-equicontinuous because of the bi-equicontinuity assumption in (2.4.1).

Next, we prove that  $\underline{T}(t)$  leaves D(A) invariant. Let  $x \in D(A)$ , so that  $x = A^{-1}y$  for some  $y \in X_0$ , and insert x in the formula (2.4.2) to obtain

$$\underline{T}(t)x = S(t)A^{-1}y = \tau \lim_{m \to \infty} F(\frac{t}{m})^m A^{-1}y = A^{-1} \tau \lim_{m \to \infty} F(\frac{t}{m})^m y = A^{-1}S(t)y \in D(A), \quad (2.4.3)$$

where we have used the bi-continuity of  $A^{-1}$  and the boundedness of  $\left(\left[\frac{m}{t}R\left(\frac{m}{t},A\right)\right]^{m}y\right)_{m\in\mathbb{N}}$ . By Proposition 2.15 (b) we can extend  $\underline{T}(t)$  to  $X_{0}$  by setting  $T(t) := A\underline{T}(t)A^{-1} \in \mathscr{L}(X_{0})$ . It follows that  $(T(t))_{t\geq 0}$  is a semigroup. By formula (2.4.3), we have  $T(t)y = A\underline{T}(t)A^{-1}y = AA^{-1}S(t)y = S(t)y$  for each  $y \in X_{0}$ . So that  $(T(t))_{t\geq 0}$ , coinciding with  $(S(t))_{t\geq 0}$ , is locally bi-equicontinuous, and hence a bi-continuous semigroup.

It remains to show that the generator of  $(T(t))_{t\geq 0}$  is A. Let B denote the generator of  $(T(t))_{t\geq 0}$ . Then, for large  $\lambda > 0$  and  $x \in \underline{X}_0$ , we have

$$R(\lambda, B)x = \int_0^\infty e^{-\lambda s} T(s)x \, \mathrm{d}s = \int_0^\infty e^{-\lambda s} \underline{T}(s)x \, \mathrm{d}s = R(\lambda, \underline{A}_0)x = R(\lambda, A)x.$$

Since  $R(\lambda, B)$  and  $R(\lambda, A)$  are sequentially  $\tau$ -continuous on norm bounded sets and since D(A) is bi-dense in  $X_0$ , we obtain  $R(\lambda, B) = R(\lambda, A)$ . This finishes the proof.

The first statement in the next proposition is proved by R. Nagel and E. Sinestrari, see [92] and [94], while the second one follows directly from the results in Section 2.1.

**Proposition 2.39.** Let A be a Hille–Yosida operator on the Banach space  $X_0$  with domain D(A). Denote by  $(\underline{T}(t))_{t\geq 0}$  the  $C_0$ -semigroup on  $\underline{X}_0 = \overline{D(A)}$  generated by the part  $\underline{A}$  of A.

- (a) There is a one-parameter semigroup  $(\overline{T}(t))_{t\geq 0}$  on  $F_0(A)$  which extends  $(\underline{T}(t))_{t\geq 0}$ . This semigroup is strongly continuous for the  $\|\cdot\|_{X_{-1}(A)}$  norm.
- (b) Suppose that for each  $t \ge 0$  the operator  $\underline{T}(t)$  leaves D(A) invariant. Then the space  $X_0$  is invariant under the semigroup operators  $\overline{T}(t)$  for every  $t \ge 0$ , i.e., for  $T(t) := \overline{T}(t)|_{X_0}$  we have  $T(t) \in \mathscr{L}(X_0)$ .

#### 2.4.2 Extrapolated semigroups

In this subsection we extend a bi-continuous semigroup on  $X_0$  to the extrapolation space  $X_{-1}$  as a bi-continuous semigroup. We have to handle two topologies, and the next proposition leads to an additional locally convex topology on  $X_{-1}$  still satisfying Assumption 1.1.

**Proposition 2.40.** Let the triple  $(X_0, \|\cdot\|, \tau)$  satisfy Assumption 1.1, let  $\mathcal{P}$  be as in Remark 1.2(iv), let E be a vector space over  $\mathbb{C}$ , and let  $B : X_0 \to E$  be a bijective linear mapping. We define for  $e \in E$  and  $p \in \mathcal{P}$ 

$$||e||_E := ||B^{-1}e||$$
 and  $p_E(e) := p(B^{-1}e).$ 

Then the following assertions hold:

- (a)  $\|\cdot\|_E$  is a norm,  $p_E$  is a seminorm for each  $p \in \mathcal{P}$ .
- (b) For the topology  $\tau_E$  generated by  $\mathcal{P}_E := \{p_E : p \in \mathcal{P}\}$  the triple  $(E, \|\cdot\|_E, \tau_E)$  satisfies the conditions in Assumption 1.1.

(c) If  $(T(t))_{t\geq 0}$  is a bi-continuous semigroup on  $X_0$  with respect to the topology  $\tau$ , then  $T_E(t) := BT(t)B^{-1}$  defines a bi-continuous semigroup on E. If A is the generator of  $(T(t))_{t\geq 0}$ , then  $BAB^{-1}$  is the generator of  $(T_E(t))_{t\geq 0}$ .

*Proof.* Assertion (a) is evident. The conditions (1) and (2) from Assumption 1.1 are satisfied by the definition of  $\|\cdot\|_E$  and  $p_E$ . Since

$$||e||_E = ||B^{-1}e|| = \sup_{p \in \mathcal{P}} p(B^{-1}e) = \sup_{p_E \in \mathcal{P}_E} p_E(e),$$

and by Remark 1.2(iii) in Assumption 1.1 is fulfilled. The proof of (b) is complete.

(c) For  $e \in E$  we have  $||T_E(t)||_E = ||B^{-1}BT(t)B^{-1}e|| = ||T(t)B^{-1}e|| \le ||T(t)|| \cdot ||e||_E$ , which shows that  $T_E(t) \in \mathscr{L}(E)$ . Clearly,  $(T_E(t))_{t\geq 0}$  satisfies the semigroup property. For  $e \in E$ and  $p_E \in \mathcal{P}_E$  we have

$$p_E(T_E(t)e - e) = p(B^{-1}BT(t)B^{-1}e - B^{-1}e) = p(T(t)B^{-1}e - B^{-1}e) \to 0 \text{ for } t \to 0,$$

showing the  $\tau_E$ -strong continuity of  $(T_E(t))_{t\geq 0}$ . If  $(e_n)$  is a  $\|\cdot\|_E$ -bounded,  $\tau_E$ -null sequence, then  $(B^{-1}e_n)$  is a  $\|\cdot\|$ -bounded  $\tau$ -null sequence, so that by assumption  $T_E(t)e_n = T(t)B^{-1}e_n \to 0$  uniformly for t in compact intervals. If A is the generator of  $(T(t))_{t\geq 0}$ , then by means of (1.3.2) we can conclude that  $B^{-1}AB$  is the generator of  $(T_E(t))_{t\geq 0}$ .

**Definition 2.41.** Let  $(T(t))_{t>0}$  be a bi-continuous semigroup in  $X_0$  with generator A.

- (a) For  $B = A^{-1} : X_0 \to X_1$  and  $E = X_1$  in Proposition 2.40 define  $\mathcal{P}_1 := \mathcal{P}_E, \tau_1 := \tau_E, (T_1(t))_{t \ge 0} := (T_E(t))_{t \ge 0}.$
- (b) For  $B = A_{-1} : X_0 \to X_{-1}$  and  $E = X_{-1}$  in Proposition 2.40 define  $\mathcal{P}_{-1} := \mathcal{P}_E, \tau_{-1} := \tau_E, (T_{-1}(t))_{t \ge 0} := (T_E(t))_{t \ge 0}.$

We obtain immediately the next result.

**Proposition 2.42.** The semigroups  $(T_1(t))_{t\geq 0}$  and  $(T_{-1}(t))_{t\geq 0}$  are bi-continuous with generators  $A_1 = A|_{D(A)}$  and  $A_{-1}$ , respectively.

Iterating the procedure in Definition 2.41 we obtain the full scale of (extrapolated) semigroups  $(T_n(t))_{t\geq 0}$  for  $n \in \mathbb{Z}$ .

**Definition 2.43.** Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup on  $X_0$  with generator A and suppose that  $(T_{\pm n}(t))_{t\geq 0}$  and  $\mathcal{P}_{\pm n}$  have been defined for some  $n \in \mathbb{N}$  already.

- (a) For  $B = A_n^{-1} : X_n \to X_{n+1}, E = X_{n+1}$  and the semigroup  $(T_n(t))_{t \ge 0}$  in Proposition 2.40 define  $\mathcal{P}_{n+1} := \mathcal{P}_E, \tau_{n+1} := \tau_E, (T_{n+1}(t))_{t \ge 0} := (T_E(t))_{t \ge 0}$ .
- (b) For  $B = A_{-n-1} : X_{-n} \to X_{-n-1}$ ,  $E = X_{-n-1}$  and the semigroup  $(T_{-n}(t))_{t\geq 0}$  in Proposition 2.40 define  $\mathcal{P}_{-n-1} := \mathcal{P}_E$ ,  $\tau_{-n-1} := \tau_E$ ,  $(T_{-n-1}(t))_{t\geq 0} := (T_E(t))_{t\geq 0}$ .

**Proposition 2.44.** For each  $n \in \mathbb{Z}$  the semigroup  $(T_n(t))_{t\geq 0}$  is bi-continuous on  $(X_n, \|\cdot\|_n, \tau_n)$  with generator  $A_n : X_{n+1} \to X_n$ . Its space of norm strong continuity is  $\underline{X}_n$ .

*Proof.* The first statement follows directly from Proposition 2.42 by induction. For n = 0 the second assertion is the content of Lemma 2.31, for general  $n \in \mathbb{Z}$  one can argue inductively.

The following diagram summarizes the situation:



The spaces  $\underline{X}_{n+1}$  are bi-dense in  $X_n$  for the topology  $\tau_n$  and dense in  $\underline{X}_n$  for the norm  $\|\cdot\|_{X_n}$ . The semigroups  $(T_n(t))_{t\geq 0}$  are bi-continuous on  $X_n$ , while  $(\underline{T}_n(t))_{t\geq 0}$  are  $C_0$ -semigroups (strongly continuous for the norm) on  $\underline{X}_n$ .

#### 2.4.3 Hölder spaces of bi-continuous semigroups

Suppose A generates the bi-continuous semigroup  $(T(t))_{t\geq 0}$  of negative growth bound on  $X_0$ . Recall from Theorem 2.38 that the restricted operators  $\underline{T}(t) := T(t)|_{\underline{X}_0}$  form a  $C_0$ -semigroup  $(\underline{T}(t))_{t\geq 0}$  on  $\underline{X}_0$ . Also recall from Proposition 2.32 that for  $\alpha \in (0, 1]$ 

$$F_{\alpha}(A) = F_{\alpha}(T) = \left\{ x \in \underline{X}_{0} : \sup_{t > 0} \frac{\|\underline{T}(t)x - x\|}{t^{\alpha}} < \infty \right\} = \left\{ x \in X_{0} : \sup_{t > 0} \frac{\|T(t)x - x\|}{t^{\alpha}} < \infty \right\}$$

with the norm

$$\|x\|_{F_{\alpha}} = \sup_{t>0} \frac{\|\underline{T}(t)x - x\|}{t^{\alpha}},$$

and for  $\alpha \in (0, 1)$ :

$$\underline{X}_{\alpha}(A) := \left\{ x \in \underline{X}_0 : \lim_{t \to 0} \frac{\|\underline{T}(t)x - x\|}{t^{\alpha}} = 0 \right\} = \left\{ x \in X_0 : \lim_{t \to 0} \frac{\|T(t)x - x\|}{t^{\alpha}} = 0 \right\}.$$

We have the (continuous) inclusions

$$\underline{X}_1 \hookrightarrow X_1 \to \underline{X}_\alpha(A) \hookrightarrow F_\alpha(A) \to \underline{X}_0 \hookrightarrow X_0;$$

all these spaces are invariant under  $(T(t))_{t\geq 0}$ . We now extend this diagram by a space which lies between  $\underline{X}_{\alpha}$  and  $F_{\alpha}$ .

**Definition 2.45.** Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup of negative growth bound on a Banach space  $X_0$  with respect to a locally convex topology  $\tau$  that is generated by a family  $\mathcal{P}$  of seminorms satisfying (1.1.2). For  $\alpha \in (0, 1)$  we define the space

$$X_{\alpha} := X_{\alpha}(T) := \Big\{ x \in X_0 : \ \tau \lim_{t \to 0} \frac{T(t)x - x}{t^{\alpha}} = 0 \text{ and } \sup_{t > 0} \frac{\|T(t)x - x\|}{t^{\alpha}} < \infty \Big\}, \qquad (2.4.4)$$

and endow it with the norm  $\|\cdot\|_{F_{\alpha}}$ . We further equip  $F_{\alpha}$  and  $X_{\alpha}$  with the locally convex topology  $\tau_{F_{\alpha}}$  generated by the family of seminorms  $\mathcal{P}_{F_{\alpha}} := \{p_{F_{\alpha}} : p \in \mathcal{P}\}$ , where  $p_{F_{\alpha}}$  is defined as

$$p_{F_{\alpha}}(x) := \sup_{t>0} \frac{p(T(t)x - x)}{t^{\alpha}}.$$
(2.4.5)

It is easy to see that  $X_{\alpha}$  is a Banach space, i.e., as closed subspace of  $F_{\alpha}$ . By construction we have that indeed  $\underline{X}_{\alpha}(A) \subseteq X_{\alpha} \subseteq F_{\alpha}(A)$ . Next we discuss some properties of this space.

- **Lemma 2.46.** (a) Let  $(x_n)$  be a  $\|\cdot\|_{F_{\alpha}}$ -norm bounded sequence in  $F_{\alpha}$  with  $x_n \to x \in X_0$  in the topology  $\tau$ . Then  $x \in F_{\alpha}$ .
- (b) The triple  $(F_{\alpha}, \|\cdot\|_{F_{\alpha}}, \tau_{F_{\alpha}})$  satisfies the conditions in Assumption 1.1.
- (c)  $X_{\alpha}$  is bi-closed in  $F_{\alpha}$ , i.e., every  $\|\cdot\|_{F_{\alpha}}$ -bounded an  $\tau_{F_{\alpha}}$ -convergent sequence in  $X_{\alpha}$  has its limit in  $X_{\alpha}$ .

*Proof.* (a) The statement follows from the fact that the norm  $\|\cdot\|_{F_{\alpha}}$  is lower semicontinuous for the topology  $\tau$ . If

$$\frac{\|T(t)x_n - x_n\|}{t^{\alpha}} \le \|x_n\|_{F_{\alpha}} \le M$$

for each  $n \in \mathbb{N}$ , t > 0 and for some  $M \ge 0$  we can estimate:

$$\sup_{t>0} \frac{\|T(t)x - x\|}{t^{\alpha}} = \sup_{t>0} \sup_{p\in\mathcal{P}} p\Big(\frac{T(t)x - x}{t^{\alpha}}\Big) = \sup_{t>0} \sup_{p\in\mathcal{P}} \lim_{n\to\infty} p\Big(\frac{T(t)x_n - x_n}{t^{\alpha}}\Big)$$
$$\leq \sup_{t>0} \sup_{p\in\mathcal{P}} \limsup_{n\to\infty} \left\|\frac{T(t)x_n - x_n}{t^{\alpha}}\right\| \leq \sup_{t>0} \sup_{n\in\mathbb{N}} \left\|\frac{T(t)x_n - x_n}{t^{\alpha}}\right\| \leq M.$$

(b) We have for  $p \in \mathcal{P}$  and  $x \in F_{\alpha}$  that

$$p_{F_{\alpha}}(x) = \sup_{t>0} \frac{p(T(t)x - x)}{t^{\alpha}} \le \sup_{t>0} \frac{\|T(t)x - x\|}{t^{\alpha}} = \|x\|_{F_{\alpha}}.$$

This proves that  $\tau_{F_{\alpha}}$  is coarser than the  $\|\cdot\|_{F_{\alpha}}$ -topology, but is still Hausdorff by construction. For the second property of Assumption 1.1 let  $(x_n)_{n\in\mathbb{N}}$  be a  $\tau_{F_{\alpha}}$ -Cauchy sequence in  $F_{\alpha}$  such that there exists M > 0 with  $||x_n||_{F_{\alpha}} \leq M$  for each  $n \in \mathbb{N}$ . Since  $\tau$  is coarser than  $\tau_{F_{\alpha}}$ , we conclude that  $(x_n)$  is  $\tau$ -Cauchy sequence which is also bounded in  $|| \cdot ||_{F_{\alpha}}$ , hence in  $|| \cdot ||$ . By assumption there is  $x \in X_0$  such that  $x_n \to x$  in  $\tau$ . By part (a) we obtain  $x \in F_{\alpha}$ . It remains to prove that  $x_n \to x$  in  $\tau_{F_{\alpha}}$ . Let  $\varepsilon > 0$ , and take  $N \in \mathbb{N}$  such that for each  $n, m \in \mathbb{N}$  with  $n, m \geq N$  we have  $p_{F_{\alpha}}(x_n - x_m) < \varepsilon$ . For t > 0

$$p\Big(\frac{T(t)(x_n-x)-(x_n-x)}{t^{\alpha}}\Big) = \lim_{m \to \infty} p\Big(\frac{T(t)(x_n-x_m)-(x_n-x_m)}{t^{\alpha}}\Big) \le p_{F_{\alpha}}(x_n-x_m) < \varepsilon$$

for each  $n \ge N$ . Taking the supremum in t > 0 we obtain  $p_{F_{\alpha}}(x - x_n) \le \varepsilon$  for each  $n \ge N$ .

The norming property in (1.1.1) follows again from Remark 1.2 and the fact that the family  $\mathcal{P}$  is norming by assumption.

(c) Let  $(x_n)_{n \in \mathbb{N}}$  be a  $\|\cdot\|_{F_{\alpha}}$ -bounded and  $\tau_{F_{\alpha}}$  convergent sequence in  $X_{\alpha}$  with limit  $x \in X_0$ . For  $p \in \mathcal{P}$  we then have

$$\sup_{t>0} p\Big(\frac{T(t)(x_n-x)-(x_n-x)}{t^\alpha}\Big) \to 0.$$

Since  $x_n \in X_\alpha$  for each  $n \in \mathbb{N}$ , we have

$$\lim_{t \to 0} p\left(\frac{T(t)x_n - x_n}{t^{\alpha}}\right) = 0, \quad \text{and} \quad \sup_{t > 0} \left\|\frac{T(t)x_n - x_n}{t^{\alpha}}\right\| < \infty.$$

We now conclude for a fixed  $p \in \mathcal{P}$ 

$$p\left(\frac{T(t)x-x}{t^{\alpha}}\right) = p\left(\frac{T(t)(x-x_n) - (x-x_n) + T(t)x_n - x_n}{t^{\alpha}}\right)$$
$$\leq p\left(\frac{T(t)(x-x_n) - (x-x_n)}{t^{\alpha}}\right) + p\left(\frac{T(t)x_n - x_n}{t^{\alpha}}\right)$$
$$\leq p_{F_{\alpha}}(x-x_n) + p\left(\frac{T(t)x_n - x_n}{t^{\alpha}}\right) < \frac{\varepsilon}{2} + \frac{\varepsilon}{2} = \varepsilon,$$

where we first fix  $n \in \mathbb{N}$  such that  $p_{F_{\alpha}}(x-x_n) < \frac{\varepsilon}{2}$ , and then we take  $\delta > 0$  such that  $0 < t < \delta$ implies  $p(\frac{T(t)x_n - x_n}{t^{\alpha}}) < \frac{\varepsilon}{2}$ .

The next goal is to verify that  $(T(t))_{t\geq 0}$  can be restricted to  $X_{\alpha}$  to obtain a bi-continuous semigroup with respect to the topology  $\tau_{F_{\alpha}}$ .

**Lemma 2.47.** If  $(T(t))_{t\geq 0}$  is a bi-continuous semigroup, then  $X_{\alpha}$  is invariant under the semigroup.

*Proof.* We notice that in order to prove

$$\tau \lim_{s \to 0} \frac{T(s)x - x}{s^{\alpha}} = 0$$

we only have to check that

$$\frac{p(T(s_n)x - x)}{s_n^{\alpha}} \to 0$$

for  $n \to \infty$  for every null-sequence  $(s_n)_{n \in \mathbb{N}}$  in  $[0, \infty)$  and for each  $p \in \mathcal{P}$ . Let  $x \in X_{\alpha}$ . Then we have that  $y_n := \frac{T(s_n)x - x}{s_n^{\alpha}}$  converges to 0 with respect to  $\tau$  if  $(s_n)_{n \in \mathbb{N}}$  is any null-sequence and  $n \to \infty$ . Moreover, this sequence  $(y_n)_{n \in \mathbb{N}}$  is  $\|\cdot\|$ -bounded by the assumption that  $x \in X_{\alpha}$ . Whence we conclude

$$\tau \lim_{n \to \infty} T(t) y_n = \tau \lim_{n \to \infty} \frac{T(s_n) T(t) x - T(t) x}{s_n^{\alpha}} = 0,$$

so that  $T(t)x \in X_{\alpha}$ .

We now prove that  $(T(t))_{t\geq 0}$  is bi-continuous on  $X_{\alpha}$  and notice first that the local boundedness and the semigroup property are trivial.

**Lemma 2.48.** If  $(T(t))_{t\geq 0}$  is a bi-continuous semigroup on  $X_0$  and  $\alpha \in (0,1)$ , then  $(T(t))_{t\geq 0}$  is strongly  $\tau_{F_{\alpha}}$ -continuous on  $X_{\alpha}$ .

*Proof.* We have to show that  $p_{F_{\alpha}}(T(t_n)x-x) \to 0$  for all  $p \in \mathcal{P}$  whenever  $t_n \downarrow 0$ . Let  $s_n, t_n > 0$  be with  $s_n, t_n \to 0$ . Then

$$\frac{p(T(s_n)T(t_n)x - T(s_n)x - T(t_n)x + x)}{s_n^{\alpha}} \le \frac{p(T(t_n)T(s_n)x - T(t_n)x)}{s_n^{\alpha}} + \frac{p(T(s_n)x - x)}{s_n^{\alpha}} = \frac{p(T(t_n)(T(s_n)x - x))}{s_n^{\alpha}} + \frac{p(T(s_n)x - x)}{s_n^{\alpha}}.$$
 (2.4.6)

The sequence  $(y_n)$  given by  $y_n := \frac{T(s_n)x-x}{s_n^{\alpha}}$  is  $\|\cdot\|$ -bounded and  $\tau$ -convergent to 0, because  $x \in X_{\alpha}$ . So that the last term in the previous equation (2.4.6) converges to 0. But since  $\{T(t_n) : n \in \mathbb{N}\}$  is bi-equicontinuous, also the first term in (2.4.6) converges to 0. This proves strong continuity with respect to  $\tau_{F_{\alpha}}$ .

**Lemma 2.49.** Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup on  $X_0$ . Then  $(T(t))_{t\geq 0}$  is locally bi-equicontinuous on  $F_{\alpha}$ .

*Proof.* Let  $(x_n)_{n \in \mathbb{N}}$  be a  $\|\cdot\|_{F_{\alpha}}$ -bounded sequence which converges to zero with respect to  $\tau_{F_{\alpha}}$  and assume that  $(T(t)x_n)_{n \in \mathbb{N}}$  does not converge to zero uniformly for  $t \in [0, t_0]$  for some  $t_0 > 0$ . Hence there exists  $p \in \mathcal{P}$ ,  $\delta > 0$  and a sequence  $(t_n)_{n \in \mathbb{N}}$  of positive real numbers such that

$$p_{F_{\alpha}}(T(t_n)x_n) > \delta$$

for all  $n \in \mathbb{N}$ . As a consequence there exists a null-sequence  $(s_n)_{n \in \mathbb{N}}$  in  $\mathbb{R}$  such that

$$\frac{p(T(s_n)T(t_n)x_n - T(t_n)x_n)}{s_n^{\alpha}} > \delta$$

for each  $n \in \mathbb{N}$ . Now notice that the sequence  $(y_n)_{n \in \mathbb{N}}$  defined by  $y_n := \frac{T(s_n)x_n - x_n}{s_n^{\alpha}}$  is a  $\tau$ -null sequence since:

$$\frac{q(T(s_n)x_n - x_n)}{s_n^{\alpha}} \le \sup_{s>0} \frac{q(T(s)x_n - x_n)}{s^{\alpha}}, \quad q \in \mathcal{P},$$

and the term on the right hand side converges to zero as  $n \to \infty$  by assumption. Using the local bi-equicontinuity of the semigroup  $(T(t))_{t\geq 0}$  with respect to  $\tau$ , we conclude that  $\frac{T(t)T(s_n)x_n-T(t)x_n}{s_n}$  converges to zero uniformly for  $t \in [0, t_0]$ , which is a contradiction. Hence  $(T(t))_{t\geq 0}$  is locally bi-equicontinuous on  $X_{\alpha}$ .

**Remark 2.50.** Notice that the local bi-equicontinuity with respect to  $\tau_{F_{\alpha}}$  holds on the whole space  $F_{\alpha}$ , while strong  $\tau_{F_{\alpha}}$ -continuity holds on  $X_{\alpha}$  only. In particular, we will see in Theorem 2.52 that  $X_{\alpha}$  is the space of strong  $\tau_{F_{\alpha}}$ -continuity.

We can summarize the previous results in the following theorem.

**Theorem 2.51.** Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup on  $X_0$ . Then the restricted operators  $T_{\alpha}(t) := T(t)|_{X_{\alpha}}$  on  $X_{\alpha}$  form a bi-continuous semigroup. Moreover, the generator  $A_{\alpha}$  of  $(T_{\alpha}(t))_{t\geq 0}$  is the part of A in  $X_{\alpha}$ .

*Proof.* Because of the previous series of lemmas it remains to prove that the part of A in  $X_{\alpha}$  generates the restricted semigroup on  $X_{\alpha}$ . We can argue as in the proof of the proposition in [52, Chap. II, Par. 2.3]. Since the embedding  $X_{\alpha} \subseteq X_0$  is continuous for the topologies  $\tau_{F_{\alpha}}$  and  $\tau$ , we conclude that  $A_{\alpha} \subseteq A|_{X_{\alpha}}$ . For the converse let C denote the generator of  $(T_{\alpha}(t))_{t\geq 0}$  and take  $\lambda \in \mathbb{R}$  large enough such that

$$R(\lambda, C)x = \int_0^\infty e^{-\lambda s} T(s)x \, \mathrm{d}s = R(\lambda, A)x, \ x \in X_\alpha.$$

For  $x \in D(A|_{X_{\alpha}})$  we obtain

$$x = R(\lambda, A)(\lambda - A)x = R(\lambda, C)(\lambda - A)x \in D(C)$$

and hence  $A|_{X_{\alpha}} \subseteq A_{\alpha}$ . This proves that the part of A in  $X_{\alpha}$  generates the restricted semigroup.

By similar reasoning as in Lemma 2.31 one can prove the following:

**Theorem 2.52.** Let  $\alpha \in (0,1)$  and let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup on X. Then D(A) is  $\tau_{F_{\alpha}}$ -bi-dense in  $X_{\alpha}$  and

$$X_{\alpha} = \{ x \in F_{\alpha} : \ \tau_{F_{\alpha}} \lim_{t \to 0} T(t) x = x \},$$
(2.4.7)

i.e., for  $x \in F_{\alpha}$  the mapping  $t \mapsto T(t)x$  is  $\tau_{F_{\alpha}}$ -continuous if and only if  $x \in X_{\alpha}$ .

*Proof.* Denote by  $X_{\alpha,\text{cont}}$  the right-hand side of (2.4.7), i.e., the space of  $\tau_{F_{\alpha}}$ -strong continuity. Notice that  $D(A) \subseteq \underline{X}_{\alpha} \subseteq X_{\alpha} \subseteq X_{\alpha,\text{cont}}$ .

Suppose  $x \in X_{\alpha,\text{cont}}$ . For each  $n \in \mathbb{N}$  we have

$$x_n := n \int_0^{\frac{1}{n}} T_{\alpha}(t) x \, \mathrm{d}t = n \int_0^{\frac{1}{n}} T(t) x \, \mathrm{d}t \in D(A)$$

as a  $\tau$ - and  $\tau_{F_{\alpha}}$ -convergent Riemann integral. Whence it follows that  $x_n \xrightarrow{\tau_{F_{\alpha}}} x$ , whereas the  $\|\cdot\|_{F_{\alpha}}$ -boundedness of  $(x_n)_{n\in\mathbb{N}}$  clear. We conclude that  $x \in X_{\alpha}$  (because  $X_{\alpha}$  is bi-closed in  $F_{\alpha}$ ), implying  $X_{\alpha,\text{cont}} \subseteq X_{\alpha}$ . As a byproduct we also obtain that D(A) is bi-dense in  $X_{\alpha}$ .

**Proposition 2.53.** For  $0 \le \alpha < \beta \le 1$  we have

$$X_1 = D(A) \hookrightarrow F_\beta \hookrightarrow \underline{X}_\alpha \subseteq X_\alpha,$$

where the embeddings are continuous for the respective norms and for the respective topologies  $\tau_1, \tau_{F_{\beta}}, \tau_{F_{\alpha}}$ . The space D(A) bi-dense in  $X_{\alpha}$ , and as a consequence  $X_{\beta}$  is bi-dense in  $X_{\alpha}$ .

#### 2.4.4 Representation of Hölder spaces by generators

Analogously to Proposition 2.32 we have a representation of the Hölder space  $X_{\alpha}$  by means of the semigroup generator.

**Theorem 2.54.** Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup with negative growth bound and generator A. For  $\alpha \in (0, 1)$  we have

$$X_{\alpha} = \Big\{ x \in X_0 : \ \lim_{\lambda \to \infty} \lambda^{\alpha} AR(\lambda, A) x = 0 \ and \ \sup_{\lambda > 0} \|\lambda^{\alpha} AR(\lambda, A) x\| < \infty \Big\}.$$
(2.4.8)

*Proof.* Suppose  $x \in X_{\alpha}$ . From Proposition 2.32 we deduce immediately

$$\sup_{\lambda>0} \|\lambda^{\alpha} AR(\lambda, A)x\| < \infty.$$

Let now  $\varepsilon > 0$  be arbitrary. For  $x \in X_{\alpha}$  and  $p \in \mathcal{P}$  we can find  $\delta > 0$  such that  $0 \leq t < \delta$  implies  $\frac{p(T(t)x-x)}{t^{\alpha}} < \varepsilon$ . Recall the following formula:

$$\lambda^{\alpha} AR(\lambda, A)x = \lambda^{\alpha+1} \int_0^\infty e^{-\lambda s} (T(s)x - x) ds.$$

From this we deduce

$$\begin{split} p(\lambda^{\alpha}AR(\lambda,A)x) &\leq \lambda^{\alpha+1} \int_{0}^{\infty} e^{-\lambda s} \cdot \frac{p(T(s)x-x)}{s^{\alpha}} s^{\alpha} ds \\ &= \lambda^{\alpha+1} \int_{0}^{\delta} e^{-\lambda s} \cdot \frac{p(T(s)x-x)}{s^{\alpha}} s^{\alpha} ds + \lambda^{\alpha+1} \int_{\delta}^{\infty} e^{-\lambda s} \cdot \frac{p(T(s)x-x)}{s^{\alpha}} s^{\alpha} ds \\ &< \lambda^{\alpha+1} \varepsilon \int_{0}^{\delta} e^{-\lambda s} s^{\alpha} ds + \lambda^{\alpha+1} \int_{\delta}^{\infty} e^{-\lambda s} \cdot \frac{\|T(s)x-x\|}{s^{\alpha}} s^{\alpha} ds \\ &\leq \lambda^{\alpha+1} \varepsilon \int_{0}^{\delta} e^{-\lambda s} s^{\alpha} ds + \|x\|_{F_{\alpha}} \lambda^{\alpha+1} \int_{\delta}^{\infty} e^{-\lambda s} \cdot s^{\alpha} ds \\ &= \varepsilon \int_{0}^{\lambda\delta} e^{-t} t^{\alpha} dt + \|x\|_{F_{\alpha}} \int_{\lambda\delta}^{\infty} e^{-t} t^{\alpha} dt \\ &\leq L\varepsilon + \|x\|_{F_{\alpha}} \int_{\lambda\delta}^{\infty} e^{-t} t^{\alpha} dt \end{split}$$

where  $L := \int_0^\infty e^{-\lambda s} s^\alpha \, ds < \infty$ . Notice that the last part of the sum tends to zero if  $\lambda \to \infty$  since  $\delta > 0$  is fixed. So we obtain  $\tau \lim_{\lambda \to \infty} \lambda^\alpha AR(\lambda, A)x = 0$ .

For the converse inclusion suppose that

$$\tau \lim_{\lambda \to \infty} \lambda^{\alpha} A R(\lambda, A) x = 0 \text{ and } \sup_{\lambda > 0} \|\lambda^{\alpha} A R(\lambda, A) x\| < \infty,$$

the latter immediately implying  $||x||_{F_{\alpha}(T)} < \infty$  (see Proposition 2.32). We have to show that  $\tau \lim_{t \to 0} \frac{T(t)x-x}{t^{\alpha}} = 0$ . For  $\lambda > 0$  define  $x_{\lambda} = \lambda R(\lambda, A)$  and  $y_{\lambda} = AR(\lambda, A)$ , then we have

$$x = \lambda R(\lambda, A)x - AR(\lambda, A)x = x_{\lambda} - y_{\lambda}$$

First notice that for  $p \in \mathcal{P}$ 

$$\frac{p(T(t)x_{\lambda} - x_{\lambda})}{t^{\alpha}} \le \frac{1}{t^{\alpha}} p(T(t)\lambda R(\lambda, A)x - \lambda R(\lambda, A)x) \le \frac{\lambda^{1-\alpha}}{t^{\alpha}} \int_{0}^{t} p(T(s)\lambda^{\alpha}AR(\lambda, A)x) \, \mathrm{d}s.$$
(2.4.9)

By assumption the term  $\lambda^{\alpha}AR(\lambda, A)x$  is norm-bounded and converges in the topology  $\tau$  to zero as  $\lambda \to \infty$ , hence by the local bi-equicontinuity we conclude that  $p(T(s)\lambda^{\alpha}AR(\lambda, A)x) \to 0$  uniformly for  $s \in [0, 1]$ . Now let  $\varepsilon > 0$  and  $\lambda_0 > 1$  so large that for  $\lambda > \lambda_0$  and  $s \in [0, 1]$  we have  $p(T(s)\lambda^{\alpha}AR(\lambda, A)x) < \varepsilon$ . If  $t < \frac{1}{\lambda_0}$ , then  $\lambda := \frac{1}{t} > \lambda_0$  and we obtain that the expression in (2.4.9) becomes smaller than  $\varepsilon$ .

For the estimate of the second part involving  $y_{\lambda}$  we observe that

$$\frac{p(T(t)y_{\lambda} - y_{\lambda})}{t^{\alpha}} \leq \frac{1}{(t\lambda)^{\alpha}} p(T(t)\lambda^{\alpha}AR(\lambda, A)x) + \frac{1}{(t\lambda)^{\alpha}} p(\lambda^{\alpha}AR(\lambda, A)x).$$

By taking  $t < \frac{1}{\lambda_0}$  and  $\lambda := \frac{1}{t}$  we obtain the estimate

$$\frac{p(T(t)y_{\lambda} - y_{\lambda})}{t^{\alpha}} \le p(T(\frac{1}{\lambda})\lambda^{\alpha}AR(\lambda, A)x) + p(\lambda^{\alpha}AR(\lambda, A)x) < \varepsilon + \varepsilon,$$
(2.4.10)

by the choice of  $\lambda_0$ . Altogether we obtain for  $t < \frac{1}{\lambda_0}$  that  $\frac{p(T(t)x-x)}{t^{\alpha}} < 3\varepsilon$ , showing

$$\tau \lim_{t \to 0} \frac{T(t)x - x}{t^{\alpha}} = 0,$$

i.e.,  $x \in X_{\alpha}$  as required.

**Remark 2.55.** It is possible to define the space  $X_{\alpha}(A)$  as in (2.4.8) also for operators which are not necessarily generators of bi-continuous semigroups. However, we have to suppose that the resolvent fulfills certain continuity assumptions with respect to a topology satisfying, say, Assumption 1.1.

Again, we put our spaces  $X_{\alpha}$  in the general context of Theorem 2.16.

**Proposition 2.56.** For  $\alpha \in (0,1)$  and  $\mathcal{A}$ ,  $\lambda$  and  $\mathscr{E}$  as in Theorem 2.16 we have

$$X_{-\alpha} = \Big\{ (\lambda - \mathcal{A}) y \in X_{-1} : \sup_{t > 0} \frac{\|T(t)y - y\|}{t^{1-\alpha}} < \infty, \ \tau \lim_{t \to 0} \frac{T(t)y - y}{t^{1-\alpha}} = 0 \Big\}.$$

Finally, we extend the scale of spaces  $X_{\alpha}$  to the whole range  $\alpha \in \mathbb{R}$ .

**Definition 2.57.** For  $\alpha \in \mathbb{R} \setminus \mathbb{Z}$  we write  $\alpha = m + \beta$  with  $m \in \mathbb{Z}$  and  $\beta \in (0, 1]$ , and define

$$X_{\alpha}(A) := X_{\beta}(A_m),$$

with the corresponding norms. The locally convex topology on  $X_{\alpha}$  comes from  $X_{\beta}$  via the mapping  $A_m$ .

**Remark 2.58.** We summarize all previous results in the following diagram:



where  $\alpha \in (0, 1)$ . Here  $A_{\alpha-1}$  and  $\underline{A}_{\alpha-1}$  are defined to be the part of  $A_{-1}$  in  $X_{\alpha-1}$  and the part of  $\underline{A}_{-1}$  in  $\underline{X}_{\alpha-1}$ , respectively. They are all continuous with respect to the norms and topologies on these spaces. In addition, we recall that  $X_{\alpha-1}$  and  $\underline{X}_{\alpha-1}$  are the extrapolation spaces of  $X_{\alpha}(A_{-1})$  and  $\underline{X}_{\alpha}(A_{-1})$ , respectively. All horizontal arrows represent continuous inclusions, while the vertical arrows represent the action(s) of the semigroup(s). All the spaces are dense in the underlined ones containing them, while the spaces with underlining are bi-dense in each of the bigger ones.

# § 2.5 Examples

In this section we present examples for extrapolation and intermediate spaces for (generators of) bi-continuous semigroups. We will use Theorem 2.16 and its variants to identify the space  $X_{\alpha}$  for  $\alpha < 0$ .

#### 2.5.1 The left-translation semigroup

We now consider the left-translation semigroup from Section 1.3.1 and use Theorem 2.16 to determine the corresponding extrapolation spaces. To this purpose let  $\mathscr{E} = \mathscr{D}'(\mathbb{R})$  be the space of all distributions on  $\mathbb{R}$ , let  $\mathcal{A} := D : \mathscr{D}'(\mathbb{R}) \to \mathscr{D}'(\mathbb{R})$  be the distributional derivative, and let  $i : C_{\mathrm{b}}(\mathbb{R}) \to \mathscr{D}'(\mathbb{R})$  be the regular embedding. From Theorem 2.16 it then follows

$$\underline{X}_{-1} = \{ F \in \mathscr{D}'(\mathbb{R}) : F = f - Df \text{ for some } f \in UC_{\mathrm{b}}(\mathbb{R}) \}, \\ X_{-1} = \{ F \in \mathscr{D}'(\mathbb{R}) : F = f - Df \text{ for some } f \in C_{\mathrm{b}}(\mathbb{R}) \}.$$

For the Favard and Hölder spaces we have

$$F_{\alpha} = \left\{ f \in \mathcal{C}_{\mathbf{b}}(\mathbb{R}) : \sup_{\substack{x,y \in \mathbb{R} \\ x \neq y}} \frac{|f(x) - f(y)|}{|x - y|^{\alpha}} < \infty \right\} = \mathcal{C}_{\mathbf{b}}^{\alpha}(\mathbb{R}),$$

$$\underline{X}_{\alpha} = \left\{ f \in \mathrm{UC}_{\mathrm{b}}(\mathbb{R}) : \lim_{t \to 0} \sup_{\substack{x, y \in \mathbb{R} \\ 0 < |x-y| < t}} \frac{|f(x) - f(y)|}{|x-y|^{\alpha}} = 0 \right\} = \mathrm{h}_{\mathrm{b}}^{\alpha}(\mathbb{R}).$$

Hence  $F_{\alpha}$  is the space of bounded  $\alpha$ -Hölder-continuous functions and  $\underline{X}_{\alpha}$  with the so-called little Hölder space  $h_{b}^{\alpha}(\mathbb{R})$  (see also [85]). The abstract Hölder space  $X_{\alpha}$  corresponding to the bi-continuous semigroup yields the local version  $h_{b,loc}^{\alpha}(\mathbb{R})$  of the little Hölder space

$$\mathbf{h}_{\mathbf{b},\mathbf{loc}}^{\alpha} = \Big\{ f \in \mathbf{C}_{\mathbf{b}}^{\alpha}(\mathbb{R}) : \lim_{t \to 0} \sup_{\substack{x,y \in K \\ 0 < |x-y| < t}} \frac{|f(x) - f(y)|}{|x-y|^{\alpha}} = 0 \text{ for each } K \subseteq \mathbb{R} \text{ compact} \Big\}.$$

Then  $X_{\alpha} = h_{\mathrm{b,loc}}^{\alpha}(\mathbb{R}).$ 

It is easy to see  $\underline{X}_{\alpha} \subsetneq \overline{X}_{\alpha} \subsetneq F_{\alpha}$ . The extrapolated Favard class  $F_0$  can be identified with  $L^{\infty}(\mathbb{R})$ . To prove this we argue as follows: We know from the general theory that  $F_0(T) = (1-D)F_1(T)$  where  $F_1(T)$  are precisely the bounded Lipschitz functions on  $\mathbb{R}$ . Now using the fact that  $\operatorname{Lip}_{\mathrm{b}}(\mathbb{R}) = \mathrm{W}^{1,\infty}(\mathbb{R})$  with equivalent norms we obtain that indeed  $F_0 = L^{\infty}(\mathbb{R})$ . For an alternative proof of this fact we refer to [52, Chapter II.5(b)].

Moreover, from Corollary 2.26 we obtain for  $\alpha \in (0, 1)$ 

$$F_{-\alpha} = \Big\{ f \in \mathscr{D}'(\mathbb{R}) : \quad F = f - Df \text{ for } f \in \mathcal{C}_{\mathrm{b}}^{1-\alpha}(\mathbb{R}) \Big\},\$$

and

$$X_{-\alpha} = \Big\{ f \in \mathscr{D}'(\mathbb{R}) : \quad F = f - Df \text{ for } f \in \mathrm{h}^{1-\alpha}_{\mathrm{b,loc}}(\mathbb{R}) \Big\}.$$

We summarize this example by the diagram:

$$C^{1}_{b}(\mathbb{R}) \hookrightarrow \operatorname{Lip}_{b}(\mathbb{R}) \hookrightarrow h^{\alpha}_{b}(\mathbb{R}) \hookrightarrow h^{\alpha}_{b, \operatorname{loc}}(\mathbb{R}) \hookrightarrow C^{\alpha}_{b}(\mathbb{R}) \hookrightarrow \operatorname{UC}_{b}(\mathbb{R}) \hookrightarrow \operatorname{C}_{b}(\mathbb{R}) \hookrightarrow \operatorname{L}^{\infty}(\mathbb{R})$$

according to the abstract chain of spaces

$$X_1 \hookrightarrow F_1 \hookrightarrow \underline{X}_{\alpha} \hookrightarrow X_{\alpha} \hookrightarrow F_{\alpha} \hookrightarrow \underline{X}_0 \hookrightarrow X_0 \hookrightarrow F_0$$

for  $\alpha \in (0, 1)$ . For the higher order spaces we have

$$X_n := D(A^n) = \left\{ f \in \mathcal{C}_{\mathbf{b}}(\mathbb{R}) : f \text{ is } n \text{-times differentiable and } f^{(n)} \in \mathcal{C}_{\mathbf{b}}(\mathbb{R}) \right\}$$
$$= \left\{ f \in \mathcal{C}_{\mathbf{b}}(\mathbb{R}) : f^{(k)} \in \mathcal{C}_{\mathbf{b}}(\mathbb{R}), \ k = 1, \dots, n \right\} = \mathcal{C}_{\mathbf{b}}^n(\mathbb{R})$$

for  $n \in \mathbb{N}$ . For  $n \in \mathbb{N}$  and  $\alpha \in [0, 1)$ 

$$F_{n+\alpha} = \left\{ f \in \mathcal{C}^n_{\mathcal{b}}(\mathbb{R}) : \sup_{\substack{x,y \in \mathbb{R} \\ x \neq y}} \frac{|f^{(n)}(x) - f^{(n)}(y)|}{|x-y|^{\alpha}} < \infty \right\} = \mathcal{C}^{n,\alpha}_{\mathcal{b}}(\mathbb{R}).$$

This example complements the corresponding one in Nagel, Nickel, Romanelli [93, Sec. 3.2].

#### 2.5.2 The multiplication semigroup

Let  $\Omega$  be a locally compact space and  $X_0 = C_b(\Omega)$ . Let  $q : \Omega \to \mathbb{C}$  be continuous such that  $\sup_{x \in \Omega} \operatorname{Re}(q(x)) < 0$ . We define the multiplication operator  $M_q : D(M_q) \to C_b(\Omega)$  by  $M_q f = qf$  on the maximal domain

$$D(M_q) = \{ f \in \mathcal{C}_{\mathbf{b}}(\Omega) : qf \in \mathcal{C}_{\mathbf{b}}(\Omega) \}.$$

This operator generates the semigroup  $(T_q(t))_{t\geq 0}$  defined by

$$(T_q(t)f)(x) = e^{tq(x)}f(x), \quad t \ge 0, \ x \in \Omega, \ f \in \mathcal{C}_{\mathbf{b}}(\Omega),$$

which is bi-continuous on  $C_{\rm b}(\Omega)$  with respect to the compact-open topology. Now let  $\mathscr{E} = C(\Omega)$  the space of all continuous functions on  $\Omega$ , let  $\mathcal{M}_q : C(\Omega) \to C(\Omega)$  be the multiplication operator  $\mathcal{M}_q f := qf$  and  $i : C_{\rm b}(\Omega) \to C(\Omega)$  the identical embedding. Then by Theorem 2.16 we obtain

$$X_{-1} = \{g \in \mathcal{C}(\Omega) : q^{-1}g \in \mathcal{C}_{\mathbf{b}}(\Omega)\}.$$

For  $\alpha \in (0, 1)$ , the (abstract) Favard space is

$$F_{\alpha} = \{ f \in \mathcal{C}_{\mathbf{b}}(\Omega) : |q|^{\alpha} f \in \mathcal{C}_{\mathbf{b}}(\Omega) \}.$$

$$(2.5.1)$$

To see this suppose first that  $f \in F_{\alpha}$ , which means

$$\sup_{t>0} \sup_{x\in\Omega} \frac{|\mathrm{e}^{tq(x)}f(x) - f(x)|}{t^{\alpha}} < \infty.$$

By taking supremum only for  $t = \frac{1}{|q(x)|}$  we obtain

$$\sup_{x\in\Omega} \left| \mathrm{e}^{\frac{q(x)}{|q(x)|}} - 1 \right| \cdot |f(x)| \cdot |q(x)|^{\alpha} < \infty,$$

since

$$\frac{|e^{tq(x)}f(x) - f(x)|}{t^{\alpha}} = \frac{|e^{tq(x)} - 1| \cdot |f(x)||q(x)|^{\alpha}}{|q(x)|^{\alpha}t^{\alpha}}.$$
(2.5.2)

Hence  $|q|^{\alpha} f \in C_{b}(\Omega)$ , so that the inclusion " $\subseteq$ " in (2.5.1) is established. For the converse assume that  $|q|^{\alpha} f \in C_{b}(\Omega)$ . Since the function  $g(z) = \frac{|e^{z}-1|}{|z|^{\alpha}}$  is bounded on the left half-plane, we obtain that  $f \in F_{\alpha}$  by (2.5.2). This proves the equality. We also conclude that  $F_{\alpha} = X_{\alpha}$ since

$$\sup_{x \in K} \left| \frac{e^{tq(x)} f(x) - f(x)}{t^{\alpha}} \right| = \sup_{x \in K} \left| \frac{e^{tq(x)} - 1}{tq(x)} \right| \cdot |f(x)| \cdot |q(x)|^{\alpha} t^{1-\alpha}$$

for each compact set  $K \subseteq \Omega$ . The extrapolated Favard spaces are then given by

$$F_{-\alpha} = \{ f \in \mathcal{C}_{\mathbf{b}}(\Omega) : |q|^{1-\alpha} f \in \mathcal{C}_{\mathbf{b}}(\Omega) \} = X_{-\alpha}.$$

The spaces  $\underline{X}_{\alpha}$  are more difficult to describe in general since the space of strong continuity  $\underline{X}_0$  depends substantially on the choice of q. For example, if  $\frac{1}{q} \in C_0(\Omega)$ , then  $\underline{X}_0 = C_0(\Omega)$ .

To see this notice that  $C_0(\Omega) \subseteq \underline{X}_0$  trivially. On the other hand

$$|f| = \left|\frac{1}{q}\right| \cdot |fq|$$

which shows that  $D(M_q) \subseteq C_0(\Omega)$  and hence that  $\underline{X}_0 \subseteq C_0(\Omega)$ . For  $\alpha \in [0, 1]$  this yields

$$\underline{X}_{\alpha} = \{ f \in \mathcal{C}_0(\Omega) : |q|^{\alpha} f \in \mathcal{C}_0(\Omega) \}$$

and

$$\underline{X}_{-\alpha} = \{qf : f \in \mathcal{C}_0(\Omega), \ |q|^{1-\alpha} f \in \mathcal{C}_0(\Omega)\} = \{f \in \mathcal{C}(\Omega) : |q|^{-\alpha} f \in \mathcal{C}_0(\Omega)\}.$$

This example extends Section 3.2 in [93] by Nagel, Nickel and Romanelli.

#### 2.5.3 The Gauss–Weierstrass semigroup

On  $X_0 = C_b(\mathbb{R}^d)$   $(d \ge 1)$  we consider the Gauß–Weierstraß semigroup defined by T(0) = Iand

$$T(t)f(x) = \frac{1}{(4\pi t)^{\frac{d}{2}}} \int_{\mathbb{R}^d} e^{-\frac{|x-y|^2}{4t}} f(y) \, \mathrm{d}y, \quad t > 0, \ x \in \mathbb{R}^d.$$
(2.5.3)

If we equip  $C_b(\mathbb{R}^d)$  with the compact-open topology  $\tau_{co}$ , then  $(T(t))_{t\geq 0}$  becomes a bi-continuous semigroup, and its space of strong continuity is  $UC_b(\mathbb{R}^d)$ . From [82, Proposition 2.3.6] we know that the generator A of this semigroup is given  $Af = \Delta f$  on the maximal domain

$$D(A) = \{ f \in C_{\mathbf{b}}(\mathbb{R}^d) : \quad \Delta f \in C_{\mathbf{b}}(\mathbb{R}^d) \},\$$

where  $\Delta$  is the distributional Laplacian. Now the extrapolation space can again be obtained by Theorem 2.16. If we take  $\mathscr{E} = \mathscr{D}'(\mathbb{R}^d)$ ,  $\mathcal{A} = \Delta$  and  $i : C_{\mathrm{b}}(\mathbb{R}^d) \to \mathscr{D}'(\mathbb{R}^d)$  the regular embedding we then have

$$X_{-1} = \{ F \in \mathscr{D}'(\mathbb{R}^d) : F = f - \Delta f \text{ for some } f \in \mathcal{C}_{\mathbf{b}}(\mathbb{R}^d) \}$$

The domain of the generator can be given explicitly, see, e.g., [82] or [85]. For d = 1 it is

$$D(A) = \mathcal{C}^2_{\mathbf{b}}(\mathbb{R}),$$

while for  $d \geq 2$ 

$$D(A) = \left\{ f \in \mathcal{C}_{\mathbf{b}}(\mathbb{R}^d) \cap \mathcal{W}^{2,p}_{\mathrm{loc}}(\mathbb{R}^d), \text{ for all } p \in [1,\infty) \text{ and } \Delta f \in \mathcal{C}_{\mathbf{b}}(\mathbb{R}^d) \right\}.$$

For  $\alpha \in (0,1) \setminus \{\frac{1}{2}\}$  the Favard spaces are

$$F_{\alpha} = \mathcal{C}_{\mathbf{b}}^{2\alpha}(\mathbb{R}^d),$$

while for  $\alpha = \frac{1}{2}$  one obtains

$$F_{\frac{1}{2}} = \left\{ f \in \mathcal{C}_{\mathbf{b}}(\mathbb{R}^d) : \sup_{x \neq y} \frac{|f(x) + f(y) - 2f(\frac{x+y}{2})|}{|x-y|} < \infty \right\}$$

From Corollary 2.26 it follows that for  $\alpha \in (0, 1), \ \alpha \neq \frac{1}{2}$ 

$$F_{-\alpha} = \Big\{ F \in \mathscr{D}'(\mathbb{R}^d) : F = f - \Delta f \text{ for some } f \in \mathcal{C}_{\mathrm{b}}^{2(1-\alpha)}(\mathbb{R}^d) \Big\},\$$

and

$$F_{-\frac{1}{2}} = \Big\{ F \in \mathscr{D}'(\mathbb{R}^d): \ F = f - \Delta f \text{ for some } f \in F_{\frac{1}{2}} \Big\}.$$

# § 2.6 Notes

As already mentioned in the introduction this chapter is based the joint work with B. Farkas [28]. In this thesis we added Section 2.3.1 where we highlight the theory of interpolation spaces. In particular we give a proof of Proposition 2.37.

Another interesting example, which is originally part of [28], is now part of Chapter 6 where we discuss the implemented semigroups. A further application of extrapolation spaces is going to be the issue of the following Chapter 3, where we deal with so called fiberwise extrapolation spaces of unbounded operator-valued multiplication operators. An abstract usage of extrapolation spaces takes place in Chapter 4 in conjunction with perturbation theory.

In actual fact, the article [28], where this chapter is based on, came into being while working on the Desch–Schappacher perturbation which we present in Chapter 4. In particular, the absence of extrapolation spaces for non-densely defined Hille–Yosida operators was decisive for the emergence of this work. Even if we can construct extrapolation spaces by restricting to the space of strong continuity this was not adequate in order to stay in the category of bi-continuous semigroups. During the process of writing also intermediate spaces attract attention since they are commonly discussed in the same breath with extrapolation spaces, cf. [52, Chapter II, Sect. 5], hence we added this part of research to put it into a bigger picture.

# Chapter 3

# **Fiberwise Multiplication**

# Introduction

Non-autonomous problems arise naturally for example in the context of diffusion processes with time-dependent diffusion coefficients or boundary conditions or in connection with quantum mechanical systems with time-varying potential. They can be described by an abstract Cauchy problem on a Banach space X of the form

$$\begin{cases} \dot{u}(t) = A(t)u(t), & t, s \in \mathbb{R}, \ t \ge s, \\ u(s) = x, \end{cases}$$
(nACP)

where  $(A(t), D(A(t)))_{t \in \mathbb{R}}$  is a family of linear operators. The solution of such a problem, if it exists, is no longer given by a semigroup but a so-called evolution family  $(U(t, s))_{t \geq s}$ , see for example [52, Chapter VI, Def. 9.2]. The existence of solutions and the well-posedness of (nACP), cf. [52, Chapter VI, Def. 9.1] and [113, Def. 3.4.1], is a challenging topic. In particular, there are just a few different independent and not unified results on existence of solutions for the non-autonomous case due to P. Acquistapace and B. Terreni [2, 1] or T. Kato and H. Tanabe [111, 112, 72, 73]. Nevertheless, if we have a solution by means of an evolution family, we obtain a strongly continuous semigroup  $(T(t))_{t\geq 0}$  on the Bochner space  $L^p(\mathbb{R}, X)$ by

$$(T(t)f)(s) := U(s, s-t)f(s-t), \quad t \ge 0, \ f \in L^p(\mathbb{R}, X), \ s \in \mathbb{R}.$$
(3.0.1)

One important challenge is to determine the exact domain of the corresponding generator (G, D(G)). By [113, Thm. 3.4.7] operator-valued multiplication operators come up naturally since  $Gf = A(\cdot)f - f'$  on an invariant core  $D \subseteq W^{1,p}(\mathbb{R}, X) \cap D(A(\cdot))$ . However, one wants to have the generator (G, D(G)) explicitly in hand in order to infer properties of the evolution semigroup and hence of the evolution family. For this purpose, one has to consider extrapolation spaces of operator-valued multiplication operators. One of the attempts, in the special case where  $A(t) \equiv A$  for some semigroup generator (A, D(A)), is due to Nagel, Nickel and Romanelli [93, Sect. 4]. In [59] T. Graser studied bounded and unbounded operator-valued multiplication operators on the space of continuous functions  $C_0(\mathbb{R}, X)$  as well as their extrapolation spaces. We will see that extrapolation spaces of multiplication operators on  $L^p(\mathbb{R}, X)$  behave similarly. Later on, S. Thomaschewski studied properties of such multiplication operators on Bochner L<sup>p</sup>-spaces [113, Sect. 2.2 & 2.3] in connection with non-

autonomous problems. Especially, she connects multiplication semigroups with unbounded operator-valued multiplication operators. Although we are now interested in extrapolation spaces of such unbounded multiplication operators on Bochner  $L^p$ -spaces, we can use some of the results from S. Thomaschewski regarding multiplication operators. In order to construct extrapolation spaces, we recall the notion of fiber integrable functions due to R. Heymann [65].

We start this chapter with some preliminaries on fiber integrable functions and continue with unbounded multiplication operators in the second section. In Section 3.3 we discuss multiplication semigroups whose generators are actually multiplication operators. Furthermore, we determine the extrapolation spaces of such multiplication operators by means of  $L^p$ -fiber spaces.

# § 3.1 $L^p$ -fiber spaces

Firstly, we introduce the essential notion of a measurable Banach fiber set which was studied by R. Heymann, cf. [65, Def. VI.1.i]. To do so, let  $(\Omega, \Sigma, \mu)$  be a  $\sigma$ -finite measure space. Furthermore, let V be a complex vector space together with a family of seminorms  $\{||\cdot|||_s : s \in \Omega\}$ on V. Assume that there exists a countable set of elements  $\mathcal{B} := \{b_k : k \in \mathbb{N}\} \subseteq V$  such that  $\mathcal{B}$  is a vector space over  $\mathbb{Q}+i\mathbb{Q}$  and such that for each  $k \in \mathbb{N}$  the map  $s \mapsto |||b_k|||_s$  is measurable as a map from  $\Omega$  to  $\mathbb{R}$ . For every  $s \in \Omega$  we define the set  $N_s := \{b_k \in \mathcal{B} : |||b_k|||_s = 0\}$  and take the completion of the quotient space  $\mathcal{B}/N_s$  with respect to the induced norms  $\|\cdot\|_s$  on  $\mathcal{B}/N_s$ . This Banach space is denoted by  $X_s$ .

**Definition 3.1.** The family of Banach spaces  $(X_s, \|\cdot\|_s)_{s \in \Omega}$  is called a *measurable Banach* fiber set.

Next, we follow [65, Def. VI.1.iii] in order to define what it means for a function  $f : \Omega \to \bigcup_{s \in \Omega} X_s$  to be measurable

**Definition 3.2.** Let  $(X_s, \|\cdot\|_s)_{s\in\Omega}$  be a measurable Banach fiber set. We define a function  $f: \Omega \to \bigcup_{s\in\Omega} X_s$  with  $f(s) \in X_s$  for  $\mu$ -almost every  $s \in \Omega$  to be *fiber measurable* if it is almost everywhere a pointwise limit with respect to  $\|\cdot\|_s$  of measurable simple functions with values in  $\mathcal{B}$ . More precisely, this means that there exists a sequence  $(f_j)_{j\in\mathbb{N}}$  of functions  $f_j: \Omega \to \bigcup_{s\in\Omega} X_s$  such that

- 1.  $f_j = \sum_{i=1}^{n_j} (b_{\pi(i)} + N_s) \mathbf{1}_{\Omega_i}$ , where  $\pi : \mathbb{N} \to \mathbb{N}$ ,  $n_j \in \mathbb{N}$ ,  $\Omega_i \in \Sigma$ ,  $\Omega_i \cap \Omega_j = \emptyset$ ,  $i \neq j$ , and  $b_i \in \mathcal{B}$  for each  $1 \leq i \leq n_j$ ,
- 2.  $f(s) = \lim_{j \to \infty} f_j(s)$  with respect to  $\|\cdot\|_s$  for  $\mu$ -almost every  $s \in \Omega$ .

The set of fiber measurable functions on  $\Omega$  together with the pointwise addition and scalar multiplication is a  $\mathbb{C}$ -vector space, which we will call a *measurable Banach fiber space*.

Having the concept of measurability we continue with the notion of integrability for functions from  $\Omega$  to measurable Banach fiber sets, see [65, Def. VI.1.vi]. Especially, we define what it means to be *p*-integrable for  $1 \leq p < \infty$ . To do so, we remark that the map  $s \mapsto ||f(s)||_s^p$  is measurable.

**Definition 3.3.** Let  $1 \leq p < \infty$ . We call a fiber measurable function  $f : \Omega \to \bigcup_{s \in \Omega} X_s$  fiber *p*-integrable if the integral

$$\int_{\Omega} \|f(s)\|_s^p \, \mathrm{d}\mu(s)$$

exists and is finite. In this case, we call

$$\|f\|_p := \left(\int_{\Omega} \|f(s)\|_s^p \,\mathrm{d}\mu(s)\right)^{\frac{1}{p}}$$

the  $L^p$ -fiber norm of f.

- **Remark 3.4.** (i) Observe that the set of fiber *p*-integrable functions with pointwise addition and scalar multiplication is a vector space.
  - (ii) The relation defined by  $f \sim g \Leftrightarrow f = g \mu$ -almost everywhere is an equivalence relation on the set of fiber *p*-integrable functions.
- (iii) The set of equivalence classes of fiber *p*-integrable functions with the canonical vector space structure is called a  $L^p$ -fiber space and is denoted by  $L^p(\Omega, (X_s)_{s \in \Omega})$ .
- (iv) By [65, Prop. VI.1.xi] the space  $L^p(\Omega, (X_s)_{s \in \Omega})$  is actually a Banach space with respect to the  $L^p$ -fiber norm.

### § 3.2 Unbounded operator-valued multiplication operators

The main objects of this section are unbounded operator-valued multiplication operators, cf. [113, Def. 2.3.1].

**Definition 3.5.** Let X be a Banach space and let  $(M(s), D(M(s)))_{s \in \Omega}$  be a family of unbounded linear operators on X, i.e.,  $M(s) : D(M(s)) \subseteq X \to X$  for  $s \in \Omega$ . The operator  $(\mathcal{M}, D(\mathcal{M}))$  on  $L^p(\Omega, X)$  defined by

$$D(\mathcal{M}) := \{ f \in L^p(\Omega, X) : f(s) \in D(M(s)) \ \mu\text{-a.e.}, (s \mapsto M(s)f(s)) \in L^p(\Omega, X) \}, \\ (\mathcal{M}f)(s) := M(s)f(s), \quad f \in D(\mathcal{M}), s \in \Omega, \ \mu\text{-almost everywhere}, \end{cases}$$

is called the unbounded operator-valued multiplication operator. The operators (M(s), D(M(s))),  $s \in \Omega$ , are called *fiber operators*.

As already mentioned in the beginning, the concept of unbounded multiplication operators was studied by S. Thomaschewski [113] on Bochner L<sup>*p*</sup>-spaces in connection with nonautonomous Cauchy problems. Here we summarize some important results to have an overview of the properties of the operator  $(\mathcal{M}, D(\mathcal{M}))$ . Firstly, the closedness of the fiber operators implies the closedness of the multiplication operator, see [113, Lemma 2.3.4].

**Lemma 3.6.** If (M(s), D(M(s))) is closed for  $\mu$ -almost every  $s \in \Omega$ , then  $(\mathcal{M}, D(\mathcal{M}))$  is closed.

In what follows we actually assume that  $(\mathcal{M}, D(\mathcal{M}))$  is a closed operator valued multiplication operator with closed fiber operator  $(M(s), D(M(s)))_{s \in \Omega}$ . The following result [113, Lemma 2.3.5] shows that the resolvent operator of  $(\mathcal{M}, D(\mathcal{M}))$  gives also rise to an multiplication operator. For this we remind the reader of the following definition [113, Def. 2.2.3]:

 $\mathcal{L}^{\infty}(\Omega, \mathscr{L}_{s}(X)) := \{ M : \Omega \to \mathscr{L}(X) : s \mapsto M(s)x \in \mathcal{L}^{\infty}(\Omega, X) \text{ for all } x \in X \}.$ 

**Lemma 3.7.** Let  $(\mathcal{M}, D(\mathcal{M}))$  be a multiplication operator and assume that  $\lambda \in \rho(\mathcal{M})$ . Then  $R(\lambda, \mathcal{M})$  is a bounded multiplication operator, i.e., there exists  $M \in L^{\infty}(\Omega, \mathscr{L}_{s}(X))$ , such that  $(R(\lambda, \mathcal{M})f)(s) = M(s)f(s)$  for all  $f \in L^{p}(\Omega, X)$ .

Unfortunately, the converse of Lemma 3.7 fails due to the characterization of bounded multiplication operators, cf. [113, Thm. 2.2.17]. However, the following result [113, Thm. 2.3.6] holds.

**Theorem 3.8.** Let  $(\mathcal{M}, D(\mathcal{M}))$  be a densely defined closed operator on  $L^p(\Omega, X)$ . Assume that there exists an unbounded sequence  $(\lambda_n)_{n \in \mathbb{N}}$  in  $\rho(\mathcal{M})$  such that for all  $f \in L^p(\Omega, X)$  one has  $\lim_{n\to\infty} \lambda_n R(\lambda_n, \mathcal{M})f = f$ . If  $R(\lambda_n, \mathcal{M})$  is a bounded multiplication operator for every  $n \in \mathbb{N}$ , then there exists a family  $(\mathcal{M}(s), D(\mathcal{M}(s)))_{s \in \Omega}$  of densely defined closed operators on Xsuch that  $(\mathcal{M}, D(\mathcal{M}))$  is a multiplication operator with fiber operators  $(\mathcal{M}(s), D(\mathcal{M}(s)))_{s \in \Omega}$ . Furthere there exists a  $\mu$ -null-set  $\mathcal{N}$  such that for every  $s \in \Omega \setminus \mathcal{N}$  and for each  $n \in \mathbb{N}$  one has  $\lambda_n \in \rho(\mathcal{M}(s))$ .

Last but not least, if  $(\mathcal{M}, D(\mathcal{M}))$  is already supposed to be a multiplication operator on  $L^p(\Omega, X)$ , then the resolvent of  $\mathcal{M}$  and the resolvents of the fiber operators are related by the following result [113, Prop. 2.3.7].

**Proposition 3.9.** Let  $(\mathcal{M}, D(\mathcal{M}))$  be a closed multiplication operator with closed fiber operators  $(M(s), D(M(s)))_{s \in \Omega}$ .

- (a) If  $\lambda \in \rho(M(s))$  for  $\mu$ -almost every  $s \in \Omega$  and  $R(\lambda, M(\cdot)) \in L^{\infty}(\Omega, \mathscr{L}_{s}(X))$ , then  $\lambda \in \rho(\mathcal{M})$ and  $(R(\lambda, \mathcal{M})f)(s) = R(\lambda, M(s))f(s)$  for all  $f \in L^{p}(\Omega, X)$  and  $\mu$ -almost every  $s \in \Omega$ .
- (b) If there exists an unbounded sequence  $(\lambda_n)_{n\in\mathbb{N}}$  in  $\rho(\mathcal{M})$  such that for all  $f \in L^p(\Omega, X)$ one has  $\lambda_n R(\lambda_n, \mathcal{M}) f \to f$  for  $n \to \infty$ , then for  $\mu$ -almost all  $s \in \Omega$  and all  $n \in \mathbb{N}$  one has  $\lambda_n \in \rho(\mathcal{M}(s))$  and  $(R(\lambda_n, \mathcal{M})f)(s) = R(\lambda_n, \mathcal{M}(s))f(s)$  for all  $f \in L^p(\Omega, X)$  and  $\mu$ -almost every  $s \in \Omega$ .

In [113, Sect. 2.2.3] S. Thomaschewski proceeds with the discussion on multiplication operators on  $L^p(\Omega, X)$ . In particular, S. Thomaschewski gives a characterization of multiplication semigroups by means of multiplication operators as their generators.

**Definition 3.10.** A  $C_0$ -semigroup  $(\mathcal{T}(t))_{t\geq 0}$  on  $L^p(\Omega, X)$  is called a *multiplication semigroup* if for every  $t \geq 0$  the operator  $\mathcal{T}(t)$  is a bounded multiplication operator, i.e., for every  $t \geq 0$ there exists  $T_{(\cdot)}(t) \in L^{\infty}(\Omega, \mathscr{L}_{s}(X))$  such that  $(\mathcal{T}(t)f)(s) = T_s(t)f(s)$  for  $\mu$ -almost every  $s \in \Omega$ .

By [113, Thm. 2.3.15] these multiplication semigroups are characterized in the following way.

**Theorem 3.11.** Let  $(\mathcal{M}, D(\mathcal{M}))$  be the generator of a strongly continuous semigroup  $(\mathcal{T}(t))_{t\geq 0}$ on  $L^p(\Omega, X)$  such that  $\|\mathcal{T}(t)\| \leq M e^{\omega t}$  for some  $M \geq 0$ ,  $\omega \in \mathbb{R}$  and for all  $t \geq 0$ . The following are equivalent.

(a)  $(\mathcal{T}(t))_{t\geq 0}$  is a multiplication semigroup.

(b) (M, D(M)) is an unbounded operator valued multiplication operator with fiber operators (M(s), D(M(s)))<sub>s∈Ω</sub>. Moreover, for μ-almost every s ∈ Ω, λ ∈ ρ(M(s)) whenever Re(λ) > ω, (R(λ, M)f)(·) = R(λ, M(·))f(·) and (M(s), D(M(s))) is the generator of a C<sub>0</sub>-semigroup (T<sub>s</sub>(t))<sub>t>0</sub> such that (T(t)f)(s) = T<sub>s</sub>(t)f(s) for all t ≥ 0.

# § 3.3 Extrapolation of unbounded multiplication operators

In what follows we consider extrapolation spaces of unbounded multiplication fiber operators. In particular, our standing assumption is that the multiplication operator  $(\mathcal{M}, D(\mathcal{M}))$  generates a multiplication semigroup  $(\mathcal{T}(t))_{t\geq 0}$ , cf. Theorem 3.11. By  $(M(s), D(M(s)))_{s\in\Omega}$ we denote the fiber-operators corresponding to  $(\mathcal{M}, D(\mathcal{M}))$ , i.e.,  $(\mathcal{M}f)(s) = M(s)f(s)$ ,  $f \in L^p(\Omega, X), s \in \Omega$ . By Theorem 3.11 the operator  $(M(s), D(M(s))), s \in \Omega$ , generates a  $C_0$ -semigroup  $(T_s(t))_{t\geq 0}$ . We assume without loss of generality that  $0 \in \rho(M(s))$  for  $\mu$ almost every  $s \in \Omega$ . The extrapolated operators will be denoted by  $(M_{-1}(s), D(M_{-1}(s)))$ ,  $s \in \Omega$ . The extrapolation space of  $(\mathcal{M}, D(\mathcal{M}))$  will be denoted by  $\mathcal{X} := (L^p(\Omega, X))_{-1}(\mathcal{M})$  is formally given by

$$\mathcal{X} = \left\{ f \in \prod_{s \in \Omega} X_{-1,s} : \exists g \in \mathcal{L}^p(\Omega, X) : f = M_{-1}(\cdot)g \right\}$$

Later, we will prove another characterization of this space by means of  $L^p$ -fiber spaces. A first step towards this is the following.

**Lemma 3.12.** Let  $(\mathcal{M}, D(\mathcal{M}))$  be a multiplication operator on  $L^p(\Omega, X)$  with fiber operator  $(M(s), D(M(s)))_{s \in \Omega}$ . Moreover, let  $(\mathcal{T}(t))_{t \geq 0}$  the multiplication semigroup generated by  $(\mathcal{M}, D(\mathcal{M}))$ . Moreover, denote by  $(T_s(t))_{t \geq 0}$  the  $C_0$ -semigroups generated by the fiber operators  $(M(s), D(M(s)))_{s \in \Omega}$ . The associated extrapolated semigroups are denoted by  $(T_{-1,s}(t))_{t \geq 0}$ ;  $s \in \Omega$ . Define

$$(\mathcal{S}(t)f)(s) := T_{-1,s}(t)f(s), \quad t \ge 0, \ f \in \mathcal{X}, \ s \in \Omega.$$

This defines a  $C_0$ -semigroup on  $\mathcal{X}$  which is generated by the operator  $(\mathcal{M}_{-1}, D(\mathcal{M}_{-1}))$  defined by

$$(\mathcal{M}_{-1}f)(s) = M_{-1}(s)f(s), \quad D(\mathcal{M}_{-1}) = L^p(\Omega, X).$$
 (3.3.1)

Proof. First of all, to see that  $(\mathcal{S}(t))_{t\geq 0}$  is indeed a semigroup is easy, since  $(T_{-1,s}(t))_{t\geq 0}$  is a semigroup for each  $s \in \Omega$ . As a matter of fact,  $(T_{-1,s}(t))_{t\geq 0}$  is an extension of  $(T_s(t))_{t\geq 0}$  for each  $s \in \Omega$  and hence  $(\mathcal{S}(t))_{t\geq 0}$  extends  $(\mathcal{T}(t))_{t\geq 0}$ . Since by construction  $L^p(\Omega, X)$  is dense in  $\mathcal{X}$  the semigroup  $(\mathcal{S}(t))_{t\geq 0}$  is strongly continuous. In order to show that the generator of  $(\mathcal{S}(t))_{t\geq 0}$  is of the form mentioned in the lemma, let us denote the generator of  $(\mathcal{S}(t))_{t\geq 0}$  by  $(\mathcal{A}, D(\mathcal{A}))$ . Observe that by definition

$$(\mathcal{A}f)(s) = \lim_{t \to 0} \frac{\mathcal{S}(t)f(s) - f(s)}{t} = \lim_{t \to 0} \frac{T_{-1,s}(t)f(s) - f(s)}{t}.$$

This implies that  $f(s) \in D(M_{-1}(s))$  and  $s \in \Omega$  and that  $(\mathcal{A}f)(s) = M_{-1}(s)f(s)$  for  $\mu$ almost every  $s \in \Omega$ . In particular, we obtain  $f \in L^p(\Omega, X)$  and  $\mathcal{A}f = M_{-1}(\cdot)f$  since  $f(s) = (R(\lambda, M_{-1}(s))(\lambda - \mathcal{A})f)(s)$ . Conversely, the condition  $f \in L^p(\Omega, X)$  implies that  $f \in D(\mathcal{A})$  and  $\mathcal{A}f = M_{-1}(\cdot)f$ .

The previous result shows actually that the extrapolated multiplication operator is again a multiplication operator. In this case the fiber operators are the extrapolated fiber operators  $(M_{-1}(s), D(M_{-1}(s)))_{s \in \Omega}$ , i.e., (3.3.1) holds. As promised above, we give a characterization of the space  $\mathcal{X} := (L^p(\Omega, X))_{-1}(\mathcal{M})$ . To do so, we assume in this sequel that the Banach space X we are working with is separable, i.e., there exists a countable dense set in X. Denote the extrapolation spaces corresponding to the fiber operator  $(M(s), D(M(s))) \ge (X_{-1,s}, \|\cdot\|_{-1,s})$ ,  $s \in \Omega$ . The following result prepares for the extrapolation procedure.

**Lemma 3.13.** Suppose X is a separable Banach space. If  $0 \in \rho(M(s))$  for almost every  $s \in \Omega$  and  $s \mapsto M(s)^{-1}x$  is measurable for each  $x \in X$ , then the family  $(X_{-1,s}, \|\cdot\|_{-1,s})_{s \in \Omega}$  is a measurable Banach fiber set.

*Proof.* We make use of the separability of X and take a dense countable subset of X and make a  $\mathbb{Q}+i\mathbb{Q}$  vector space  $\mathcal{B}$  out of it. Observe that  $\mathcal{B}$  is still countable, i.e,  $\mathcal{B} := \{b_k : k \in \mathbb{N}\} \subseteq X$ . We define a family of seminorms  $\{||\cdot|||_s : s \in \Omega\}$  on X by

$$|||x|||_s := ||M(s)^{-1}x||, \quad x \in X, \ s \in \Omega.$$

Then  $\|\|\cdot\|\|_s$  is actually a norm on X. By the assumption  $s \mapsto M(s)^{-1}x$  is measurable for each  $x \in X$  and hence so is the map  $s \mapsto \|b_k\|_s$  for each  $k \in \mathbb{N}$ . Since  $N_s = \{0\}$  for each  $s \in \Omega$  we obtain  $\mathcal{B}/N_s = \mathcal{B}$ . Finally, the completion of  $\mathcal{B}$  with respect to  $\|\|\cdot\|\|_s$  is just the space  $X_{-1,s}$ ,  $s \in \Omega$ . By Definition 3.1 we therefore obtain that  $(X_{-1,s}, \|\cdot\|_{-1,s})_{s\in\Omega}$  is a measurable Banach fiber set.

Since we know that  $(X_{-1,s}, \|\cdot\|_{-1,s})_{s\in\Omega}$  is a measurable Banach fiber set, we can consider the space of fiber *p*-integrable functions over this set of Banach spaces. In what follows we relate this space to the extrapolation space of  $L^p(\Omega, (X_s)_{s\in\Omega})$  with respect to the unbounded operator-valued multiplication operator  $(\mathcal{M}, D(\mathcal{M}))$ .

**Theorem 3.14.** Let  $1 \leq p < \infty$  and consider the unbounded multiplication operator  $(\mathcal{M}, D(\mathcal{M}))$ on  $L^p(\Omega, X)$ , induced by the family of unbounded operators  $(M(s), D(M(s)))_{s \in \Omega}$  on X. Let (M(s), D(M(s))) be a semigroup generator for  $\mu$ -almost every  $s \in \Omega$ . Suppose that  $0 \in$  $\rho(M(s))$  for  $\mu$ -almost every  $s \in \Omega$  and that  $s \mapsto M(s)b$  and  $s \mapsto M(s)^{-1}b$  are measurable for each  $b \in \mathcal{B}$ . Then

$$\left[\mathrm{L}^{p}(\Omega, X)\right]_{-1}(\mathcal{M}) = \mathrm{L}^{p}(\Omega, (X_{-1,s})_{s \in \Omega}).$$

Proof. Let  $f \in [L^p(\Omega, X)]_{-1}(\mathcal{M})$  and find  $g \in L^p(\Omega, X)$  such that  $f = \mathcal{M}_{-1}g$ , where  $\mathcal{M}_{-1}$ :  $L^p(\Omega, X) \to L^p(\Omega, X)_{-1}(\mathcal{M})$ . Since g is measurable, we can find a sequence  $(g_n)_{n \in \mathbb{N}}$  of simple functions approximating g pointwise, i.e.,

$$g_n := \sum_{i=1}^{m_n} x_i \mathbf{1}_{\Omega_i}$$
 and  $g_n \to g \ \mu$ -almost everywhere.

In order to show that  $f \in L^p(\Omega, (X_{-1,s})_{s \in \Omega})$  define  $f_n := \mathcal{M}_{-1}g_n$ . Then

$$f_n(s) := (\mathcal{M}_{-1}g_n)(s) = \sum_{i=1}^{m_n} M_{-1}(s) x_i \mathbf{1}_{\Omega_i}(s),$$
where we use Lemma 3.12 as well as the fact that  $M_{-1}(s)x_i \in X_{-1,s}$ ,  $s \in \Omega$ . However, in general  $M_{-1}(s)x_i \in X_{-1,s}$  is not an element of  $\mathcal{B}$ , cf. Lemma 3.13. To bypass this problem observe that  $\mathcal{B}$  is dense in  $X_{-1,s}$  for each  $s \in \Omega$ . For that reason take  $\varepsilon > 0$  arbitrarily and find  $b_{k_i} \in \mathcal{B}$  such that  $\|b_{k_i} - M_{-1}(s)x_i\|_{-1,s} < \varepsilon$  for  $\mu$ -almost every  $s \in \Omega$  and define

$$\widetilde{f}_n := \sum_{i=1}^{m_n} b_{k_i} \mathbf{1}_{\Omega_i}$$

Observe, that since the measurable sets  $\Omega_i$  are disjoint, we have

$$\left\|f_n(s) - \widetilde{f}_n(s)\right\|_{-1,s} = \begin{cases} \left\|M_{-1}(s)x_j - b_{k_j}\right\|_{-1,s}, & \text{if } s \in \Omega_j, \\ 0, & \text{if } s \notin \Omega_i \text{ for all } 1 \le i \le m_n, \end{cases}$$

 $\mu$ -almost everywhere and hence by the choice of  $b_{k_i} \in \mathcal{B}$  that  $\|f_n(s) - \tilde{f}_n(s)\|_{-1,s} < \varepsilon$  for  $\mu$ -almost every  $s \in \Omega$ . We now show that  $f(s) = \lim_{j \to \infty} \tilde{f}_j(s)$  with respect to  $\|\cdot\|_{-1,s}$  for  $\mu$ -almost every  $s \in \Omega$ . To do so, we observe that by construction we can find  $N \in \mathbb{N}$  such that for all  $n \geq N$ :

$$\|f(s) - f_n(s)\|_{-1,s} = \|(\mathcal{M}_{-1}g)(s) - (\mathcal{M}_{-1}g_n)(s)\|_{-1,s} = \|g(s) - g_n(s)\| < \varepsilon, \quad s \in \Omega,$$

whence,

$$\left\| f(s) - \tilde{f}_n(s) \right\|_{-1,s} \le \| f(s) - f_n(s) \|_{-1,s} + \left\| f_n(s) - \tilde{f}_n(s) \right\|_{-1,s} < 2\varepsilon.$$

Since  $\varepsilon > 0$  was arbitrary, we conclude that f is fiber measurable. Furthemore,

$$\|f\|_{\mathcal{L}^{p}(\Omega,(X_{-1,s})_{s\in\Omega})}^{p} = \int_{\Omega} \|f(s)\|_{-1,s}^{p} \, \mathrm{d}s = \int_{\Omega} \|g(s)\|^{p} \, \mathrm{d}s = \|g\|_{\mathcal{L}^{p}(\Omega,X)}^{p} < \infty,$$

showing that f is a fiber p-integrable function, i.e.,  $f \in L^p(\Omega, (X_{-1,s})_{s \in \Omega})$ . For the converse

inclusion, suppose that  $f \in L^p(\Omega, (X_{-1,s})_{s \in \Omega})$ . We have to show that there exists  $g \in L^p(\Omega, X)$ such that  $f = \mathcal{M}_{-1}g$ . Since f is fiber measurable, there exists a sequence  $(f_j)_{j \in \mathbb{N}}$  of simple functions  $f_j : \Omega \to \bigcup_{s \in \Omega} X_{-1,s}$  with  $f(s) \in X_{-1,s}$  for  $\mu$ -almost every  $s \in \Omega$  and

$$f_j = \sum_{i=1}^{n_j} b_{k_i} \mathbf{1}_{\Omega_i},$$

where  $b_{k_i} \in \mathcal{B}$ ,  $\Omega_i \in \Sigma$ ,  $\Omega_i \cap \Omega_j = \emptyset$ ,  $i \neq j$ , for  $1 \leq i \leq n_j$ , and

$$||f(s) - f_j(s)||_{-1,s} \to 0,$$
 (3.3.2)

for  $j \to \infty$  and  $\mu$ -almost every  $s \in \Omega$ . By the assumption that  $0 \in \rho(M(s))$  for  $\mu$ -almost every  $s \in \Omega$  we conclude by Proposition 3.9 that  $0 \in \rho(\mathcal{M})$ . So we define

$$g_j := (\mathcal{M}_{-1}^{-1} f_j)(\cdot) = \sum_{i=1}^{n_j} \left( M_{-1}^{-1}(\cdot) b_{k_i} \right) \mathbf{1}_{\Omega_i}(\cdot).$$

We observe that  $M_{-1}^{-1}(s)b_{k_i} \in X$  for  $\mu$ -almost every  $s \in \Omega$  and hence  $g_j$  is a simple function. By (3.3.2) we conclude that  $(g_j(s))_{j\in\mathbb{N}}$  is a Cauchy sequence with respect to  $\|\cdot\|_s$  for  $\mu$ -almost every  $s \in \Omega$  and hence convergent. This yields a measurable function  $g : \Omega \to X$  by taking the pointwise limit, i.e.,

$$g(s) := \lim_{j \to \infty} g_j(s), \quad s \in \Omega$$

By the continuity of  $M_{-1}(s)$ ,  $s \in \Omega$ , on X and the fact that  $M_{-1}(s)M_{-1}^{-1}(s) = I$  for  $\mu$ -almost every  $s \in \Omega$ , we directly obtain that  $\mathcal{M}_{-1}g = f$ . Moreover,

$$\|g\|_{\mathrm{L}^{p}(\Omega,X)}^{p} = \int_{\Omega} \|g(s)\|^{p} \, \mathrm{d}s = \int_{\Omega} \|f(s)\|_{-1,s}^{p} \, \mathrm{d}s = \|f\|_{\mathrm{L}^{p}(\Omega,(X_{-1,s})_{s\in\Omega}}^{p} < \infty,$$

and therefore  $g \in L^p(\Omega, X)$ .

# § 3.4 Notes

The content of this chapter was worked out in cooperation with R. Heymann during the authors' research stay at the University of Stellenbosch (South Africa) in summer 2018. The research idea of this chapter was developed during a discussion between B. Farkas, R. Heymann and the author after a talk of R. Heymann at the Functional Analysis Seminar of our workgroup at the Bergische Universität Wuppertal in November 2017. The first contact with R. Heymann was built during the last AGFA meeting in Blaubeuren in December 2016.

Observe that in Lemma 3.12 and Theorem 3.14 we considered the case when for almost every  $t \in \mathbb{R}$  the operator (A(t), D(A(t))) is the generator of a strongly continuous semigroup on X. However, this assumption is not too restrictive since the generation results by P. Acquistapace and B. Terreni as well as T. Kato and H. Tanabe, mentioned in the introduction, assume that  $(A(t), D(A(t)))_{t \in \mathbb{R}}$  are in fact generators of analytic semigroups.

As already mentioned in the introduction, a possible application of our results it the determination of the explicit domain of the evolution semigroup corresponding (nACP). The idea how to find the explicit domain of the evolution family is to enlarge the space by extrapolation and obtain the generator domain by taking the part of the extrapolated generator in the original space. This appears to be possible only in connection with some uniformity conditions on the extrapolation spaces corresponding to the operators  $(A(t), D(A(t)))_{t \in \mathbb{R}}$  as they are for example mentioned by H. Amann [7, Sect. 7] and J. Kisyński [74]. To be more exact,  $X_{-1,s} \cong X_{-1,s} \cong X_{-1}$  for all  $s \in \mathbb{R}$  such that there exists a constant  $\kappa > 0$  such that  $\frac{1}{\kappa} \|x\|_{X_{-1}} \leq \|x\|_{X_{-1,s}} \leq \kappa \|x\|_{X_{-1}}$  for all  $x \in X_{-1}$  and  $t \in \mathbb{R}$ .

# Chapter 4

# **Desch–Schappacher Perturbations**

As suggested by G. Greiner in [60] abstract perturbation theory of one-parameter semigroups provides good means to change the domain of a semigroup generator. For this an enlargement of the underlying Banach space may be necessary and extrapolation spaces become important. One of the well-known results in this direction goes back to the papers of W. Desch and W. Schappacher, see [38] and [39]. Another prominent example of such general perturbation techniques is due to Staffans and Weiss [108, 107], and an elegant abstract operator theoretic/algebraic approach has been developed by Adler, Bombieri and Engel in [3]. A general theory of unbounded domain perturbations is given by Hadd, Manzo and Rhandi [64]. A more recent paper by Bátkai, Jacob, Voigt and Wintermayr [20] extends the notion of positivity to extrapolation spaces, and studies positive perturbations for positive semigroups on AM-spaces. Hence, the study of abstract Desch–Schappacher type perturbations is a lively research field, to which we contribute with the present article. The reason for such an active interest in this area is that the range of application is vast. We mention here only a selection from the most recent ones: boundary perturbations by Nickel [96], boundary feedback by Casarino, Engel, Nagel and Nickel [31], boundary control by Engel, Kramar Fijavž, Klöss, Nagel and Sikolya [49] and Engel and Kramar Fijavž [47], port-Hamiltonian systems by Baroun and Jacob [19], control theory by Jacob, Nabiullin, Partington and Schwenninger [68, 67] and Jacob, Schwenninger and Zwart [69] and vertex control in networks by Engel and Kramar Fijavž [50, 48].

All the previously mentioned abstract perturbation results were developed for strongly continuous semigroups of linear operators on Banach spaces,  $C_0$ -semigroups for short. This is, for certain applications, e.g., for the theory of Markov transition semigroups, far too restrictive. For this situation the Banach space of bounded and continuous functions over a Polish space is the most adequate, but on this space the strong continuity with the respect to norm is, in general, a too stringent requirement.

F. Kühnemund in [78] has developed the abstract theory of bi-continuous semigroups, which has the advantage that not only Markov transition semigroup, but also semigroups induced by jointly continuous flows or implemented semigroups, just to mention a few, can be handled in a unified manner. Some perturbation result for bi-continuous semigroups are known, see [55, 56, 54], however, none of which is suitable for domain perturbations.

As first step this chapter treats a Desch–Schappacher type perturbation theorem for this class of semigroups. Since the theory of bi-continuous semigroups uses a Banach space norm and a additional locally convex topology, it is fundamental to relate our results to the existing, analogous ones on locally convex spaces. We recall the next result, due to Jacob, Wegner, Wintermayr, from [70].

**Theorem.** Let X be a sequentially complete, locally convex space with fundamental system  $\Gamma$  of continuous seminorms, and let (A, D(A)) be the generator of a locally equicontinuous  $C_0$ -semigroup  $(T(t))_{t\geq 0}$  on X. Moreover, let  $\overline{X}$  be a sequentially complete locally convex space such that

- (a)  $X \subseteq \overline{X}$  is dense and the inclusion map is continuous,
- (b)  $\overline{A}$  with domain  $D(\overline{A}) = X$  generates a locally equicontinuous  $C_0$ -semigroup  $(\overline{T}(t))_{t\geq 0}$  on  $\overline{X}$  such that  $\overline{T}(t)_{|X} = T(t)$  holds for all  $t \geq 0$ .

Let  $B: X \to \overline{X}$  be a linear and continuous operator and  $t_0 > 0$  be a number such that

(c) 
$$\forall f \in C([0, t_0], X) : \int_0^{t_0} \overline{T}(t_0 - t) Bf(t) dt \in X,$$
  
(d)  $\forall p \in \Gamma \exists K \in (0, 1) \ \forall f \in C([0, t_0], X) : p\left(\int_0^{t_0} \overline{T}(t_0 - t) Bf(t) dt\right) \leq K \cdot \sup_{t \in [0, t_0]} p(f(t)).$ 

Then the operator (C, D(C)) defined by

$$Cx = (\overline{A} + B)x$$
 for  $x \in D(C) = \left\{x \in X : (\overline{A} + B)x \in X\right\}$ 

generates a locally equicontinuous  $C_0$ -semigroup on X if and only if  $D(C) \subseteq X$  is dense.

We will prove a similar result for bi-continuous semigroups with the advantage that we can relax condition (c) of the previous theorem in the sense that we allow different seminorms on the left- and the right-hand side of the inequality. Moreover, one has to change and expand the conditions for the bi-continuous case carefully to obtain a good interplay between the Banach space norm and the locally convex topology. A space  $\overline{X}$  with the properties used in the theorem above is called an extrapolation space. For  $C_0$ -semigroups on Banach spaces the classical construction is presented in [52, Chapter II, Sect. 5a] in a self-contained manner. Extrapolation spaces for  $C_0$ -semigroups on locally convex spaces are constructed by Wegner in [123]. Extrapolated bi-continuous semigroups and extrapolation spaces were treated in Chapter 2.

This chapter is organized as follows. In the first section we recall some definitions and results for bi-continuous semigroups and give some preliminary constructions needed for the Desch– Schappacher perturbation result, which is stated and proved as Theorem 4.4 in Section 4.1. Section 4.2 contains a sufficient condition for operators to satisfy the hypothesis of the abstract perturbation theorem, see Theorem 4.7. In Section 4.3 we prove that for a large class of bounded functions  $g : \mathbb{R} \to \mathbb{C}$  which are continuous up to a discrete set of jump discontinuities and for each bounded (complex) Borel measure  $\mu$  on  $\mathbb{R}$  the operator

$$Cf := f' + \int_{\mathbb{R}} f \, \mathrm{d}\mu \cdot g$$

with appropriate domain generates a bi-continuous semigroup on the Banach space  $C_b(\mathbb{R})$  of bounded, continuous functions on  $\mathbb{R}$ .

#### § 4.1 An abstract Desch–Schappacher perturbation result

Let us first fix some notation. Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup on a Banach space  $X_0$  with respect to  $\tau$  and with generator (A, D(A)), where  $\tau$  is generated by the family of seminorms  $\mathcal{P}$ . Furthermore, let  $B \in \mathscr{L}(X_0, X_{-1})$  such that  $B : (X_0, \tau) \to (X_{-1}, \tau_{-1})$  is a continuous linear operator and define for  $t_0 > 0$  the following space:

$$\mathfrak{X}_{t_0} := \left\{ F : [0, t_0] \to \mathscr{L}(X) \middle| \begin{array}{l} F \text{ is } \tau \text{-strongly continuous, norm bounded} \\ \text{and} \{F(t) : t \in [0, t_0]\} \text{ is bi-equicontinuous} \end{array} \right\}.$$
(4.1.1)

**Remark 4.1.** In [55, Lemma 3.2] it was shown that for  $t_0 > 0$  the space  $\mathfrak{X}_{t_0}$  is indeed a Banach space (and in particular a Banach algebra) with respect to the norm

$$||F|| := \sup_{t \in [0,t_0]} ||F(t)||.$$

For  $F \in \mathfrak{X}_{t_0}$  and  $t \in [0, t_0]$  we define the so-called *(abstract) Volterra operator*  $V_B$  on  $\mathfrak{X}_{t_0}$  by

$$(V_B F)(t)x := \int_0^t T_{-1}(t-r)BF(r)x \, \mathrm{d}r.$$
(4.1.2)

The integral has to be understood in the sense of a  $\tau_{-1}$ -Riemann integral. Notice that in general for  $x \in X_0$  we have  $(V_B F)(t) x \in X_{-1}$ . For the formulation of our main result we need the following definition.

**Definition 4.2.** Let  $B \in \mathscr{L}(X_0, X_{-1})$  such that also  $B : (X_0, \tau) \to (X_{-1}, \tau_{-1})$  is continuous. The operator B is said to be *admissible*, if there is  $t_0 > 0$  such that the following conditions are satisfied:

- 1.  $V_B F(t) x \in X_0$  for all  $t \in [0, t_0]$  and  $x \in X_0$ .
- 2.  $\operatorname{Ran}(V_B) \subseteq \mathfrak{X}_{t_0}$ .
- 3.  $||V_B|| < 1.$

The set of all admissible operators  $B: (X_0, \tau) \to (X_{-1}, \tau_{-1})$  will be denoted by  $\mathcal{S}_{t_0}^{DS, \tau}$ . We write  $B \in \mathcal{S}_{t_0}^{DS, \tau}(T)$  whenever it is important to emphasize for which semigroup  $(T(t))_{t \geq 0}$  the operator B is admissible.

**Remark 4.3.** By construction  $A_{-1}: (X_0, \tau) \to (X_{-1}, \tau_{-1})$  is continuous, and actually an isomorphism. In particular, we have

$$\forall p \in \mathcal{P} \; \exists L > 0 \; \exists \gamma \in \mathcal{P}_{-1} \forall x \in X_0 : \; p(x) \le L(\gamma(x) + \gamma(A_{-1}x)).$$

This section contains the formulation of the Desch–Schappacher type perturbation result and its proof. Observe that the proof of the following theorem is at some points verbatim the same as the one of [52, Chapter III, Thm. 3.1]. **Theorem 4.4.** Let (A, D(A)) be the generator of a bi-continuous semigroup  $(T(t))_{t\geq 0}$  on a Banach space  $X_0$  with respect to  $\tau$ . Let  $B: X_0 \to X_{-1}$  such that  $B \in \mathcal{S}_{t_0}^{DS,\tau}$  for some  $t_0 > 0$ . Then the operator  $(A_{-1} + B)_{|X_0}$  with domain

$$D((A_{-1}+B)_{|X_0}) := \{x \in X_0 : A_{-1}x + Bx \in X_0\}$$

generates a bi-continuous semigroup  $(S(t))_{t\geq 0}$  on  $X_0$  with respect to  $\tau$ . Moreover, the semigroup  $(S(t))_{t\geq 0}$  satisfies the variation of parameters formula

$$S(t)x = T(t)x + \int_0^t T_{-1}(t-r)BS(r)x \, \mathrm{d}r, \qquad (4.1.3)$$

for every  $t \ge 0$  and  $x \in X_0$ .

*Proof.* Since  $||V_B|| < 1$  by hypothesis, we conclude that  $1 \in \rho(V_B)$ . Now let t > 0 be arbitrary and write  $t = nt_0 + t_1$  for  $n \in \mathbb{N}$  and  $t_1 \in [0, t_0)$ . Define

$$S(t) := ((R(1, V_B)T_{|[0,t_0]})^n(t_0) \cdot (R(1, V_B)T_{|[0,t_0]})(t_1).$$

We first show that  $(S(t))_{t\geq 0}$  is a semigroup. For  $0 \leq s, t \leq s+t \leq t_0$  and  $n \in \mathbb{N}$  we prove the following identity (cf. [52, p. 184])

$$(V_B^n T)(t+s) = \sum_{k=0}^n (V_B^{n-k} T_{|[0,t_0]})(s) \cdot (V_B^k T)(t), \quad \forall n \in \mathbb{N}$$
(4.1.4)

by induction. We abbreviate  $V := V_B$ . Since  $V^0 = I$ , equation (4.1.4) is trivially satisfied for n = 0. Now assume that (4.1.4) is true for some  $n \in \mathbb{N}$ . Then we obtain by this hypothesis that

$$\begin{split} \sum_{k=0}^{n+1} (V^{n+1-k}T)(s) \cdot (V^kT)(t) \\ &= \sum_{k=0}^n \left( \int_0^s T_{-1}(s-r)BV^{n-k}T(r) \, \mathrm{d}r \right) \cdot V^kT(t) + T(s) \int_0^t T_{-1}(t-r)BV^nT(r) \, \mathrm{d}r \\ &= \int_0^s T_{-1}(s-r)B\sum_{k=0}^n V^{n-k}T(r) \cdot V^kT(t) \, \mathrm{d}r + \int_0^t T_{-1}(s+t-r)BV^nT(r) \, \mathrm{d}r \\ &= \int_0^s T_{-1}(s-r)BV^nT(r+t) \, \mathrm{d}r + \int_0^t T_{-1}(s+t-r)BV^nT(r) \, \mathrm{d}r \\ &= \int_t^{s+t} T_{-1}(s+t-r)BV^nT(r) \, \mathrm{d}r + \int_0^t T_{-1}(s+t-r)BV^nT(r) \, \mathrm{d}r \\ &= V^{n+1}T(s+t). \end{split}$$

By this we can conclude that  $(S(t))_{t\geq 0}$  satisfies the semigroup law for  $0 \leq s, t \leq s + t \leq t_0$ . Indeed, for each  $t \in [0, t_0]$  the point evaluation  $\delta_t : \mathfrak{X}_{t_0} \to \mathscr{L}(X_0)$  is a contraction and since ||V|| < 1 by hypothesis the inverse of I - V is given by the Neumann series. Therefore,

$$S(t) = \delta_t \left( \sum_{n=0}^{\infty} V^n T \right) = \sum_{n=0}^{\infty} (V^n T)(t), \quad t \in [0, t_0].$$

Moreover, we have

$$\|(V^n T)(t)\| \le \|V^n\| \cdot \|T_{[0,t_0]}\|,$$

and we conclude that the series above converges absolutely. Hence

$$\begin{split} S(s)S(t) &= \sum_{n=0}^{\infty} \, (V^n T)(s) \cdot \sum_{n=0}^{\infty} \, (V^n T)(t) \\ &= \sum_{n=0}^{\infty} \sum_{k=0}^{n} \, (V^{n-k} T)(s)(V^k T)(t) \\ &= \sum_{n=0}^{\infty} \, (V^n T)(s+t) = S(s+t). \end{split}$$

Now we show that S(t)S(s) = S(t+s) for all t, s > 0. For that let t, s > 0 be arbitrary and  $n, m \in \mathbb{N}$  and  $t_1, t_2 \in [0, t_0)$  such that  $t = nt_0 + t_1$  and  $s = mt_0 + t_2$ . Then we obtain the following

$$\begin{split} S(t)S(s) &= S(t_0)^n S(t_1) S(t_0)^m S(t_2) \\ &= S(t_0)^n S(t_0)^m S(t_1) S(t_2) \\ &= \begin{cases} S(t_0)^{n+m} S(t_1+t_2), & \text{if } t_1+t_2 < t_0, \\ S(t_0)^{n+m+1} S(t_2-(t_0-t_1)), & \text{if } t_1+t_2 \ge t_0. \end{cases} \end{split}$$

But in both cases the right-hand side equals S(t + s) by definition. Hence  $(S(t))_{t\geq 0}$  satisfies the semigroup law. The next step is to show that it is a bi-continuous semigroup with respect to  $\tau$ . Notice that

$$S_{|[0,t_0]}(t) = R(1, V_B)T_{|[0,t_0]}(t)$$

and hence  $(S(t))_{t\geq 0}$  is locally bounded and the set  $\{S(t): t \in [0, t_0]\}$  is bi-equicontinuous. For t > 0 let  $m := \lfloor \frac{t}{t_0} \rfloor$  and notice that  $\{S(t_0)^k: 1 \leq k \leq m\}$  is bi-equicontinuous, hence we conclude that the set

$$\left\{ S(t_0)^k : 1 \le k \le m \right\} \cdot \left\{ S(s) : s \in [0, t_0] \right\}$$

is also bi-equicontinuous. By definition of  $(S(t))_{t\geq 0}$  we obtain  $\tau$ -strong continuity, and hence  $(S(t))_{t\geq 0}$  is a bi-continuous semigroup with respect to  $\tau$ . We now prove

$$S(t)x = T(t)x + \int_0^t T_{-1}(t-r)BS(r)x \, \mathrm{d}r$$

for each t > 0 and  $x \in X_0$  by proceeding similarly to [52, Chapter III, Sect. 3]. For  $t = nt_0+t_1$ ,  $n \in \mathbb{N}$  and  $t_1 \in [0, t_0)$ , we obtain:

$$\int_{0}^{t} T_{-1}(t-r)BS(r) dr$$
  
=  $\sum_{k=0}^{n-1} \int_{kt_0}^{(k+1)t_0} T_{-1}(t-r)BS(r) dr + \int_{nt_0}^{t} T_{-1}(t-r)BS(r) dr$ 

$$=\sum_{k=0}^{n-1} T_{-1}(t - (k+1)t_0) \int_0^{t_0} T_{-1}(t_0 - r)BS(r) \, \mathrm{d}r \cdot S(kt_0) + \int_0^{t_1} T_{-1}(t_1 - r)BS(r) \, \mathrm{d}r \cdot S(nt_0)$$
$$=\sum_{k=0}^{n-1} T(t - (k+1)t_0)(S(t_0) - T(t_0))S(kt_0) + (S(t_1) - T(t_1))S(nt_0) = S(t) - T(t).$$

The next step is to show that the resolvent set of  $(A_{-1} + B)_{|X_0}$  is non-empty. For this we claim that  $R(\lambda, A_{-1})B$  is bounded with  $||R(\lambda, A_{-1})B|| < 1$  for  $\lambda$  large enough. Choose  $M \ge 0$  and  $\omega \in \mathbb{R}$  such that  $||T(t)|| \le M e^{\omega t}$  for all t > 0. Then for  $\lambda > \omega$  we obtain:

$$R(\lambda, A_{-1})B = \int_0^\infty e^{\lambda r} T_{-1}(r)B \, dr = \sum_{n=0}^\infty e^{-\lambda n t_0} T(nt_0) (V_B F_\lambda)(t_0)$$

where  $F_{\lambda}(r) := e^{-\lambda(t_0-r)}I \in \mathfrak{X}_{t_0}$ . From this we obtain the following estimate:

$$||R(\lambda, A_{-1})B|| \le ||V_B|| + \frac{M e^{(\omega - \lambda)t_0}}{1 - e^{(\omega - \lambda)t_0}} ||V_B||.$$

Since  $||V_B|| < 1$  we conclude for sufficient large  $\lambda$ :

$$||R(\lambda, A_{-1})B|| < 1.$$

This yields  $1 \in \rho(R(\lambda, A_{-1})B)$  for large  $\lambda$  and then invertibility of  $\lambda - (A_{-1} + B)|_{X_0}$ , since

$$\lambda - (A_{-1} + B)|_{X_0} = (\lambda - A)(I - R(\lambda, A_{-1})B)$$

Hence the resolvent set of  $(A_{-1} + B)_{|X_0}$  contains each sufficiently large  $\lambda$ . In the last step we will show that  $(A_{-1} + B)_{|X_0}$  is actually the generator of the bi-continuous semigroup  $(S(t))_{t\geq 0}$  with respect to  $\tau$ . Denote by (C, D(C)) the generator of  $(S(t))_{t\geq 0}$ . Let  $\lambda > \max(\omega_0(T), \omega_0(S))$ , then by the variation of constant formula (4.1.3), the resolvent representation as Laplace transform [79, Lemma 7] and the fact that we may interchange the improper  $\tau$ -Riemann integral and the  $\tau_{-1}$ -Riemann integral by an application of [78, Lemma 1.7] we obtain

$$R(\lambda, C) = R(\lambda, A) + R(\lambda, A_{-1})BR(\lambda, C).$$

Whence we conclude

$$(I - R(\lambda, A_{-1})B)R(\lambda, C) = R(\lambda, A),$$

and therefore

$$I = (\lambda - A)(I - R(\lambda, A_{-1})B)R(\lambda, C) = (\lambda - (A_{-1} + B)|_{X_0})R(\lambda, C).$$

It follows that  $C \subseteq (A_{-1} + B)_{|X_0}$  and by the previous observations  $C = (A_{-1} + B)_{|X_0}$ .  $\Box$ 

#### 4.1.1 Abstract Favard Spaces and comparison of semigroups

In this subsection we want to combine the perturbation theory with the theory of abstract Favard spaces as described in Section 2.2. In the next proposition we show that Desch– Schappacher perturbations of bi-continuous semigroups, which satisfy a special range condition concerning the extrapolated Favard class, gives us semigroups which are close to each other in some sense.

**Proposition 4.5.** Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup on  $X_0$  with respect to  $\tau$  generated by (A, D(A)). Suppose that  $B \in \mathcal{S}_{t_0}^{DS,\tau}$  with  $\operatorname{Ran}(B) \subseteq F_0(A)$  and let  $(S(t))_{t\geq 0}$  be the perturbed semigroup. Then there exists  $C \geq 0$  such that for each  $t \in [0, 1]$  one has

$$||T(t) - S(t)|| \le Ct.$$

Proof. We may assume  $\omega_0(T) < 0$ . We find  $M \ge 0$  such that  $||T(t)|| \le M$  and  $||S(t)|| \le M$  for every  $t \in [0, 1]$ . Since  $\operatorname{Ran}(B) \subseteq F_0(A)$  we conclude that  $A_{-1}^{-1}B : X_0 \to F_1(A)$ . Hence  $A_{-1}^{-1}B$ is bounded by the closed graph theorem and we find  $K \ge 0$  such that  $||A_{-1}^{-1}Bx||_{F_1(A)} \le K ||x||$ for each  $x \in X_0$ . Let  $\mathcal{P}_{-1}$  the family of seminorms corresponding to the first extrapolation space (see Section 2.4.2). By using (4.1.3) we obtain

$$\begin{split} \|S(t)x - T(t)x\| &= \left\| A_{-1} \int_{0}^{t} T(t-r) A_{-1}^{-1} BS(r)x \, \mathrm{d}r \right\| \\ &= \left\| \tau_{-1} \lim_{h \to 0} \frac{T_{-1}(h) - I}{h} \int_{0}^{t} T(t-r) A_{-1}^{-1} BS(r)x \, \mathrm{d}r \right\| \\ &= \left\| \tau_{-1} \lim_{h \to 0} \int_{0}^{t} \frac{T(h) - I}{h} T(t-r) A_{-1}^{-1} BS(r)x \, \mathrm{d}r \right\| \\ &= \sup_{p \in \mathcal{P}_{-1}} \lim_{h \to 0} p \left( \int_{0}^{t} \frac{T(h) - I}{h} T(t-r) A_{-1}^{-1} BS(r)x \, \mathrm{d}r \right) \\ &\leq \sup_{p \in \mathcal{P}_{-1}} \lim_{h \to 0} \int_{0}^{t} p \left( \frac{T(h) - I}{h} T(t-r) A_{-1}^{-1} BS(r)x \right) \, \mathrm{d}r \\ &\leq \lim_{h \to 0} \int_{0}^{t} \left\| \frac{T(h) - I}{h} T(t-r) A_{-1}^{-1} BS(r)x \right\| \, \mathrm{d}r \\ &\leq M \int_{0}^{t} \left\| A_{-1}^{-1} BS(r)x \right\|_{F_{1}(A)} \, \mathrm{d}r \\ &\leq t K M^{2} \cdot \|x\| \end{split}$$

for each  $x \in X_0$  and  $t \in [0, 1]$ .

**Corollary 4.6.** Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup on  $X_0$  with respect to  $\tau$  generated by (A, D(A)). If  $B \in \mathcal{S}_{t_0}^{DS,\tau}$  and  $\operatorname{Ran}(B) \subseteq F_0(A)$ , then the perturbed semigroup  $(S(t))_{t\geq 0}$  leaves the space of strong continuity  $\underline{X}_0 := \overline{D(A)}^{\|\cdot\|}$  invariant.

#### § 4.2 Admissible operators

Next we consider a sufficient condition for  $B : (X_0, \tau) \to (X_{-1}, \tau_{-1})$  to be admissible. Throughout this section we denote the space of continuous functions  $f : [0, t_0] \to (X_0, \tau)$  which are  $\|\cdot\|$ -bounded by  $C_b([0, t_0], (X_0, \tau))$ . If equipped with the sup-norm,  $C_b([0, t_0], (X_0, \tau))$ becomes a Banach space. The proof of the following theorem is at points verbatim the same as the one of [52, Chapter III, Cor. 3.3]. **Theorem 4.7.** Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup with generator (A, D(A)) on a Banach space  $X_0$  with respect to  $\tau$ . Let  $\mathscr{P}$  be the set of generating continuous seminorms corresponding to  $\tau$ . Let  $B \in \mathscr{L}(X_0, X_{-1})$  such that  $B : (X_0, \tau) \to (X_{-1}, \tau_{-1})$  is a linear and continuous operator, and let  $t_0 > 0$  be such that

(a) 
$$\int_{0}^{t_0} T_{-1}(t_0 - r)Bf(r) \, \mathrm{d}r \in X_0 \text{ for each } f \in \mathcal{C}_{\mathrm{b}}([0, t_0], (X_0, \tau)).$$

(b) For every  $\varepsilon > 0$  and every  $p \in \mathcal{P}$  there exists  $q \in \mathcal{P}$  and K > 0 such that for all  $f \in C_b([0, t_0], (X_0, \tau))$ 

$$p\left(\int_{0}^{t_{0}} T_{-1}(t_{0}-r)Bf(r) \, \mathrm{d}r\right) \leq K \cdot \sup_{r \in [0,t_{0}]} |q(f(r))| + \varepsilon \, \|f\|_{\infty} \,. \tag{4.2.1}$$

(c) There exists  $M \in (0,1)$  such that for all  $f \in C_b([0,t_0],(X_0,\tau))$ 

$$\left\| \int_{0}^{t_{0}} T_{-1}(t_{0} - r)Bf(r) \, \mathrm{d}r \right\| \leq M \, \|f\|_{\infty} \,. \tag{4.2.2}$$

Then  $B \in \mathcal{S}_{t_0}^{DS,\tau}$ , and as a consequence the operator  $(A_{-1}+B)_{|X_0}$  defined on the domain

$$D((A_{-1}+B)_{|X_0}) := \{x \in X_0 : A_{-1}x + Bx \in X_0\}$$

generates a bi-continuous semigroup with respect to  $\tau$ .

*Proof.* We first show  $\operatorname{Ran}(V_B) \subseteq \mathfrak{X}_{t_0}$ . Let  $f \in \operatorname{C_b}([0, t_0], (X_0, \tau))$  and define for  $t \in [0, t_0]$  the auxiliary function  $f_t : [0, t_0] \to X_0$  by

$$f_t(r) := \begin{cases} f(0), & r \in [0, t_0 - t], \\ f(r + t - t_0), & r \in [t_0 - t, t_0]. \end{cases}$$

Then  $f_t \in C_b([0, t_0], (X_0, \tau))$  and

$$\int_0^t T_{-1}(t-r)Bf(r) \, \mathrm{d}r = \int_0^{t_0} T_{-1}(t_0-r)Bf_t(r) \, \mathrm{d}r - \int_t^{t_0} T_{-1}(r)Bf(0) \, \mathrm{d}r.$$
(4.2.3)

By Theorem 1.7

$$\int_{t}^{t_0} T_{-1}(r)Bf(0) \, \mathrm{d}r = T(t) \int_{0}^{t_0-t} T_{-1}(r)Bf(0) \, \mathrm{d}r \in D(A_{-1}) = X_0$$

We conclude that the map  $\psi : [0, t_0] \to X_{-1}$  defined by

$$\psi(t) := \int_0^t T_{-1}(t-r)Bf(r) \,\mathrm{d}r \tag{4.2.4}$$

has values in  $X_0$ . Moreover, for  $\varepsilon > 0$  and  $p \in \mathcal{P}$  we have the following estimate:

$$p(\psi(t) - \psi(s))$$

$$\begin{split} &= p\left(\int_{0}^{t} T_{-1}(t-r)Bf(r) \, \mathrm{d}r - \int_{0}^{s} T_{-1}(s-r)Bf(r) \, \mathrm{d}r\right) \\ &\leq p\left(\int_{0}^{t_{0}} T_{-1}(t_{0}-r)B(f_{t}(r)-f_{s}(r)) \, \mathrm{d}r\right) + p\left(\int_{s}^{t} T_{-1}(r)Bf(0) \, \mathrm{d}r\right) \\ &\leq K \cdot \sup_{r \in [0,t_{0}]} q(f_{t}(r)-f_{s}(r)) + p\left(\int_{s}^{t} T_{-1}(r)Bf(0) \, \mathrm{d}r\right) + \varepsilon \, \|f_{t}-f_{s}\|_{\infty} \\ &\leq K \cdot \sup_{r \in [0,t_{0}]} q(f_{t}(r)-f_{s}(r)) \\ &+ L \cdot \left(\gamma \left(\int_{s}^{t} T_{-1}(r)Bf(0) \, \mathrm{d}r\right) + \gamma \left((T_{-1}(t)-T_{-1}(s))Bf(0)\right)\right) + \varepsilon \, \|f_{t}-f_{s}\| \\ &\leq K \cdot \sup_{r \in [0,t_{0}]} |q(f_{t}(r)-f_{s}(r))| \\ &+ L \cdot \left(\int_{s}^{t} \gamma (T_{-1}(r)Bf(0)) \, \mathrm{d}r + \gamma \left((T_{-1}(t)-T_{-1}(s))Bf(0)\right)\right) + 2\varepsilon \, \|f\|_{\infty} \end{split}$$

where the  $\gamma \in \mathcal{P}_{-1}$  of the second to last inequality comes from Remark 4.3. The extrapolated semigroup  $(T_{-1}(t))_{t\geq 0}$  is strongly  $\tau_{-1}$ -continuous and  $\gamma \in \mathcal{P}_{-1}$ , so that we can find  $\delta_1 > 0$  such that

$$\gamma(T_{-1}(t) - T_{-1}(s)Bf(0)) < \varepsilon$$
 whenever  $|t - s| < \delta_1$ .

Moreover, f is  $\tau$ -continuous and therefore uniformly  $\tau$ -continuous on compact sets, which gives us  $\delta_2 > 0$  such that

$$\sup_{r\in[0,t_0]} |q(f_t(r) - f_s(r))| < \varepsilon \text{ if } |t-s| < \delta_2.$$

Last but not least,  $\gamma(T_{-1}(r)Bf(0))$  is bounded by some constant M > 0, so for  $\delta_3 = \frac{\varepsilon}{M}$  we have

$$|s-t| < \delta_3 \implies \int_s^t \gamma(T_{-1}(r)Bf(0)) \, \mathrm{d}r < \varepsilon.$$

Now, we take  $\delta := \min \{\delta_1, \delta_2, \delta_3\}$  and obtain

$$p(\psi(t) - \psi(s)) < (K + 2 \|f\|_{\infty} + 2L)\varepsilon,$$

showing that  $\psi : [0, t_0] \to X_0$  is  $\tau$ -continuous.

Next, we prove the norm-boundedness using the same techniques and arguments as in [52, Chapter III, Sect. 3]. Let  $f \in C_b([0, t_0], (X_0, \tau))$  and write

$$f = \tilde{f}_{\delta} + h_{\delta},$$

where

$$h_{\delta}(r) := \begin{cases} \left(1 - \frac{r}{\delta}\right) f(r), & 0 \le r < \delta, \\ 0, & \delta \le r \le t_0 \end{cases}$$

for some  $\delta > 0$ . Then  $\tilde{f}_{\delta}$  and  $h_{\delta}$  are norm-bounded and continuous with respect to  $\tau$ ,  $\tilde{f}_{\delta}(0) = 0$ 

and  $\|\widetilde{f}_{\delta}\|_{\infty} \leq \|f\|_{\infty}$ . Now we obtain

$$\begin{split} \left\| \int_{0}^{t} T_{-1}(t-r)Bf(r) \, \mathrm{d}r \right\| &\leq \left\| \int_{0}^{t} T_{-1}(t-r)B\tilde{f}_{\delta}(r) \, \mathrm{d}r \right\| + \left\| \int_{0}^{t} T_{-1}(t-r)Bh_{\delta}(r) \, \mathrm{d}r \right\| \\ &\leq M \left\| \tilde{f}_{\delta} \right\|_{\infty} + K \left( \left\| \int_{0}^{t} T_{-1}(t-r)Bh_{\delta}(r) \, \mathrm{d}r \right\|_{-1} + \left\| A_{-1} \int_{0}^{t} T_{-1}(t-r)Bh_{\delta}(r) \, \mathrm{d}r \right\|_{-1} \right) \\ &\leq M \left\| \tilde{f}_{\delta} \right\|_{\infty} + K \left\| \int_{0}^{\delta} T_{-1}(t-r) \left( 1 - \frac{r}{\delta} \right) Bf(0) \, \mathrm{d}r \right\|_{-1} \\ &+ K \left\| T_{-1}(t)Bf(0) - \frac{1}{\delta} \int_{0}^{\delta} T_{-1}(t-r)Bf(0) \, \mathrm{d}r \right\|_{-1}. \end{split}$$

By taking  $\delta \searrow 0$  we obtain

$$\left\| \int_{0}^{t} T_{-1}(t-r)Bf(r) \, \mathrm{d}r \right\| \le M \, \|f\|_{\infty} \,. \tag{4.2.5}$$

We proceed with showing local bi-equicontinuity. For that let  $(x_n)_{n\in\mathbb{N}}$  be a norm-bounded  $\tau$ -null-sequence. Let  $\varepsilon > 0$  and  $p \in \mathscr{P}$ , then by taking  $f^n(r) = f(r)x_n$ , we can find  $q \in \mathcal{P}$  such that

$$p(V_BF(t)x_n) = p\left(\int_0^t T_{-1}(t-r)BF(r)x_n \, dr\right)$$
  

$$\leq p\left(\int_0^{t_0} T_{-1}(t_0-r)Bf^n(r) \, dr - \int_t^{t_0} T_{-1}(r)Bf^n(0) \, dr\right)$$
  

$$\leq K \cdot \sup_{r \in [0,t_0]} |q(f^n(r))| + p\left(\int_t^{t_0} T_{-1}(r)Bf^n(0) \, dr\right) + \varepsilon \|f_t^n\|$$
  

$$\leq K \cdot \sup_{r \in [0,t_0]} |q(f^n(r))| + \varepsilon \|f_t^n\|$$
  

$$+ L \cdot \left(\gamma \left(\int_t^{t_0} T_{-1}(r)Bf^n(0) \, dr\right) + \gamma \left((T_{-1}(t_0) - T_{-1}(t)) Bf^n(0)\right)\right).$$

Now we can argue by the local bi-equicontinuity of  $(T(t))_{t\geq 0}$  and  $(T_{-1}(t))_{t\geq 0}$  and with the arbitrarily small  $\varepsilon > 0$  to conclude the local bi-equicontinuity of  $V_B F$ . Hence we see that  $V_B$  maps  $\mathfrak{X}_{t_0}$  to  $\mathfrak{X}_{t_0}$  and by (4.2.5) that  $||V_B|| < 1$  since by assumption  $M \in (0, 1)$ .

## § 4.3 Perturbations of the translation semigroup

In this section we want to give an application of the Desch–Schappacher perturbation result to an explicit example. Another example is part of Chapter 6. Recall from Section 1.3.1 that on the space  $X_0 = C_b(\mathbb{R})$  the left-translation semigroup  $(T(t))_{t\geq 0}$  is bi-continuous with respect to the compact-open topology  $\tau_{co}$ . Moreover, the resulting extrapolation spaces, in the notation we used in Chapter 2, are given by (cf. Section 2.5.1):

$$\underline{X}_{-1} = \{ F \in \mathscr{D}'(\mathbb{R}) : F = f - Df \text{ for some } f \in UC_{\mathrm{b}}(\mathbb{R}) \}, \\ X_{-1} = \{ F \in \mathscr{D}'(\mathbb{R}) : F = f - Df \text{ for some } f \in C_{\mathrm{b}}(\mathbb{R}) \}.$$

where  $\mathrm{UC}_{\mathrm{b}}(\mathbb{R})$  denotes the space of bounded uniformly continuous functions and Df the distributional derivative of f. Recall that the generator of  $(T(t))_{t\geq 0}$  is  $A = \frac{\mathrm{d}}{\mathrm{d}x}$  with domain  $D(A) := \mathrm{C}_{\mathrm{b}}^{1}(\mathbb{R})$ , and  $A_{-1} = D$  with domain  $D(A_{-1}) = \mathrm{C}_{\mathrm{b}}(\mathbb{R})$ , where D denotes the distributional derivative. The extrapolated semigroup  $(T_{-1}(t))_{t\geq 0}$  is the restriction to  $X_{-1}$  of the left-translation semigroup on the space  $\mathscr{D}'(\mathbb{R})$  of distributions. Consider the function  $g: \mathbb{R} \to \mathbb{R}$  defined by

$$g(x) = \begin{cases} 0, & x \le -1, \ x > 1, \\ x, & -1 < x \le 0, \\ 2 - x, & 0 < x \le 1. \end{cases}$$
(4.3.1)

The graph of this function is the following.



Notice that  $g \in X_{-1}$ , since g = h - Dh where h is the tent function on the real line defined by



We now construct an operator  $B \in \mathscr{L}(X, X_{-1})$  satisfying all conditions of Theorem 4.7, i.e.,  $((A_{-1}+B)_{|X_0}, D((A_{-1}+B)_{|X_0}))$  generates a bi-continuous semigroup. For this purpose let  $\mu$  be a bounded regular Borel measure on  $\mathbb{R}$  and define the continuous functional  $\Phi : C_b(\mathbb{R}) \to \mathbb{R}$  by  $\Phi(f) = \int_{\mathbb{R}} f \, d\mu$  and the operator  $B : X_0 \to X_{-1}$  by

$$Bf := \Phi(f)g.$$

This operator B is by construction continuous with respect to the local convex topologies on the spaces  $X_0$  and  $X_{-1}$ , and also for the norms. Moreover, B has all properties required in Theorem 4.7. To see this let  $f \in C_b([0, t_0], (X_0, \tau))$  be arbitrary. Define a map  $\psi : \mathbb{R} \to \mathbb{R}$ by

$$\psi(\cdot) = \int_0^{t_0} T_{-1}(t_0 - r) Bf(r)(\cdot) \, \mathrm{d}r.$$

Observe that

$$T_{-1}(t_0 - r)Bf(r)(x) = T_{-1}(t_0 - r)\Phi(f(r))g(x) = \Phi(f(r))g(x + t_0 - r)$$

We claim that  $\psi$  is continuous. Indeed, let  $\varepsilon > 0$  be arbitrary, and notice that by substitution for each  $x \in \mathbb{R}$ 

$$\int_0^{t_0} \Phi(f(r))g(x+t_0-r) \, \mathrm{d}r = \int_x^{x+t_0} \Phi(f(x+t_0-s))g(s) \, \mathrm{d}s.$$

After this substitution we can make the following calculation for each  $x, y \in \mathbb{R}$ 

$$\begin{split} \psi(x) - \psi(y) &= \int_{x}^{x+t_0} \Phi(f(x+t_0-s))g(s) \, \mathrm{d}s - \int_{y}^{y+t_0} \Phi(f(x+t_0-s))g(s) \, \mathrm{d}s \\ &= \int_{0}^{x+t_0} \Phi(f(x+t_0-s))g(s) \, \mathrm{d}s - \int_{0}^{x} \Phi(f(x+t_0-s))g(s) \, \mathrm{d}s \\ &- \int_{0}^{y+t_0} \Phi(f(y+t_0-s))g(s) \, \mathrm{d}s + \int_{0}^{y} \Phi(f(y+t_0-s))g(s) \, \mathrm{d}s \\ &= \int_{y+t_0}^{x+t_0} \left( \Phi(f(x+t_0-s) - f(y+t_0-s)) \right)g(s) \, \mathrm{d}s \\ &+ \int_{x}^{y} \left( \Phi(f(x+t_0-s) - f(y+t_0-s)) \right)g(s) \, \mathrm{d}s. \end{split}$$

By the assumptions there exists M > 0 such that

$$\|(\Phi(f(x+t_0-\cdot)-f(y+t_0-\cdot)))g(\cdot)\|_{\infty} \le M.$$

For  $\delta := \frac{\varepsilon}{2M} > 0$  and for  $x, y \in \mathbb{R}$  with  $|x - y| < \delta$  we have

$$\begin{split} & |\psi(x) - \psi(y)| \\ \leq \int_{y+t_0}^{x+t_0} |(\Phi(f(x+t_0-s) - f(y+t_0-s))) g(s)| \, \mathrm{d}s \\ & + \int_x^y |(\Phi(f(x+t_0-s) - f(y+t_0-s))) g(s)| \, \mathrm{d}s \\ \leq & 2 |x-y| \cdot \|(\Phi(f(x+t_0-\cdot) - f(y+t_0-\cdot))) g(\cdot)\|_{\infty} \end{split}$$

$$\leq 2M \cdot |x - y| < \varepsilon.$$

This proves that  $\psi \in C_b(\mathbb{R})$ .

Observe that in general we only have

$$Q := \int_0^{t_0} T_{-1}(t_0 - r)Bf(r) \, \mathrm{d}r \in X_{-1},$$

so that point evaluation of this expression at  $x \in \mathbb{R}$  does not make sense. We know, however, that  $\psi \in C_b(\mathbb{R})$ , and that the pointwise Riemann-sums  $R_n(x)$  for the integral

$$\int_0^{t_0} \Phi(f(r))g(x+t_0-r)\mathrm{d}r$$

converges for all  $x \in \mathbb{R}$  to  $\psi(x)$ . If we can show that the sequence  $(R_n)_{n \in \mathbb{N}}$  converges in the sense of distributions we can conclude that  $Q := \int_0^{t_0} T_{-1}(t_0 - r)Bf(r) dr = \psi \in X_0$ . Let  $\tilde{\psi} \in \mathscr{D}(\mathbb{R})$  be a test function and define  $\varphi := \tilde{\psi} - \mathscr{D}\tilde{\psi}$ . Then

$$\left\langle (1-A_{-1})^{-1}R_n,\varphi\right\rangle \to \left\langle (1-A_{-1})^{-1}Q,\varphi\right\rangle.$$

By the meaning of this pairing we conclude that  $\langle R_n, \tilde{\psi} \rangle \to \langle Q, \tilde{\psi} \rangle$ . By the above we conclude that  $Q \in X_0$ .

The next step is to estimate the norm. Notice that

$$\begin{split} \left\| \int_{0}^{t_{0}} T_{-1}(t_{0}-r)Bf(r) \, \mathrm{d}r \right\|_{\infty} &= \sup_{x \in \mathbb{R}} \left| \int_{0}^{t_{0}} \Phi(f(r))g(x+t_{0}-r) \, \mathrm{d}r \right| \\ &\leq \sup_{x \in \mathbb{R}} \int_{0}^{t_{0}} |\Phi(f(r))| \cdot |g(x+t_{0}-r)| \, \mathrm{d}r \leq 2 \int_{0}^{t_{0}} |\Phi(f(r))| \, \mathrm{d}r \\ &\leq 2 \int_{0}^{t_{0}} \int_{\mathbb{R}} |f(r)(x)| \, \mathrm{d} |\mu|(x) \, \mathrm{d}r \leq 2 |\mu|(\mathbb{R}) \int_{0}^{t_{0}} \|f(r)\|_{\infty} \, \mathrm{d}r \\ &= 2 |\mu|(\mathbb{R}) \int_{0}^{t_{0}} \|f(r)\|_{\infty} \, \mathrm{d}r \leq 2 |\mu|(\mathbb{R}) t_{0} \, \|f\|_{\infty} \, . \end{split}$$

In particular we can choose  $t_0$  so small that  $M := 2 |\mu| (\mathbb{R}) t_0 < 1$ . Hence condition (c) of Theorem 4.7 is fulfilled. Condition (b) from Theorem 4.7 can be proven similarly. Let  $K \subseteq \mathbb{R}$  be an arbitrary compact set and  $\varepsilon > 0$ . Then

$$p_{K}\left(\int_{0}^{t_{0}} T_{-1}(t_{0}-r)Bf(r)(x) \, \mathrm{d}r\right) \leq \sup_{x \in K} \int_{0}^{t_{0}} |\Phi(f(r))| \cdot |g(x+t_{0}-r)| \, \mathrm{d}r$$
$$\leq 2 \sup_{x \in K} \int_{0}^{t_{0}} |\Phi(f(r))| \, \mathrm{d}r$$
$$\leq 2t_{0} |\mu|(\mathbb{R}) \sup_{r \in [0,t_{0}]} \sup_{y \in K'} |f(r)(y)| + \varepsilon \, ||f||_{\infty} \,,$$

since by the regularity of the measure  $\mu$  we choose  $K' \subseteq \mathbb{R}$  such that  $|\mu|(\mathbb{R} \setminus K') < \varepsilon$ . By Theorem 4.7 we conclude that  $(A_{-1} + B)|_{X_0}$  generates again a bi-continuous semigroup on  $C_b(\mathbb{R})$  with respect to  $\tau_{co}$ . We now give an expression for the generator. Observe that  $f \in D((A_{-1} + B)|_{X_0})$  if and only if  $f \in C_b(\mathbb{R})$  and  $f' + \Phi(f)g \in C_b(\mathbb{R})$  and this is precisely then, when the following conditions are satisfied.

$$\begin{cases} \lim_{t \nearrow -1} \left( f'(t) + \Phi(f)g(t) \right) = \lim_{t \searrow -1} \left( f'(t) + \Phi(f)g(t) \right), \\ \lim_{t \nearrow 0} \left( f'(t) + \Phi(f)g(t) \right) = \lim_{t \searrow 0} \left( f'(t) + \Phi(f)g(t) \right), \\ \lim_{t \nearrow 1} \left( f'(t) + \Phi(f)g(t) \right) = \lim_{t \searrow 1} \left( f'(t) + \Phi(f)g(t) \right). \end{cases}$$
(4.3.2)

By the explicit expression for  $g: \mathbb{R} \to \mathbb{R}$  we can rewrite Equation (4.3.2) as follows:

$$\begin{cases} \lim_{t \nearrow -1} f'(t) = \lim_{t \searrow -1} f'(t) - \Phi(f), \\ \lim_{t \nearrow 0} f'(t) = \lim_{t \searrow 0} f'(t) + 2\Phi(f), \\ \lim_{t \nearrow 1} f'(t) + \Phi(f) = \lim_{t \searrow 1} f'(t). \end{cases}$$
(4.3.3)

Or equivalently

$$\lim_{t \nearrow -1} f'(t) - \lim_{t \searrow -1} f'(t) = -\frac{1}{2} \left( \lim_{t \nearrow 0} f'(t) - \lim_{t \searrow 0} f'(t) \right) = \lim_{t \nearrow 1} f'(t) - \lim_{t \searrow 1} f'(t) = -\Phi(f).$$
(4.3.4)

We see that the generator (C, D(C)) of the perturbed semigroup is given by

$$Cf = f' + \int_{\mathbb{R}} f \, \mathrm{d}\mu \cdot g, \quad f \in D(C),$$
  
$$D(C) = \left\{ f \in \mathcal{C}_{\mathrm{b}}(\mathbb{R}) : f \in \mathcal{C}_{\mathrm{b}}^{1}(\mathbb{R} \setminus \{-1, 0, 1\}) \text{ and } (4.3.4) \text{ holds} \right\}.$$

The previous example uses a function  $g \in X_{-1}$  which has three points of discontinuity, with one sided limits at each of these points. We generalize this to a countable (discrete) set of jump discontinuities. For that assume that  $g \in X_{-1}$  is a function such that  $||g||_{\infty} < \infty$  and that the set of discontinuities of g is discrete. One defines again an operator  $B: X_0 \to X_{-1}$ by

$$Bf := \Phi(f)g := \int_{\mathbb{R}} f \, \mathrm{d}\mu \cdot g, \quad f \in \mathcal{C}_{\mathbf{b}}(\mathbb{R}).$$

Notice that none of previous calculations and arguments depend on the number of discontinuities (in fact, we only used that g is bounded). So we can conclude that  $(A_{-1} + B)_{|X_0}$ generates a bi-continuous semigroup on  $X_0$  with respect to  $\tau_{co}$ . The only issue we have to care about are the conditions mentioned in (4.3.4), that is an "explicit" description of the domain. Let  $Z := \{x_1, x_2, x_3, \ldots\}$  be the set of discontinuities of g that is assumed to be discrete, and we suppose that all of these points are jump discontinuities. Let us define  $a_n := \lim_{t \searrow x_n} g(t)$ and  $b_n := \lim_{t \searrow x_n} g(t)$ . We observe that  $f \in D((A_{-1} + B)_{|X_0})$  if and only if

$$\lim_{t \nearrow x_n} f'(t) + \Phi(f)a_n = \lim_{t \searrow x_n} f'(t) + \Phi(f)b_n, \quad \text{for each } n \in \mathbb{N},$$

or equivalently

$$\lim_{t \nearrow x_n} f'(t) - \lim_{t \searrow x_n} f'(t) = \Phi(f)(b_n - a_n), \quad \text{for each } n \in \mathbb{N}.$$

We conclude that the operator (C, D(C)) given by

$$Cf = f' + \int_{\mathbb{R}} f \, \mathrm{d}\mu \cdot g,$$
  
$$D(C) = \left\{ f \in \mathcal{C}_{\mathrm{b}}(\mathbb{R}) : f \in \mathcal{C}_{\mathrm{b}}^{1}(\mathbb{R} \setminus Z), \lim_{t \nearrow x_{n}} f'(t) - \lim_{t \searrow x_{n}} f'(t) = \Phi(f)(b_{n} - a_{n}), \quad n \in \mathbb{N} \right\}$$

generates a bi-continuous semigroup on  $C_b(\mathbb{R})$  with respect to  $\tau_{co}$ .

# § 4.4 Notes

This chapter is based on joint work with B. Farkas [27]. As mentioned in the notes of Chapter 2, i.e., Section 2.6, this joint work was the starting point for extrapolation spaces for bi-continuous semigroups. For the Desch–Schappacher perturbation theorem presented here in this chapter, cf. Theorem 4.7, we firstly started with perturbation operators B which are continuous from  $(X, \tau)$  to  $(\underline{X}_{-1}, \|\cdot\|_{-1})$ . During the last AGFA meeting in Blaubeuren in December 2016, the author gave a talk on this topic and during an evening of discussion he got an inspiring idea from R. Nagel how to stay with the extrapolation spaces in the category of bi-continuous semigroups. This idea actually culminate in Definition 2.9 and the results in [28].

As mentioned above in the introduction of this chapter, A. Bátkai, B. Jacob, J. Voigt and J. Wintermayr took positivity into account and proved a Desch–Schappacher perturbation theorem on AM-spaces in [20]. To do this for bi-continuous semigroups is one of the prospective research projects. A first step in this direction is a Miyadera–Voigt perturbation result for positive bi-continuous semigroups, which is in fact the subject of the following Chapter 5.

Compared with the original paper we omit the part on implemented semigroups and postpone them to Chapter 6. There we combine the results from the current chapter as well as from Chapter 2 to apply them to this special kind of semigroup.

# Chapter 5

# Positive Miyadera–Voigt perturbations

## Introduction

Various models of physical processes ask for positive solutions in order to have a reasonable interpretation, i.e., consider solutions containing the absolute temperature or a density. The maximum principle for partial differential equations guarantees positive solutions under positive initial data. This demonstrates the importance of positivity in the theory of operator semigroups. The fundamental concepts in this context such as vector lattices, Banach lattices and positive operators are studied in detail for example in [102] and [88]. Positive operator semigroups are treated for example by Arendt et al. [10] and more recently by A. Bátkai, M. Kramar Fijavž and A. Rhandi [21].

This chapter is based on an article by J. Voigt, cf. [119]. Here positive operator semigroups and perturbations are combined and the following Miyadera–Voigt perturbation result for positive  $C_0$ -semigroups was proven.

**Theorem.** [119, Thm. 0.1] Let E be a AL-space, and let A be the generator of a positive  $C_0$ -semigroup on E. Let  $B: D(A) \to E$  be a positive operator, and assume that A + B is resolvent positive. Then A + B is the generator of a positive  $C_0$ -semigroup.

Quite a number of other positive perturbation results for strongly continuous semigroups and their applications are for example handeled by Arlotti and Banasiak [14]. In this chapter we consider positive perturbations of bi-continuous semigroups in the style of Voigt's work. Especially, we use the Miyadera–Voigt perturbation theorem for bi-continuous semigroups developed by Farkas [54].

This chapter is organized as follows: in Section 1 we recall the basic terminology of positivity as well as the bounded and Miyadera–Voigt perturbations for bi-continuous semigroups and introduce Theorem 5.19 as our main result. Section 2 consists of the proof of this result. In the last section we discuss rank-one perturbations and bi-continuous semigroup on the space  $M(\mathbb{R})$  of bounded Borel measures in connection with differentiable measures.

#### § 5.1 Preliminaries

In this section we recall the notion of positivity in Banach spaces (see Section 5.1.1) and the already known perturbation results for bi-continuous semigroups, due to B. Farkas in [55, Thm. 3.5] and [54, Thm. 3.2.2]. In particular we recall bounded perturbations in Section 5.1.2 and perturbations of Miyadera–Voigt type in Section 5.1.3. The formulation for  $C_0$ -semigroups on Banach spaces can be found in [52, Chapter III, Sect. 1 & Sect. 3c].

#### 5.1.1 Positivity and bi-AL-spaces

As a matter of fact, we need a order for our underlying Banach space in order to have a concept of positivity. The following definitions are taken from cf. [102, Chapter II] and [88, Chapter 1]. Firstly, we recall the notion of a partial order.

**Definition 5.1.** Let X be a nonempty set. A *partial order* on X is a binary relation  $\leq$  satisfying the following properties for all  $x, y, z \in X$ :

1.  $x \leq x$ ,

- 2.  $x \leq y$  and  $y \leq x$  implies x = y,
- 3.  $x \leq y$  and  $y \leq z$  implies  $x \leq z$ .

Since we also have a vector space structure we wish to have a order which is consistent with the vector space. This leads to the notion of vector lattices.

**Definition 5.2.** A vector lattice or Riesz space is a vector space V equipped with a partial order  $\leq$  such that for each  $x, y, z \in V$ :

- 1.  $x \le y \Rightarrow x + z \le y + z$ ,
- 2.  $x \leq y \Rightarrow \alpha x \leq \alpha y$  for all scalars  $\alpha \geq 0$ ,
- 3. For any pair  $x, y \in V$  there exists a supremum, denoted by  $x \lor y$ , and a infimum, denoted by  $x \land y$ , in V with respect to the partial order  $\leq$ .

An element  $x \in V$  is called *positive* if  $x \ge 0$ . The set of all positive elements is denoted by  $V_+$ .

**Remark 5.3.** We observe that for a Riesz space V the set  $V_+$  is a convex proper cone, i.e., the set of positive elements satisfies the following properties:

- (i)  $V_+ + V_+ \subseteq V_+$ ,
- (ii)  $\lambda V_+ \subseteq V_+$  for all  $\lambda > 0$ ,
- (iii)  $V_+ \cap (-V_+) = \{0\}$

Pay attention to the fact that each convex proper cone  $C \subseteq V$ , i.e., C satisfies the three previous conditions, gives rise to an order  $\leq$  defined by  $x \leq y$  if and only if  $y - x \in C$ . This order turns V into a vector lattice and the set of positive elements coincides with C, i.e., the equality  $V_+ = C$  holds.

In the case of normed spaces and especially Banach spaces, we recall the notion of Banach lattices, for the purpose of compatibility of the norm and the order.

**Definition 5.4.** A Banach lattice is a Banach space  $(X, \|\cdot\|)$  which is a Riesz space with an partial order such that for all  $x, y \in X$ :  $|x| \leq |y| \Rightarrow ||x|| \leq ||y||$ , where  $|x| := x \land (-x)$  is called the absolute value.

**Remark 5.5.** As a matter of fact, by [102, Chapter II, Prop. 5.2] the positive cone  $X_+ = \{x \in X : x \ge 0\}$  of a Banach lattice X is always a closed set. Aside from that, X is even archimedean, i.e., if  $x, y \in X$  satisfy  $nx \le y$  for all  $n \in \mathbb{N}$ , then  $x \le 0$ . We remark that in contrast, vector lattices in general do not necessarily have this property as the example of  $\mathbb{R}^n$  equipped with the lexicographical order shows, cf. [125, Example 9.2(ii)]

Recall from [88, pp. 8-9], that if K is a compact Hausdorff space, then the space  $(C(K), \|\cdot\|_{\infty})$ with the pointwise order, i.e.,  $f \leq g$  if and only if  $f(x) \leq g(x)$  for all  $x \in K$ , is an example of a Banach lattice. Likewise, for a measure space  $(\Omega, \Sigma, \mu)$  the Banach space of integrable functions  $(L^1(\Omega, \mu), \|\cdot\|_1)$  is a Banach lattice with the pointwise order. Moreover, for a Polish space  $\Omega$ , the space of bounded Borel measures  $M(\Omega)$  is also a Banach lattice. We say that  $\mu \in M(\Omega)$  is positive if and only if  $\mu(A) \geq 0$  for all  $A \in \Sigma$ ,  $A \neq \emptyset$ . Equipped with the total variance norm  $\|\cdot\|_{M(\Omega)}$ , defined by

$$\|\mu\|_{\mathcal{M}(\Omega)} := |\mu|\left(\Omega\right) = \sup_{\pi} \sum_{A \in \pi} |\mu(A)|,$$

where the supremum runs over all possible partitions  $\pi$  of  $\Omega$  into a countable number of disjoint measurable sets,  $M(\mathbb{R})$  becomes a Banach lattice. On Banach lattices one can also consider linear operators. Positivity for operators is then defined as follow.

**Definition 5.6.** A bounded linear operator  $T \in \mathscr{L}(X)$  on a Banach lattice X is called *positive*, denoted by  $T \ge 0$ , if  $Tx \ge 0$  for each  $x \in X_+$ . A semigroup of bounded linear operators  $(T(t))_{t\ge 0}$  on such a Banach lattice is called positive if  $T(t) \ge 0$  for each  $t \ge 0$ .

The following term was suggested by Arendt [8].

**Definition 5.7.** A linear operator (A, D(A)) on a Banach lattice X is called *resolvent positive* if there exists  $\omega \in \mathbb{R}$  such that  $(\omega, \infty) \subseteq \rho(A)$  and such that  $R(\lambda, A) \ge 0$  for each  $\lambda > \omega$ .

In [119] the concept of so-called AL-spaces, as it is mentioned in the introduction of this chapter, is significant. These spaces satisfy a special kind of norm property, cf. [102, Chapter II, Sect. 8].

**Definition 5.8.** A Banach lattice  $(X, \|\cdot\|)$  is called an AL-space, or abstract L-space, if the norm satisfies

$$||x+y|| = ||x|| + ||y||,$$

for each  $x, y \in X_+$ .

Typical examples for AL-spaces are the already mentioned spaces  $L^1(\Omega, \mu)$  and  $M(\Omega)$ . To see that they are actually AL-spaces let  $f, g \in L^1(\Omega, \mu)$  be positive, i.e.,  $f(x) \ge 0$  and  $g(x) \ge 0$  for each  $x \in \Omega$ .

$$\left\|f+g\right\|_1 = \int_{\Omega} \left|f+g\right| \, \mathrm{d}\mu$$

$$= \int_{\Omega} f + g \, d\mu$$
$$= \int_{\Omega} f \, d\mu + \int_{\Omega} g \, d\mu$$
$$= \int_{\Omega} |f| \, d\mu + \int_{\Omega} |g| \, d\mu$$
$$= \|f\|_1 + \|g\|_1$$

In order to show that  $M(\Omega)$  is a AL-space let  $\mu, \nu \in M(\Omega)$  be two positive bounded Borel measures and observe:

$$\begin{split} \|\mu + \nu\|_{\mathcal{M}(\Omega)} &= |\mu + \nu| (\Omega) \\ &= \sup_{\pi} \sum_{A \in \pi} |(\mu + \nu)(A)| \\ &= \sup_{\pi} \sum_{A \in \pi} |\mu(A) + \nu(A)| \\ &= \sup_{\pi} \sum_{A \in \pi} \mu(A) + \nu(A) \\ &= \sup_{\pi} \sum_{A \in \pi} \mu(A) + \sum_{A \in \pi} \nu(A) \\ &= \sup_{\pi} \sum_{A \in \pi} \mu(A) + \sup_{\pi} \sum_{A \in \pi} \nu(A) \\ &= \sup_{\pi} \sum_{A \in \pi} |\mu(A)| + \sup_{\pi} \sum_{A \in \pi} |\nu(A)| \\ &= |\mu| (\Omega) + |\nu| (\Omega) \\ &= \|\mu\|_{\mathcal{M}(\Omega)} + \|\nu\|_{\mathcal{M}(\Omega)} , \end{split}$$

where the supremum runs over all possible partitions  $\pi$  of  $\Omega$  into a countable number of disjoint measurable subsets.

As we have to take care of an additional locally convex topology in the setting of bi-continuous semigroups, we introduce the following notion which is related to Definition 5.8.

**Definition 5.9.** Let  $(X, \|\cdot\|, \leq)$  be a Banach lattice with ordering and locally convex topology  $\tau$  generated by a family  $\mathcal{P}$  of seminorms which satisfies the Assumptions 1.1. We say that X is a *bi*-AL space if for all  $x, y \in X_+$  the equality

$$||x + y|| = ||x|| + ||y||$$

holds and there exists  $\mathcal{P}_+ \subseteq \mathcal{P}$  such that  $\mathcal{P}_+$  still generates the locally convex topology  $\tau$  and for all  $x, y \in X_+$ 

$$p(x+y) = p(x) + p(y)$$

for each  $p \in \mathcal{P}_+$ .

Locally convex spaces satisfying such additivity conditions as in Definition 5.9 are also mentioned in [35] to discuss regular operators on vector lattices. An example of a space satisfying both the norm and seminorm property from Definition 5.9 is  $M(\Omega)$ . Recall from Section 1.6 that  $M(\Omega)$  equipped with the total variation norm and the weak\*-topology  $\tau^{\circ} = \sigma(C_b(\Omega), M(\Omega))$  satisfies Assumption 1.1. In fact,  $\tau^{\circ}$  is the locally convex topology generated by the following set of seminorms

$$\mathcal{P} := \{ p_f : f \in \mathcal{C}_{\mathbf{b}}(\Omega) \},\$$

where

$$p_f(\mu) := \left| \int_{\Omega} f \, \mathrm{d}\mu \right|, \quad \mu \in \mathcal{M}(\Omega),$$

Now let

$$\mathcal{P}_+ := \{ p_f : f \in \mathcal{C}_{\mathbf{b}}(\Omega), f \ge 0 \}.$$

Let  $\tau_+$  denote the locally convex topology generated by  $\mathcal{P}_+$ . We first observe that for each  $p_f \in \mathcal{P}_+$  one has for  $\mu, \nu \ge 0$ 

$$p_f(\mu) + p_f(\nu) = \left| \int_{\Omega} f \, d\mu \right| + \left| \int_{\Omega} f \, d\nu \right|$$
$$= \int_{\Omega} f \, d\mu + \int_{\Omega} f \, d\nu$$
$$= \int_{\Omega} f \, d(\mu + \nu)$$
$$= \left| \int_{\Omega} f \, d(\mu + \nu) \right|$$
$$= p_f(\mu + \nu)$$

It remains to show that  $\tau^{\circ} = \tau_+$ .

**Lemma 5.10.** The family  $\mathcal{P}_+$  generates the topology  $\tau^{\circ}$ , i.e., for a net  $(\mu_{\alpha})_{\alpha \in \Lambda}$  in  $M(\Omega)$ :

$$\mu_{\alpha} \xrightarrow{\tau^{\circ}} \mu \iff \mu_{\alpha} \xrightarrow{\tau_{+}} \mu$$

*Proof.* Suppose that

$$\int_{\Omega} f \, \mathrm{d}\mu_{\alpha} \to \int_{\Omega} f \, \mathrm{d}\mu, \tag{5.1.1}$$

for each  $f \in C_{\rm b}(\Omega)$ , then obviously (5.1.1) also holds for every positive  $f \in C_{\rm b}(\Omega)$ . For the other implication notice that for arbitrary  $f \in C_{\rm b}(\Omega)$  one can decompose f into a positive and negative part, i.e., there exist positive  $f_+, f_- \in C_{\rm b}(\Omega)$  such that

$$f = f_+ - f_-$$

Then

$$\int_{\Omega} f_+ \, \mathrm{d}\mu_{\alpha} \to \int_{\Omega} f_+ \, \mathrm{d}\mu,$$

and

$$\int_{\Omega} f_{-} \, \mathrm{d}\mu_{\alpha} \to \int_{\Omega} f_{-} \, \mathrm{d}\mu.$$

Since

$$\int_{\Omega} f \, \mathrm{d}\mu_{\alpha} = \int_{\Omega} f_{+} \, \mathrm{d}\mu_{\alpha} - \int_{\Omega} f_{-} \, \mathrm{d}\mu_{\alpha},$$

we observe that

$$\int_{\Omega} f \, \mathrm{d}\mu_{\alpha} \to \int_{\Omega} f_+ \, \mathrm{d}\mu - \int_{\Omega} f_- \, \mathrm{d}\mu,$$

and hence we are done.

#### 5.1.2 Bounded perturbations

Let us continue with perturbation results for bi-continuous semigroups. The bounded perturbations are in some sense simple, since one has not to care about the domain of the perturbing operator. Moreover, the following theorem shows that in particular bounded operators with a additional continuity condition give rise to a perturbation.

**Theorem 5.11.** [54, Thm. 3.1.5] Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup on a Banach space  $X_0$  with respect to  $\tau$  and with generator (A, D(A)). Suppose that  $B \in \mathscr{L}(E)$  is  $\tau$ sequentially continuous on  $\|\cdot\|$ -bounded sets. Then (A + B, D(A)) is also a generator of a bi-continuous semigroup  $(S(t))_{t\geq 0}$ . Moreover  $(S(t))_{t\geq 0}$  is given by the Dyson–Phillips series

$$S(t) := \sum_{n=0}^{\infty} T_n(t), \quad t \ge 0,$$

which is uniformly norm-convergent on compact intervals. Here the Dyson-Phillips terms  $(T_n(t))_{t\geq 0}$  are defined as

$$T_0(t) := T(t), \quad T_n(t) := \int_0^t T(t-s)BT_{n-1}(s) \, \mathrm{d}s, \quad n > 0,$$

where the integral is under stood in the  $\tau$ -strong topology.

#### 5.1.3 Miyadera–Voigt perturbations

In contrast to Section 5.1.2 we now come to perturbations of bi-continuous semigroups with an unbounded operator (B, D(B)). Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup on a Banach space  $X_0$  with respect to  $\tau$  and generator (A, D(A)). On the domain D(A) of A one can define a locally convex topology  $\tau_A$  which is determined by the family of seminorms

$$\mathcal{P}_A := \{ p(\cdot) + q(A \cdot) : p, q \in \mathcal{P} \}.$$

The graph norm  $\| \cdot \|_A$  on D(A) is defined by

$$||x||_A := ||x|| + ||Ax||, \quad x \in D(A).$$

Due to [54, Sect. 1.2 a)] we always may assume that  $p(x) \leq ||x||$  for each  $p \in \mathcal{P}$  and  $x \in X_0$  and hence one observes that  $\tau_A$  is coarser than the graph norm topology (the topology coming from the norm  $||\cdot||_A$ ). For the Miyadera–Voigt perturbations we consider operators B:  $(D(A), \tau_A) \to (X_0, \tau)$  which are continuous on  $||\cdot||_A$ -bounded sets. The following definitions are needed to prepare the perturbation theorem.

**Definition 5.12.** Let  $D \subseteq X_0$  be an arbitrary subset and  $\eta > 1$ . We say that D is  $\eta$ -bi-dense in  $X_0$  for the locally convex topology  $\tau$ , if for each  $x \in X_0$  there exists a sequence  $(x_n)_{n \in \mathbb{N}}$  in

D which converges to x with respect to  $\tau$  and such that

$$\|x_n\| \le \eta \|x\|,$$

for each  $n \in \mathbb{N}$ .

**Definition 5.13.** A bounded function  $F : [0, t_0] \to \mathscr{L}(X_0)$  is said to be *local*, if for each  $p \in \mathcal{P}$  and  $\varepsilon > 0$  there exists a constant K > 0 and  $q \in \mathcal{P}$  such that

$$p(F(t)x) \le Kq(x) + \varepsilon ||x||,$$

for all  $t \in [0, t_0]$  and  $x \in X_0$ .

- **Remark 5.14.** (i) As examples of local functions one can consider the family  $\{R(\lambda, A) : \lambda \in [\alpha, \beta]\}$  where (A, D(A)) is the generator of a bi-continuous semigroup and  $\omega_0(T) < \alpha < \beta$ .
  - (ii) For every bi-continuous semigroup  $(T(t))_{t\geq 0}$  on  $C_b(\Omega)$ , for  $\Omega$  a Polish space, the set  $\{T(t): t \in [0, t_0]\}$  is local for each  $t_0 > 0$ .
- (iii) If  $(T(t))_{t\geq 0}$  is a bi-continuous semigroup on  $C_b(\Omega)$  and  $\mathscr{K} \subseteq M(\Omega)$  and norm-bounded and weak\*-compact subset the set  $\{T'(t)\nu : \nu \in \mathscr{K}\}$  is tight/local. As example one can take  $\mathscr{K} = M_1(\Omega)$  if  $\Omega$  is compact.

One may not expect that every operator (B, D(B)) has the property that (A+B, D(A+B))generates again a bi-continuous semigroups. For that reason one introduces the notion of admissibility. Recall from Chapter 4 the definition of the space  $\mathfrak{X}_{t_0}$ , see (4.1.1), and define the Miyadera–Voigt admissibility as follows.

**Definition 5.15.** Let (A, D(A)) be the  $\eta$ -bi-densely defined generator of a bi-continuous semigroup  $(T(t))_{t\geq 0}$ . An operator  $B : (D(A), \tau_A) \to (X_0, \tau)$  which is continuous on  $\|\cdot\|_A$ bounded sets is called *Miyadera-Voigt admissible* on  $\mathfrak{X}_{t_0}$  if there exists  $t_0 > 0$  for which  $T(t) \in \mathfrak{X}_{t_0}$  for  $t \in [0, t_0]$  and such that the following conditions hold

1. For all  $x \in D(A)$  the maps

$$s \mapsto \|BT(s)x\|, \quad s \in [0, t_0]$$

are bounded.

2. The operator

$$B(F,t)x := \int_0^t F(t-s)BT(s)x \, \mathrm{d}s$$

defined on D(A) extends to linear operator  $\overline{B}(F,t)$  on  $X_0$  which is  $\tau$ -continuous on normbounded sets for all  $t \in [0, t_0]$  and  $F \in \mathfrak{X}_{t_0}$ . We require moreover that the operator  $\overline{B}(F,t)$ is also norm-bounded.

3. The abstract Volterra operator  $V_{t_0}$  on  $\mathfrak{X}_{t_0}$ , cf. (4.1.1), defined by

$$(V_{t_0}F)(t)x := \overline{B}(F,t)x,$$

for all  $x \in X_0$  and  $t \in [0, t_0]$  is a bounded operator on  $\mathfrak{X}_{t_0}$  and we have  $||V_{t_0}|| \leq \frac{1}{n}$ 

The following theorem tells us that under the condition of Definition 5.15 the operator (A + B, D(A)) generates a bi-continuous semigroup on  $X_0$ . This generalize the perturbation theorem originally due to I. Miyadera [89] and improved by J. Voigt [118].

**Theorem 5.16.** [54, Thm. 3.2.2] Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup with generator (A, D(A)) and suppose that  $B : (D(A), \tau_A) \to (X_0, \tau)$  is Miyadera–Voigt admissible on  $\mathfrak{X}_{t_0}$ . In this case (A + B, D(A)) generates a bi-continuous semigroup  $(S(t))_{t\geq 0}$ . Moreover the semigroup  $(S(t))_{t\geq 0}$  satisfies

$$S(t)x = T(t)x + \int_0^t S(t-s)BT(s)x \, \mathrm{d}s,$$

for all  $x \in D(A)$  and  $t \ge 0$ .

The following theorem gives at least for local bi-continuous semigroups a sufficient condition for  $B: (D(A), \tau_A) \to (X_0, \tau)$  being Miyadera–Voigt admissible.

**Theorem 5.17.** [54, Thm. 3.2.3] Let  $(T(t))_{t\geq 0}$  be a bi-continuous semigroup on  $X_0$  with respect to  $\tau$  and generator (A, D(A)). Suppose that D(A) is  $\eta$ -bi-dense in  $X_0$  and that B:  $(D(A), \tau_A) \to (X_0, \tau)$  is continuous on  $\|\cdot\|_A$ -bounded sets. Suppose that there exists  $t_0 > 0$ and  $0 < K < \frac{1}{\eta}$  such that

- (a) The map  $s \mapsto ||BT(s)x||$  is bounded on  $[0, t_0]$  and for each  $x \in D(A)$ .
- (b)  $\int_0^t \|BT(s)x\| \, \mathrm{d}s < K \|x\|$  for each  $t \in [0, t_0]$  and  $x \in D(A)$ .
- (c) For all  $\varepsilon > 0$  and  $p \in \mathcal{P}$  there exists  $q \in \mathcal{P}$  and M > 0 such that

$$\int_0^{t_0} p(BT(s)x) \, \mathrm{d}s < Mq(x) + \varepsilon \left\| x \right\|,$$

for each  $x \in D(A)$ .

Then (A + B, D(A + B)) generates a bi-continuous semigroup  $(S(t))_{t\geq 0}$ . Furthermore, the semigroup  $(S(t))_{t>0}$  satisfies the variation of parameter formula

$$T(t)x = S(t)x + \int_0^t T(t-s)BS(s)x \, \mathrm{d}s, \quad x \in D(A).$$

In [55] B. Farkas considered bounded and Miyadera–Voigt of bi-continuous semigroups. In this chapter we take positivity of Miyadera–Voigt perturbations into account as J. Voigt in [119] did for strongly continuous operator semigroups on Banach spaces. In [13] W. Arendt and A. Rhandi characterize positive perturbations by multiplication operator and apply this to elliptic Schrödinger operators.

**Remark 5.18.** (i) If  $(T(t))_{t\geq 0}$  is a bi-continuous semigroup generated by (A, D(A)), then one can deduce from [14, Lemma 4.15] the following equivalence for  $\lambda \in \mathbb{R}$ :

$$\exists M \in (0,1) \ \forall t \in [0,t_0] \ \forall x \in D(A) : \int_0^t \|BT(s)x\| \ \mathrm{d}s \le M \|x\|$$
$$\iff \exists M' \in (0,1) \ \forall t \in [0,t_0] \ \forall x \in D(A) : \int_0^t \left\| \mathrm{e}^{-\lambda s} BT(s)x \right\| \ \mathrm{d}s \le M' \|x\|$$

(ii) One also easily proves the following equivalence:

$$\forall \varepsilon > 0 \ \forall p \in \mathcal{P} \ \exists K \ge 0 \ \exists q \in \mathcal{P} \ \forall x \in D(A) : \ \int_0^{t_0} p(BT(s)x) \ \mathrm{d}s \le Kq(x) + \varepsilon \|x\|$$

$$\iff \forall \varepsilon > 0 \ \forall p \in \mathcal{P} \ \exists K' \ge 0 \ \exists q' \in \mathcal{P} \ \forall x \in D(A) : \ \int_0^{t_0} p(\mathrm{e}^{-\lambda s}BT(s)x) \ \mathrm{d}s \le K'q'(x) + \varepsilon \|x\|$$

The main result of this chapter is the following.

**Theorem 5.19.** Let  $\eta > 1$ , and let (A, D(A)) be the generator of a positive local bi-continuous semigroup  $(T(t))_{t\geq 0}$  on a bi-AL space X with  $\eta$ -bi-dense domain D(A). Let  $B : D(A) \to X$  be a positive operator, i.e.,  $Bx \geq 0$  for all  $x \in D(A) \cap X_+$ , and assume that  $BR(\lambda, A)$  is local and (A + B, D(A)) is resolvent positive. Then (A + B, D(A)) is the generator of a positive bi-continuous semigroup.

The following section is devoted to the proof of Theorem 5.19.

## § 5.2 The Proof

Recall from [52, Chapter II, Def. 1.12] that the spectral bound s(A) of an linear operator is defined by

$$s(A) := \sup \{ \operatorname{Re}(\lambda) : \lambda \in \sigma(A) \}.$$

In order to prove Theorem 5.19 we need the following lemma.

**Lemma 5.20.** Let  $\eta > 1$  and let (A, D(A)) be the  $\eta$ -bi-densely defined generator of a positive local bi-continuous semigroup  $(T(t))_{t\geq 0}$  on a bi-AL space X. Let  $M := \sup_{t\in[0,1]} ||T(t)|| < \infty$  and let  $B : D(A) \to X$  be a positive operator such that there exists  $\lambda > s(A)$  such that the operator  $BR(\lambda, A)$  is local and  $||BR(\lambda, A)|| < \frac{1}{2M} < 1$ . Then A + B is the generator of a positive bi-continuous semigroup.

*Proof.* We start by establishing property (ii) in Theorem 5.17. By Remark 5.18(i) it suffices that we have the following estimate for each  $x \in D(A)_+$ 

$$\begin{split} \int_0^t \left\| B \mathrm{e}^{-\lambda s} T(s) x \right\| \, \mathrm{d}s &= \left\| \int_0^t B \mathrm{e}^{-\lambda s} T(s) x \, \mathrm{d}s \right\| \\ &= \left\| B \int_0^t \mathrm{e}^{-\lambda s} T(s) x \, \mathrm{d}s \right\| \\ &\leq \left\| B R(\lambda, A) x \right\| \\ &< \frac{1}{2M} \left\| x \right\|, \end{split}$$

where the first equality is justified by the AL-property of the space and the fact that if  $x \in D(A)_+$  then it is element of the space of strong continuity (which in fact coincides with  $\underline{X}_0 := \overline{D(A)}^{\|\cdot\|}$ ). The second equality follows by the fact that the operator B is a continuous map from  $(D(A), \tau_A)$  to  $(X, \tau)$  which is A-bounded by [14, Lemma 4.1], i.e., there exists  $a, b \geq 0$  such that  $\|Bx\| \leq a \|Ax\| + b \|x\|$  for each  $x \in D(A)$ . The first inequality follows by the Laplace transform representation of the resolvent  $R(\lambda, A)$ . As a consequence we conclude

that property (ii) of Theorem 5.17 holds. Now let  $\varepsilon > 0$  and  $p \in \mathcal{P}_+$  be arbitrary. Then

$$\int_0^{t_0} p\left(e^{-\lambda s}BT(s)x\right) \, \mathrm{d}s = p\left(\int_0^{t_0} e^{-\lambda s}BT(s)x \, \mathrm{d}s\right) = p\left(B\int_0^{t_0} e^{-\lambda s}T(s)x \, \mathrm{d}s\right)$$
$$\leq p\left(BR(\lambda, A)x\right) \leq Kq(x) + \varepsilon \|x\|.$$

Here the first step follows by the properties of the bi-AL space. The second one follows by the same argument as before and the last inequality follows by the assumption of localness of the operator  $BR(\lambda, A)$ . Hence property (iii) is fullfilled. Moreover also (i) holds since (B, D(B)) is A-bounded by [14, Lemma 4.1].

Keep in mind that we showed that the properties (a)-(c) of Theorem 5.17 hold for  $x \in D(A)_+$ . We have to show that they hold true for each  $x \in D(A)$ . In the first place we start showing that the norm condition (b) of Theorem 5.17 holds from each  $x \in D(A)$ . For this purpose, let  $x \in D(A)$  and find  $x_+, x_- \in X_+$  such that  $x = x_+ - x_-$ . Especially, take  $x_+ := x \vee 0$  and  $x_- := -(x \wedge 0)$  and observe that  $x_+ + x_- = |x|$ . For  $n \in \mathbb{N}$  define

$$x_{n,\pm} := n \int_0^{\frac{1}{n}} T(t) x_{\pm} \, \mathrm{d}t \in D(A),$$

and notice that  $x_{n,\pm} \xrightarrow{\tau} x_{\pm}$  in X and  $x_{n,+} - x_{n,-} \xrightarrow{\tau_A} x$ . The sequences  $(x_{n,+})_{n \in \mathbb{N}}$  and  $(x_{n,-})_{n \in \mathbb{N}}$ , and hence also  $(x_{n,+} - x_{n,-})_{n \in \mathbb{N}}$ , are norm-bounded, i.e.,

$$\|x_{n,\pm}\| = \left\|n\int_0^{\frac{1}{n}} T(t)x_{\pm} \, \mathrm{d}t\right\| \le n\int_0^{\frac{1}{n}} \|T(t)x_{\pm}\| \, \mathrm{d}t \le L \, \|x_{\pm}\|,$$

where  $L := \sup_{t \in [0, \frac{1}{n}]} ||T(t)|| < \infty$ . Hence we obtain

$$\int_0^t \left\| e^{-\lambda s} BT(s)(x_{n,+} - x_{n,-}) \right\| \, \mathrm{d}s < \frac{1}{2M} \left( \|x_{n,+}\| + \|x_{n,-}\| \right) = \frac{1}{2M} \|x\|.$$
(5.2.1)

We have to show that the integrand converges to  $\|e^{-\lambda s}BT(s)x\|$  in order to conclude the desired estimate. Observe that for each  $\varphi \in (X, \tau)'$ 

$$\varphi\left(\int_{0}^{t} e^{-\lambda s} BT(s) x \, ds\right) = \int_{0}^{t} \varphi\left(e^{-\lambda s} BT(s) x\right) \, ds$$
$$= \int_{0}^{t} \tau \lim_{n \to \infty} \varphi\left(e^{-\lambda s} BT(s) (x_{n,+} - x_{n,-})\right) \, ds$$
$$\leq \tau \liminf_{n \to \infty} \int_{0}^{t} \left\|e^{-\lambda s} BT(s) (x_{n,+} - x_{n,-})\right\| \, ds = \frac{1}{2M} \|x\|$$

Due to the norming property of the local convex topology (cf. Assumption 1.1) and the ALproperty we obtain by taking the supremum over all  $\varphi \in (X, \tau)'$  with  $\|\varphi\| \leq 1$  the following inequality

$$\int_0^t \left\| e^{-\lambda s} BT(s) x \right\| \, \mathrm{d}s = \frac{1}{2M} \left\| x \right\|.$$

To show the third requirement of Theorem 5.17 is valid, let  $\varepsilon > 0$  and  $p \in \mathcal{P}_+$  be arbitrary and observe that there exists K > 0 and  $q \in \mathcal{P}$  such that for each  $n \in \mathbb{N}$ 

$$\int_{0}^{t_{0}} p(e^{-\lambda s}BT(s)(x_{n,+} - x_{n,-})) \, \mathrm{d}s \leq \int_{0}^{t_{0}} p(e^{-\lambda s}BT(s)x_{n,+}) \, \mathrm{d}s + \int_{0}^{t_{0}} p(e^{-\lambda s}BT(s)x_{n,-}) \, \mathrm{d}s$$
$$\leq K \left(q(x_{n,+}) + q(x_{n,-})\right) + \varepsilon \left(\|x_{n,+}\| + \|x_{n,i}\|\right)$$
$$\leq K \left(q(x_{n,+}) + q(x_{n,-})\right) + 2L\varepsilon \|x\|$$

Since by construction  $x_{n,+} \to x_+$  and  $x_{n,-} \to x_-$  we conclude that  $q(x_{n,+}) + q(x_{n,-}) \to q(|x|) = q(x)$ . The left-hand side also converges. To see this define for a fixed  $\lambda > s(A)$  a sequence  $(y_n)_{n \in \mathbb{N}}$  in X for  $n \in \mathbb{N}$  by

$$y_n := (\lambda - A)(x_{n,+} - x_{n,-}),$$

and set

$$y := \tau \lim_{n \to \infty} y_n = (\lambda - A)x.$$

Then by localness on the operator  $BR(\lambda, A)$  and the semigroup  $(T(t))_{t\geq 0}$  we find K > 0 and  $q, q' \in \mathcal{P}$  such that

$$\begin{aligned} \left| p(\mathrm{e}^{-\lambda s} BR(\lambda, A)T(s)y_n) - p(\mathrm{e}^{-\lambda s} BR(\lambda, A)T(s)y) \right| \\ \leq p(\mathrm{e}^{-\lambda s} BR(\lambda, A)T(s)y_n - \mathrm{e}^{-\lambda s} BR(\lambda, A)T(s)y) \\ \leq \mathrm{e}^{-\lambda s} p(BR(\lambda, A)T(s)(y_n - y)) \\ \leq \mathrm{e}^{-\lambda s} \left( Kq(T(s)(y_n - y)) + \varepsilon \|T(s)(y_n - y)\| \right) \\ \leq K'q(T(s)(y_n - y)) + \mathrm{e}^{(\omega - \lambda)s} \varepsilon \|y_n - y\| \\ \leq K''q'(y_n - y) + \varepsilon (K' + \mathrm{e}^{(\omega - \lambda)s}) \|y_n - y\| \\ \leq K''q'(y_n - y) + \varepsilon M(K' + \mathrm{e}^{(\omega - \lambda)s}), \end{aligned}$$

where  $K' := Ke^{-\lambda s}$  and K'' > 0 is a product of K' and a constant coming from the localness of the semigroup  $(T(t))_{t\geq 0}$ . Moreover,  $M \geq 0$  is a constant arising from the exponential boundedness of the semigroup. Since  $y_n \xrightarrow{\tau} y$  and  $\varepsilon > 0$  was arbitrary we see that the convergence of the integrand is uniform in s. Thus the integral converges and we are done.  $\Box$ 

The main work for proving Theorem 5.19 is included in the previous Lemma 5.20. Now we are able to accomplish the proof of our main theorem which runs in fact parallel to the original proof of [119, Thm. 0.1].

**Proof of Theorem 5.19.** By [119, Thm. 1.1] there exists a  $\lambda \in \rho(A+B)$  and we know that  $r(BR(\lambda, A)) < 1$ . Moreover

$$R(\lambda, A) \le R(\lambda, A) \sum_{n=0}^{\infty} (BR(\lambda, A))^n = R(\lambda, A + B).$$

Now, by taking sB instead of B for  $s \in [0, 1]$  we obtain by the above

$$R(\lambda, A) \le R(\lambda, A + sB) \le R(\lambda, A + B).$$

Since B is positive and  $\operatorname{Ran}(R(\lambda, A+B)) = D(A)$ , we conclude that  $BR(\lambda, A+B) \in \mathscr{L}(X)$ .

Therefore, also  $2\eta BR(\lambda, A+B) \in \mathscr{L}(X)$  and there exists  $n \in \mathbb{N}$  such that

$$\|2\eta BR(\lambda, A+B)\| < n.$$

Hence

$$\left\|\frac{1}{n}BR(\lambda,A+sB)\right\| < \frac{1}{2\eta},$$

for each  $s \in [0, 1]$ . In particular, one has

$$\left\|\frac{1}{n}BR\left(\lambda,A+\frac{j}{n}B\right)\right\| < \frac{1}{2\eta},$$

for  $0 \le j \le n-1$ . Now we apply Lemma 5.20 for the perturbation  $\frac{1}{n}B$  repeatedly for  $A, A + \frac{1}{n}B, \ldots, A + \frac{n-1}{n}B$  and obtain the generation by A + B in the last step.

#### § 5.3 Examples

#### 5.3.1 Rank-one perturbations

As a first example we consider rank-one perturbations as they are treated for  $C_0$ -semigroups by W. Arendt in A. Rhandi [13, Thm. 2.2]. Let  $(T(t))_{t\geq 0}$  be a positive bi-continuous semigroup on X with respect to  $\tau$  with generator (A, D(A)). As a matter of fact every locally convex topology is generated by a family of seminorms  $\mathcal{P}$ . Since the generated topology does not change if we add another continuous seminorm, we can assume that  $\mathcal{P}$  also contains all finite linear combinations of their seminorms, see also [33, Prop. 7.1.4]. Keep in mind, that if we actually suppose this, the inequality  $p(x) \leq ||x||$  for  $p \in \mathcal{P}$  and  $x \in X$  does not hold in general. For  $\varphi : D(A) \to \mathbb{R}$  a positive  $\tau_A$ -continuous linear functional and  $y \geq 0$  in X we define the rank-one perturbation  $B : D(A) \to X$  by

$$Bx := \varphi(x)y, \quad x \in D(A).$$

The operator (B, D(A)) satisfies the assumptions of Theorem 5.19, and hence it is a Miyadera– Voigt perturbation. To see this let  $\varepsilon' > 0$  and  $p \in \mathscr{P}$  be arbitrary and observe that

$$p\left(BR(\lambda, A)x\right) = p\left(\varphi\left(R(\lambda, A)x\right)y\right) = \left|\varphi\left(R(\lambda, A)x\right)\right|p(y), \quad x \in X$$

Since we always assume that  $\mathcal{P}$  is a directed family of seminorms and  $\varphi$  is  $\tau_A$  continuous we conclude that there exists M > 0 and  $p', q' \in \mathcal{P}$  such that

$$|\varphi(R(\lambda, A)x)| \le M\left(p'(R(\lambda, A)x) + q'(x)\right), \quad x \in X.$$

By [54, Lemma 1.2.23] and Remark 5.14 the operator  $R(\lambda, A)$ ,  $\lambda \in \rho(A)$ , is local, i.e., for each  $\varepsilon' > 0$  there exists K' > 0 and  $q'' \in \mathcal{P}$  such that

$$p'(R(\lambda, A)x) \le K'q''(x) + \varepsilon' ||x||, \quad x \in X.$$

This leads with K'' := p(y)MK' to the following inequality

$$p(BR(\lambda, A)x) \le K''\left(q''(x) + \frac{1}{M}q'(x)\right) + \varepsilon \|x\|, \quad x \in X.$$

where  $\varepsilon = Mp(y)\varepsilon'$ . We conclude from the above that  $q'' + \frac{1}{M}q' \in \mathcal{P}$  and we are done. Since (A, D(A)) is a Hille–Yosida operator one gets

$$\|BR(\lambda,A)x\| \le |\varphi(R(\lambda,A)x| \cdot \|y\| \le \|\varphi\| \|R(\lambda,A)x\| \|y\| \le \|\varphi\| \frac{M}{|\lambda-\omega|} \|x\| \|y\|.$$

Since all constants are fixed from the beginning our expression becomes smaller than 1 by chosing  $\lambda > \omega$  big enough. This show that rank-one perturbations fit in our setting. Observe that the argumentation can be extended to finite rank operators.

#### 5.3.2 Gauss–Weierstrass semigroup on $M(\mathbb{R})$

Now we discuss an explicit example of a bi-continuous semigroup on the space  $M(\mathbb{R})$  of bounded Borel measures on  $\mathbb{R}$  with respect to the Borel  $\sigma$ -algebra  $\mathscr{B}(\mathbb{R})$ . We noticed in Section 1.6 that every bi-continuous semigroup on  $C_b(\mathbb{R})$  with respect to the compact-open topology  $\tau_{co}$  gives rise to a bi-continuous semigroup on  $M(\mathbb{R})$  with respect to  $\tau^\circ = \sigma(C_b(\mathbb{R}), M(\mathbb{R}))$  and vice versa.

For what follows recall from [25, Sect. 1.1] or [23, Sect. 3.8] the definition of the (centered) Gaussian measure  $\gamma_t$ , t > 0, on  $\mathbb{R}$  defined by

$$\gamma_t(\Omega) = \frac{1}{\sqrt{2\pi t}} \int_{\Omega} e^{-\frac{|x|^2}{2t}} d\lambda^1, \quad \Omega \in \mathscr{B}(\mathbb{R}),$$

where  $\lambda^1$  denotes the Lebesgue measure on  $\mathbb{R}$ . This measure  $\gamma_t$ , t > 0, is a strictly positive bounded Borel measure and in particular a Radon measure. As a matter of fact  $\gamma_t$ , t > 0, is the Lebesgue measure with density  $\varphi_t$  given by

$$\varphi_t(x) = \frac{1}{\sqrt{2\pi t}} e^{-\frac{|x|^2}{2t}}, \quad t > 0, \ x \in \mathbb{R}.$$

Recall from [23, Def. 3.9.8] and [24, Sect. 1.2] that for  $\mu, \nu \in M(\mathbb{R})$  one defines the convolution of  $\mu$  and  $\nu$  by

$$(\mu * \nu)(\Omega) = \int_{\mathbb{R}} \int_{\mathbb{R}} \mathbf{1}_{\Omega}(x+y) \, \mathrm{d}\mu(x) \, \mathrm{d}\nu(y) = \int_{\mathbb{R}} \nu(\Omega-x) \, \mathrm{d}\mu(x).$$

If one of the measures  $\mu, \nu \in \mathcal{M}(\mathbb{R})$  has some density with respect to the one-dimensional Lebesgue measure  $\lambda^1$ , say  $\nu = g \cdot \lambda^1$ , as it is the case for  $\gamma_t$ , then the convolution has density  $g * \mu$  defined by

$$(g * \mu)(x) = \int_{\mathbb{R}} g(x - y) \, \mathrm{d}\mu(y).$$

Now we define a family of operators  $(T(t))_{t\geq 0}$  on  $\mathcal{M}(\mathbb{R})$  by  $T(0)\mu = \mu$  and

$$T(t)\mu = \gamma_t * \mu, \quad t > 0, \ \mu \in \mathcal{M}(\mathbb{R}).$$
(5.3.1)

Since  $\gamma_t \geq 0$  for each t > 0 we conclude that the family  $(T(t))_{t\geq 0}$  consists of positive operators on  $\mathcal{M}(\mathbb{R})$ . The next result shows that  $(T(t))_{t\geq 0}$  is in fact a bi-continuous semigroup on  $\mathcal{M}(\mathbb{R})$ equipped with the total variation norm and with respect to  $\sigma(\mathcal{C}_{\mathrm{b}}(\mathbb{R}), \mathcal{M}(\mathbb{R}))$ . Recapitulate that the Gauss–Weierstrass semigroup  $(T_*(t))_{t\geq 0}$  on  $C_b(\mathbb{R})$  is defined by T(0) = I and

$$(T_*(t)f)(x) := (\varphi_t * f)(x) = \frac{1}{\sqrt{2\pi t}} \int_{\mathbb{R}} e^{-\frac{|x-y|^2}{2t}} f(y) \, \mathrm{d}y, \quad t > 0, \ f \in \mathcal{C}_{\mathbf{b}}(\mathbb{R}), \ x \in \mathbb{R}$$

Having this in mind, we can prove the following theorem.

**Theorem 5.21.** The semigroup  $(T(t))_{t\geq 0}$  is the adjoint of the Gauss–Weierstrass semigroup  $(T_*(t))_{t\geq 0}$  on  $C_b(\mathbb{R})$ .

*Proof.* We observe that for  $f \in C_b(\mathbb{R})$  with  $f \ge 0$  and  $\mu \in M(\mathbb{R})$  the following holds.

$$\begin{split} \langle f, T(t)\mu \rangle &= \left| \int_{\mathbb{R}} f(x) \, \mathrm{d}(T(t)\mu)(x) \right| \\ &= \left| \int_{\mathbb{R}} f(x) \, \mathrm{d}(\gamma_t * \mu)(x) \right| \\ &= \left| \int_{\mathbb{R}} f(x) \, \mathrm{d}((\varphi_t \cdot \lambda^1) * \mu)(x) \right| \\ &= \left| \int_{\mathbb{R}} f(x) \, \mathrm{d}((\varphi_t * \mu) \cdot \lambda^1)(x) \right| \\ &= \left| \int_{\mathbb{R}} f(x) (\varphi_t * \mu)(x) \, \mathrm{d}\lambda^1(x) \right| \\ &= \left| \int_{\mathbb{R}} f(x) \int_{\mathbb{R}} \varphi_t(x - y) \, \mathrm{d}\mu(y) \, \mathrm{d}\lambda^1(x) \right| \\ &= \left| \int_{\mathbb{R}} (f * \varphi_t)(y) \, \mathrm{d}\mu(y) \right| \\ &= \left| \int_{\mathbb{R}} T_*(t) f(y) \, \mathrm{d}\mu(y) \right| \\ &= \langle T_*(t) f, \mu \rangle \end{split}$$

Here  $\langle \cdot, \cdot \rangle$  denotes the pairing between  $C_b(\mathbb{R})$  and  $M(\mathbb{R})$ . This proves the assertion.

- **Remark 5.22.** (i) The notation  $(T_*(t))_{t\geq 0}$  is intuitive since  $(T_*(t))_{t\geq 0}$  is the preadjoint of  $(T(t))_{t\geq 0}$ , i.e.,  $T_*(t)^* = T(t)$  for all  $t \geq 0$ .
- (ii) Since  $(T_*(t))_{t\geq 0}$  is known to be bi-continuous on  $C_b(\mathbb{R})$  with respect to the compactopen topology we conclude by [57, Thm. 3.5] that  $(T(t))_{t\geq 0}$  on  $M(\mathbb{R})$  is bi-continuous with respect to  $\sigma(C_b(\mathbb{R}), M(\mathbb{R}))$ .
- (iii) Recall that the generator  $(A_*, D(A_*))$  of the Gauss–Weierstrass semigroup on  $C_b(\mathbb{R})$  is given by  $(\Delta, C_b^2(\mathbb{R}))$  where  $\Delta$  denotes the Laplacian and  $C_b^2(\mathbb{R})$  the space of twice continuous differentiable functions with bounded derivatives.
- (iv) We also notice that  $\mathbb{C}_+ \subseteq \rho(A_*)$ , where  $\mathbb{C}_+$  denotes the right-half plane, i.e., all complex numbers with positive real part.
- (v) To see that  $(T(t))_{t\geq 0}$  is indeed a semigroup one can also use the uniqueness of the Fourier-transform.

The next step is to determine the generator (A, D(A)) of  $(T(t))_{t\geq 0}$ . To do so we have to consider differentiable measures. The original study of such measures is due to S.V. Fomin [58] and A.V. Skorohod [105], [106, Chapter 4, Sect. 21]. In [24] V.I. Bogachev discusses differentiable measures in more details.

**Definition 5.23.** A Borel measure  $\mu$  on  $\mathbb{R}$  is called *Skorohod differentiable* or *S*-differentiable if, for every function  $f \in C_b(\mathbb{R})$ , the function

$$t \mapsto \int_{\mathbb{R}} f(x-t) \, \mathrm{d}\mu(x),$$

is differentiable.

The following theorem [24, Thm. 3.6.1] shows that the previous definition is equivalent to the existence of a Borel measure  $\nu$ , called the Skorohod derivative of  $\mu$ , such that for each bounded continuous function on  $\mathbb{R}$  one has

$$\lim_{t \to 0} \int_{\mathbb{R}} \frac{f(x-t) - f(x)}{t} \, \mathrm{d}\mu(x) = \int_{\mathbb{R}} f(x) \, \mathrm{d}\nu(x)$$
(5.3.2)

**Theorem 5.24.** Let  $\mu$  be a Skorohod differentiable Borel measure on  $\mathbb{R}$ . Then there exists a Borel measure  $\nu$  which is its Skorohod derivative, i.e.,  $\nu$  satisfies (5.3.2) for all bounded continuous functions f on  $\mathbb{R}$ .

- **Remark 5.25.** (i) The Skorohod derivative  $\nu$  of  $\mu$  as it appears in (5.3.2) and Theorem 5.24 is denoted by  $\mathscr{D}\mu$ , i.e.,  $\nu = \mathscr{D}\mu$ .
  - (ii) By [24, Prop. 3.4.1] one has that for a bounded Borel measure μ on R the measure μ is Skorohod differentiable if and only if it has density of bounded variation, in particular every Skorohod differentiable measure on R admits a bounded density. In that case the density of the Skorohod derivative is just the (generalized) derivative of the original density.
- (iii) For higher order derivatives we recall from [24, Prop. 3.7.1] that if the map  $t \mapsto \int_{\mathbb{R}} f(x-t) d\mu(x)$  is *n*-times differentiable, then  $\mu$  is *n*-times Skorohod differentiable, i.e., for all  $f \in C_{\rm b}(\mathbb{R})$  the function

$$(t_1,\ldots,t_n)\mapsto \int_{\mathbb{R}} f(x+t_1+\cdots+t_n) \, \mathrm{d}\mu(x),$$

has partial derivatives  $\partial_{t_1} \cdots \partial_{t_n}$ .

(iv)  $\mu$  is Skorohod differentiable with Skorohod derivative  $\mathscr{D}\mu$  if and only if for each  $f \in C_{\mathrm{b}}^{\infty}(\mathbb{R})$ 

$$\int_{\mathbb{R}} \frac{\mathrm{d}}{\mathrm{d}x} f(x) \, \mathrm{d}\mu(x) = -\int_{\mathbb{R}} f(x) \, \mathrm{d}(\mathscr{D}\mu)(x).$$

In order to determine the generator of  $(T(t))_{t\geq 0}$  on  $\mathcal{M}(\mathbb{R})$  we use Lemma 1.19.

**Theorem 5.26.** The generator (A, D(A)) of  $(T(t))_{t\geq 0}$  is given by

 $A\mu:=\Delta\mu, \quad D(A):=\left\{\mu\in {\rm M}(\mathbb{R}): \mu \ is \ twice \ Skorohod \ differentiable \ \right\},$ 

where  $\Delta \mu$  denotes the second order Skorohod derivative of  $\mu$ .

*Proof.* Let us denote the generator of our semigroup  $(T(t))_{t\geq 0}$  by (C, D(C)). By Lemma 1.19 and Remark 5.22 we conclude that the domain of the generator is of the following form

$$D(C) = \left\{ \mu \in \mathcal{M}(\mathbb{R}) : \exists \nu \in \mathcal{M}(\mathbb{R}) \; \forall f \in \mathcal{C}^2_{\mathrm{b}}(\mathbb{R}) : \langle f, \nu \rangle = \langle \Delta f, \mu \rangle \right\}.$$

Now let  $\mu \in M(\mathbb{R})$  be twice Skorohod differentiable, i.e.,  $\mu \in D(A)$ , and denote its second derivative with  $\nu := \Delta \mu$ . For  $f \in C_b^2(\mathbb{R})$  one has

$$\langle \Delta f, \mu \rangle = \int_{\mathbb{R}} \Delta f \, \mathrm{d}\mu = \int_{\mathbb{R}} f \, \mathrm{d}\nu = \langle f, \nu \rangle \,,$$

showing that  $D(A) \subseteq D(C)$ . For the converse let  $\mu \in D(C)$ , i.e., there exists  $\nu \in M(\mathbb{R})$  such that for each  $f \in C^2_b(\mathbb{R})$  holds that

$$\langle f, \nu \rangle = \langle \Delta f, \mu \rangle = \left\langle \lim_{t \to 0} \frac{T_*(t)f - f}{t}, \mu \right\rangle = \left\langle f, \lim_{t \to 0} \frac{T(t)\mu - \mu}{t} \right\rangle.$$

From this we may conclude that  $\sigma(C_b(\mathbb{R}), M(\mathbb{R})) - \lim_{t \to 0} \frac{T(t)\mu - \mu}{t}$  exists. Combining this with [24, Thm. 3.6.4] we obtain that  $\mu \in D(A)$ . This finishes the proof.

**Remark 5.27.** In Remark 5.22 we observed that  $\mathbb{C}_+ \subseteq \rho(A)$  and since  $\sigma(A_*) = \sigma(A)$  by [52, Chapter II, Sect. 2.5], we also obtain that  $\mathbb{C}_+ \subseteq \rho(A)$ , especially,  $\lambda \in \rho(A)$  for  $\lambda > 0$ . Moreover we observe that  $(T(t))_{t\geq 0}$  is a bounded semigroup and hence we also conclude that  $\mathbb{C}_+ \subseteq \rho(A)$ .

Now fix a positive, Lebesgue integrable and unbounded function  $\psi : \mathbb{R} \to \mathbb{R}$  and define  $B : D(A) \to M(\mathbb{R})$  by

$$B\mu := \psi \cdot \mu, \quad \mu \in D(A).$$

To show that the operator B satisfies the conditions of Theorem 5.19 let  $\lambda \in \rho(A)$ , especially, we can choose  $\lambda \in \mathbb{R}$  with  $\lambda > 0$  and observe that by the duality between  $C_{b}(\mathbb{R})$  and  $M(\mathbb{R})$ 

$$p_f(BR(\lambda, A)\mu) = |\langle f, BR(\lambda, A)\mu\rangle| = |\langle f\psi, R(\lambda, A)\mu\rangle| = |\langle R(\lambda, A_*)(f\psi), \mu\rangle|$$

for all  $f \in C_b(\mathbb{R})$  and  $\mu \in D(A)$ . Moreover, by an application of Fubini's theorem and an explicit integral one obtains

$$|R(\lambda, A_*)(f\psi)(x)| = \left| \int_0^\infty \int_{\mathbb{R}} e^{-\lambda t} \varphi(x-y) f(y) \psi(y) \, dy \, dt \right|$$
$$= \left| \int_{\mathbb{R}} \int_0^\infty e^{-\lambda t} \varphi(x-y) f(y) \psi(y) \, dt \, dy \right|$$
$$= \left| \int_{\mathbb{R}} \frac{1}{\sqrt{2\lambda}} e^{-\sqrt{2\lambda}|x-y|} f(y) \psi(y) \, dy \right|$$
$$= \left| (\xi_\lambda * f\psi)(x) \right|,$$

where  $\xi_{\lambda}(x) := \frac{1}{\sqrt{2\lambda}} e^{-\sqrt{2\lambda}|x|}$  is continuous and hence the convolution  $\xi_{\lambda} * f\psi$  is continuous. Furthermore, by Young's convolution inequality and the assumption that  $\psi$  is integrable we obtain

$$\left\|\xi_{\lambda} * f\psi\right\|_{\infty} \le \left\|\xi_{\lambda}\right\|_{\infty} \left\|f\psi\right\|_{1} \le \left\|\xi_{\lambda}\right\|_{\infty} \left\|f\right\|_{\infty} \left\|\psi\right\|_{1} < \infty$$

hence  $h := \xi_{\lambda} * f \psi \in C_{b}(\mathbb{R})$  and  $p_{f}(BR(\lambda, A)\mu) = p_{h}(\mu)$ . By using that  $M(\mathbb{R})$  is the dual of  $C_{b}(\mathbb{R})$  we obtain the following norm estimate

$$\|BR(\lambda, A)\mu\| = \sup_{\substack{f \in \mathcal{C}_{\mathcal{b}}(\mathbb{R}) \\ \|f\|_{\infty} \leq 1}} |\langle f, BR(\lambda, A)\mu\rangle| = \sup_{\substack{f \in \mathcal{C}_{\mathcal{b}}(\mathbb{R}) \\ \|f\|_{\infty} \leq 1}} |\langle R(\lambda, A_{*})f\psi, \mu\rangle| \leq \sup_{\substack{f \in \mathcal{C}_{\mathcal{b}}(\mathbb{R}) \\ \|f\|_{\infty} \leq 1}} \|R(\lambda, A_{*})f\psi\|_{\infty} \|\mu\|$$

Moreover,

$$\begin{aligned} |R(\lambda, A_*)(f\psi)(x)| &= \left| \int_0^\infty e^{-\lambda t} T_*(t) f(x) \psi(x) \, dt \right| \\ &= \left| \int_0^\infty e^{-\lambda t} \int_{\mathbb{R}} \varphi_t(x-y) f(y) \psi(y) \, dy dt \right| \\ &\leq \|f\|_\infty \int_0^\infty e^{-\lambda t} (\varphi * \psi)(x) \, dt \\ &\leq \|f\|_\infty \|\psi\|_1 \int_0^\infty e^{-\lambda t} \|\varphi_t\|_\infty \, dt \\ &= \|f\|_\infty \|\psi\|_1 \int_0^\infty \frac{e^{-\lambda t}}{\sqrt{2\pi t}} \, dt \\ &= \frac{\|f\|_\infty \|\psi\|_1}{\sqrt{2\lambda}} \end{aligned}$$

Hence, for  $\lambda$  big enough we obtain  $||R(\lambda, A_*)f\psi|| < 1$  and hence  $||BR(\lambda, A)|| < 1$ . Now we can apply Theorem 5.19 and conclude that A + B generates a positive bi-continuous semigroup on  $M(\mathbb{R})$ .

**Remark 5.28.** Suppose that  $f \in L^p(\mathbb{R})$  is unbounded. Observe that there exists  $g \in L^1(\mathbb{R})$  and  $h \in L^{\infty}(\mathbb{R})$  such that f = g+h. In particular, we consider the operator  $B : D(A) \to M(\mathbb{R})$  defined by  $B\mu := f \cdot \mu$  as above. Since the case for  $f \in L^1(\mathbb{R})$  is treated before and  $f \in L^{\infty}(\mathbb{R})$  gives rise to a bounded perturbation we conclude that we can extend our previous result to the whole scale of  $L^p$ -spaces.

# § 5.4 Notes

As already mentioned in the beginning, this work is inspired by the work of J. Voigt [119] and B. Farkas [54]. As mentioned in Section 4.4, we are also looking for a Desch–Schappacher perturbation theorem for positive bi-continuous semigroups similar to the results for strongly continuous semigroups presented in [20]. To do so we have to extend the notion of AM-spaces for our purpose as we have done it for AL-spaces in this chapter by means of Definition 5.9.

Even if the example of the Gauss–Weierstrass semigroup on  $M(\mathbb{R})$  is the adjoint of the wellknown Gauss–Weierstrass semigroup on  $C_b(\mathbb{R})$  it is interesting how differentiable measures come into play. Differentiable measures as well as Gaussian measures as they are used in this chapter are discussed in more detail in [24], [23] and [25]. There not only the space  $\mathbb{R}$ is treated but locally convex spaces in general including infinite dimensional spaces. This raises the first difficulties in this theory, especially, as a matter of fact, there is no canonical infinite dimensional substitute for Lebesgue measure. For example, on an infinite dimensional, separable Banach space X the only locally finite and translation invariant Borel measure  $\mu$  on X is the trivial measure, i.e.,  $\mu(A) = 0$  for each measurable set  $A \subseteq X$ . For more general statements on invariant measures on infinite dimensional vector spaces we refer for example to the work of V.N. Sudakov [110] or Y. Umemura [115].
## Chapter 6

# Implemented Semigroups

## Introduction

The question whether automorhisms and operators are implemented arises naturally if one studies the quantum mechanical aspects in mathematical physics. Typically issues of symmetries apply here. For example Ludwig's Theorem states that on every Hilbert space  $\mathcal{H}$  with dim $(\mathcal{H}) \geq 3$  each order-lattice automorphism  $\alpha$  of the effect algebra, defined by  $\mathcal{E} := \{T \in \mathscr{L}(\mathcal{H}) : 0 \le T \le 1\},$  is unitarily implemented, i.e., there exists a unitary operator  $U \in \mathscr{L}(\mathcal{H})$  such that  $\alpha(T) = UTU^*$  for each  $T \in \mathcal{E}$ , see for example [30]. Another important symmetry phenomenon in the operator algebraic setting of quantum mechanics is the so-called Wigner symmetry [81, Chapter 5]. In this context one also see implemented semigroups appearing. As it is stated in [26, Chapter 3, Section 2], all continuous one-parameter groups of Wigner symmetries are implemented by strongly continuous one-parameter groups of unitary operators. The most common setting in quantum mechanics is the case of Hilbert spaces. Implemented semigroups on Hilbert spaces also appear in other resources. In particular, in [45, Sect. 4.4, 6.1, 7.1, 7.2] T. Eisner discusses stability of implemented operators as well as of implemented semigroups with applications to Lyapunov's equations. In [11, Chapter D-IV] the point of interest is the asymptotic behaviour of positive implemented semigroups on  $C^*$ -algebras and von Neumann algebras. The implemented semigroups also come to light if one considers the following operator equation

$$AX + XB = Y, \ X \in \mathscr{L}(F, E),$$

where A and B are generators of strongly continuous semigroups on Banach spaces E and F. This operator-valued equation has been considered by several authors [12, 97, 98, 99, 61]. We now consider extrapolation spaces and Desch–Schappacher perturbations of implemented semigroup on Banach spaces. In particular, we round out the work of J. Alber [6], where extrapolation spaces for this semigroups are studied but only with respect to the space of strong continuity. With the construction from Chapter 2 we are able to extrapolate implemented semigroups in a more general context. We will use this to consider Desch–Schappacher perturbations of implemented semigroups have a module structure.

This chapter is organize as follows: the first section is devoted to the fact that the space  $\mathscr{L}(E)$  of bounded linear operators on a Banach space E satisfies Assumption 1.1 with re-

spect to the strong operator topology. Moreover we show that the implemented semigroup is indeed bi-continuous. In the second section we continue with intermediate and extrapolation spaces of this semigroup and apply this to Desch–Schappacher perturbations of implemented semigroups in the last section.

## § 6.1 Preliminaries

#### 6.1.1 The space of bounded linear operators $\mathscr{L}(E)$

We consider the space  $X := \mathscr{L}(E)$  of bounded linear operators on a Banach space E. The operator norm defined by

$$||T|| := \sup_{||x|| \le 1} ||Tx||, \quad T \in \mathscr{L}(E),$$

makes  $\mathscr{L}(E)$  a Banach space. The so-called strong operator topology  $\tau_{\text{sot}}$  is defined to be the coarsest topology that makes the evaluation maps  $T \mapsto Tx$  continuous for each  $x \in E$  (see also [90]). For  $S \in \mathscr{L}(E)$ ,  $x \in E$  and  $\varepsilon > 0$  we defined

$$\mathcal{U}(S, x, \varepsilon) := \left\{ T \in \mathscr{L}(E) : \|Tx - Sx\| < \varepsilon \right\}.$$

This gives us a subbase for the strong operator topology. In particular we observe that a net  $(T_{\iota})_{\iota \in I}$  in  $\mathscr{L}(E)$  converges to  $T \in \mathscr{L}(E)$  with respect to the strong operator topology if and only if  $||T_i x - Tx|| \to 0$  for each  $x \in E$ . This gives rise to a family of seminorms  $\mathcal{P}$  which generates the (locally convex) strong operator topology and is given by

$$\mathcal{P} = \{ p_x : x \in E \}, \quad p_x(T) := ||Tx||.$$

Of course  $\tau_{\text{sot}}$  is Hausdorff and coarser than the operator norm topology on  $\mathscr{L}(E)$ . In order to show that  $\tau_{\text{sot}}$  is sequentially complete on norm-bounded set consider  $(T_n)_{n\in\mathbb{N}}$  to be a norm-bounded  $\tau_{\text{sot}}$ -Cauchy sequence. We conclude that  $(T_n(x))_{n\in\mathbb{N}}$  is a Cauchy sequence in E and hence convergent. Hence we define  $T: E \to E$  by

$$Tx := \lim_{n \to \infty} T_n(x), \quad x \in E.$$

By linearity of the operators  $(T_n)_{n \in \mathbb{N}}$  we achieve that T is linear as well. In view of the fact that  $(T_n)_{n \in \mathbb{N}}$  was supposed to be bounded with respect to the operator norm, we obtain

$$||T(x)|| \le \sup_{n \in \mathbb{N}} ||T_n(x)|| < \infty,$$

and hence  $T \in \mathscr{L}(E)$ . By construction we attain that  $T_n \to T$  with respect to  $\tau_{\text{sot}}$ . For the norming property we follow [103, Thm. 2.1.6], in fact fix sequences  $(x_n)_{n \in \mathbb{N}}$  in E and  $(\varphi_n)_{n \in \mathbb{N}}$  in E' with  $||x_n|| \leq 1$  and  $||\varphi_n|| \leq 1$  for each  $n \in \mathbb{N}$  and such that

$$||T|| = \sup_{n \in \mathbb{N}} ||Tx_n||$$
 and  $||Tx_n|| = |\varphi_n(Tx_n)|$ .

Now define  $\Lambda_n \in (\mathscr{L}(E), \tau_{\text{sot}})'$ 

$$\Lambda_n(T) := \varphi_n(Tx_n)$$

and observe that  $\|\Lambda_n\| \leq 1$  for each  $n \in \mathbb{N}$ . Moreover

$$||T|| = \sup_{n \in \mathbb{N}} ||Tx_n|| = \sup_{n \in \mathbb{N}} |\varphi_n(T(x_n))| = \sup_{n \in \mathbb{N}} |\Lambda_n(T)| \le \sup_{\substack{\varphi \in (\mathscr{L}(E), \tau_{\text{sot}})' \\ \|\varphi\| \le 1}} |\varphi(T)| \le ||T||$$

which gives us the norming property. We remark that the dual of  $(\mathscr{L}(E), \tau_{sot})$  can be identified with the set  $\mathcal{F}(E)$  of finite rank operators on E (see [44]).

#### 6.1.2 The implemented semigroup

Let *E* be some Banach space and  $(T(t))_{t\geq 0}$ ,  $(S(t))_{t\geq 0}$  two  $C_0$ -semigroups on *E*. We already showed that  $\mathscr{L}(E)$  satisfies Assumption 1.1 if we equip  $\mathscr{L}(E)$  with the strong operator topology  $\tau_{\text{sot}}$ . The semigroup  $(\mathcal{U}(t))_{t\geq 0}$  on  $\mathscr{L}(E)$  defined by

$$\mathcal{U}(t)L := T(t)LS(t), \quad L \in \mathscr{L}(E), \ t \ge 0,$$

is called the *implemented semigroup*. If  $T(t) \equiv I$  or  $S(t) \equiv I$ , then the corresponding semigroups  $(\mathcal{U}_R)(t)_{t\geq 0}$  and  $(\mathcal{U}_L)(t)_{t\geq 0}$  are called *right implemented* and *left implemented semigroup*, respectively. As F. Kühnemund already mentioned in [78], these semigroups play an important role in quantum mechanics (see also [26]). To show that  $(\mathcal{U}(t))_{t\geq 0}$  is not strongly continuous in general, let  $(T(t))_{t\geq 0}$  and  $(S(t))_{t\geq 0}$  be two strongly continuous semigroups on the Banach space  $E = L^1(\mathbb{R}, \mathbb{C})$  defined by

$$(T(t)f)(x) := f(x+t), \quad (S(t)f)(x) = \overline{(T(t)f)(s)}, \quad t \ge 0, \ f \in L^1(\mathbb{R}, \mathbb{C}), \ x \in \mathbb{R}.$$

By considering the operator  $L \in \mathscr{L}(E)$ , defined by

$$Lf := \mathbf{1}_{[0,1]}f, \quad f \in \mathrm{L}^1(\mathbb{R}, \mathbb{C}),$$

one recognizes that the implemented semigroup on the space  $\mathscr{L}(E)$  defined by  $\mathcal{U}(t)L := T(t)LS(t), t \geq 0, L \in \mathscr{L}(E)$ , is not strongly continuous with respect to the operator norm, i.e., let  $(f_n)_{n\in\mathbb{N}}$  be a sequence in  $L^1(\mathbb{R},\mathbb{C})$  such that  $||f_n|| = 1$  and  $\operatorname{supp}(f_n) \subseteq \left[0, \frac{1}{n}\right]$  for all  $n \in \mathbb{N}$  and observe that

$$\left\| \mathcal{U}\left(\frac{1}{n}\right) Lf_n - Lf_n \right\|_1 = \left\| \mathbf{1}_{\left[\frac{1}{n}, \frac{1}{n}+1\right]} f_n - \mathbf{1}_{[0,1]} f_n \right\|_1 = \|f_n\|_1 = 1.$$

Consequently,  $(\mathcal{U}(t))_{t\geq 0}$  is not strongly continuous with respect to the norm on  $L^1(\mathbb{R}, \mathbb{C})$ , cf. [5]. We remark that  $(\mathcal{U}(t))_{t\geq 0}$  is a  $C_0$ -semigroup if and only if both semigroups  $(T(t))_{t\geq 0}$  and  $(S(t))_{t\geq 0}$  are uniformly continuous meaning that

$$\lim_{t \to 0} \|T(t) - I\| = \lim_{t \to 0} \|S(t) - I\| = 0.$$

Now we show that the implemented semigroup is bi-continuous with respect to  $\tau_{\text{sot}}$  (see also [78, Prop. 3.16]). Since  $(T(t))_{t\geq 0}$  and  $(S(t))_{t\geq 0}$  are (automatically) exponentially bounded, also  $(\mathcal{U}(t))_{t\geq 0}$  is. For  $x \in E$  we obtain:

$$p_x(\mathcal{U}(t)L - L) = \|T(t)LS(t)x - Lx\| \le \|T(t)L(S(t)x - x)\| + \|T(t)Lx - Lx\|$$

which proves the strong continuity. For the local bi-equicontinuity let  $(L_n)_{n\in\mathbb{N}}$  be a normbounded strongly convergent null sequence. For  $t_0 > 0$  and  $x \in E$  holds

$$\sup_{t \in [0,t_0]} p_x(\mathcal{U}(t)L_n) = \sup_{t \in [0,t_0]} \|T(t)L_nS(t)x\| \le M \cdot \sup_{t \in [0,t_0]} \|L_nS(t)x\|,$$

where  $M := \sup_{t \in [0,t_0]} ||T(t)||$ , which is finite by Proposition [52, Chapter I, Sect. 5, Prop. 5.5]. Since  $(L_n)_{n \in \mathbb{N}}$  converges uniformly on compact sets and both semigroups are strongly continuous on E we conclude that the right-hand side tends to zero, which proves that  $(\mathcal{U}(t))_{t \geq 0}$ is locally bi-equicontinuous.

## § 6.2 Intermediate and extrapolation spaces

#### 6.2.1 Favard- and Hölder spaces

In what follows we restrict our self to left implemented semigroups  $(\mathcal{U}_L(t))_{t\geq 0}$ . Moreover assume that the implementing  $C_0$ -semigroup  $(T(t))_{t\geq 0}$  has negative growth bound, see Definition 1.4. As we just noticed,  $(\mathcal{U}_L(t))_{t\geq 0}$  is a bi-continuous semigroup on  $\mathscr{L}(E)$  with respect to the strong operator topology and still has negative growth bound. We determine the intermediate and extrapolation spaces for this semigroup. We can write:

$$||B||_{F_{\alpha}(\mathcal{U}_{L})} = \sup_{t>0} \frac{||\mathcal{U}_{L}(t)B - B||}{t^{\alpha}} = \sup_{t>0} \frac{||T(t)B - B||}{t^{\alpha}}$$
$$= \sup_{t>0} \sup_{\|x\| \le 1} \frac{||T(t)Bx - Bx||}{t^{\alpha}} = \sup_{\|x\| \le 1} \sup_{t>0} \frac{||T(t)Bx - Bx||}{t^{\alpha}} = \sup_{\|x\| \le 1} ||Bx||_{F_{\alpha}(T)}.$$

From this we conclude the following.

**Proposition 6.1.** Let  $(\mathcal{U}_L(t))_{t\geq 0}$  be the semigroup which is left implemented by  $(T(t))_{t\geq 0}$ . Then for  $\alpha \in (0, 1]$ 

$$F_{\alpha}(\mathcal{U}_L) = \mathscr{L}(E, F_{\alpha}(T))$$

with the same norms.

From the definition we obtain that

$$X_{\alpha}(\mathcal{U}_{L}) = \left\{ B \in \mathscr{L}(E) : \ \tau \lim_{t \to 0} \frac{\mathcal{U}_{L}(t)B - B}{t^{\alpha}} = 0, \ \|B\|_{F_{\alpha}(\mathcal{U}_{L})} < \infty \right\}$$
$$= \left\{ B \in \mathscr{L}(E) : \ \lim_{t \to 0} \frac{\|T(t)Bx - Bx\|}{t^{\alpha}} = 0 \quad \text{for all } x \in E \right\},$$
$$\underline{X}_{\alpha}(\mathcal{U}_{L}) = \left\{ B \in \mathscr{L}(E) : \ \lim_{t \to 0} \frac{T(t)B - B}{t^{\alpha}} = 0 \right\}.$$

**Proposition 6.2.** Let  $(\mathcal{U}_L(t))_{t\geq 0}$  be the semigroup which is left implemented by  $(T(t))_{t\geq 0}$ . Then

$$X_{\alpha}(\mathcal{U}_L) = \mathscr{L}(E, X_{\alpha}(T))$$

with the same norms.

### 6.2.2 Extrapolation

We now turn to the extrapolation spaces. For the  $C_0$ -semigroup  $(\underline{\mathcal{U}}_L(t))_{t\geq 0}$  on the space  $\underline{X}_0$ these have been studied by J. Alber in [6]. He has shown that the generator  $\mathcal{G}$  of  $(\mathcal{U}_L(t))_{t\geq 0}$ is given by

$$\mathcal{G}V = A_{-1}V$$

on

$$D(\mathcal{G}) = \{ V \in \mathscr{L}(E) : A_{-1}V \in \mathscr{L}(E) \}$$

where  $A_{-1}$  denotes the generator of the extrapolated  $C_0$ -semigroup  $(T_{-1}(t))_{t\geq 0}$  on  $E_{-1}$ . The extrapolation spaces  $X_{-1}$  and  $\underline{X}_{-1}$  can now be obtained by Theorem 2.16. For that let

$$\mathscr{E} = \{ S : E \to E_{-\infty} : \text{ linear and continuous} \},\$$

where  $E_{-\infty}$  is the universal extrapolation space of  $(T(t))_{t\geq 0}$  (see the paragraph preceding Theorem 2.16), and let  $i: \mathscr{L}(E) \to \mathscr{E}$  be the identity. Consider the operator-valued multiplication operator

$$\mathcal{A}V = A_{-\infty}V, \quad V \in \mathscr{E},$$

where  $A_{-\infty}x = A_{-(n-1)}x$  for  $x \in E_{-n}$ . Notice that  $\lambda - \mathcal{A} : X_0 \to \mathscr{E}$  is injective for  $\lambda > 0$ since  $A_{-\infty}$  and  $A_{-1}$  coincide on E. Hence by applying Theorem 2.16 we obtain

$$X_{-1} = \{A_{-1}V : V \in \mathscr{L}(E)\}$$

and

$$\underline{X}_{-1} = \{A_{-1}V : \quad V \in \underline{X}_0\}.$$

From this we conclude that

$$X_{-1} = \left\{ V \in \mathscr{L}(E, E_{-1}) : \exists (V_n)_{n \in \mathbb{N}} \subseteq \mathscr{L}(E) \text{ with } V_n \to V \text{ strongly} \right\} = \overline{\mathscr{L}(E)}^{\mathscr{L}_{\text{stop}}(E, E_{-1})}.$$

Since for any  $C \in \mathscr{L}(E, E_{-1})$  we have  $nR(n, A_{-1})C \in \mathscr{L}(E)$  and  $nR(n, A_{-1})C \to C$  strongly as  $n \to \infty$ , we obtain

$$X_{-1} = \mathscr{L}(E, E_{-1}).$$

For  $\underline{X}_{-1}$  we have:

$$\underline{X}_{-1} = \left\{ V \in \mathscr{L}(E, E_{-1}) : \exists (V_n)_{n \in \mathbb{N}} \subseteq \mathscr{L}(E) \text{ with } V_n \to V \text{ in } \mathscr{L}(E, E_{-1}) \right\} = \overline{\mathscr{L}(E)}^{\mathscr{L}(E, E_{-1})}$$

This last statement is a result of J. Alber, see [6, Thm. 2.4], which we could recover as a simple consequence of the abstract techniques described in Section 2.1.2. Finally, we obtain by Corollary 2.26 and Remark 2.58 that for  $\alpha \in [0, 1)$ 

$$F_{-\alpha}(\mathcal{U}_L) = A_{-1}\mathscr{L}(E, F_{1-\alpha}(S)) = \mathscr{L}(E, F_{-\alpha}(T))$$

and

$$X_{-\alpha}(\mathcal{U}_L) = A_{-1}\mathscr{L}(E, X_{1-\alpha}(S)) = \mathscr{L}(E, X_{-\alpha}(T)).$$

### § 6.3 Perturbations

#### 6.3.1 Ideals in $\mathscr{L}(E)$ and module homomorphisms

Our purpose is to relate Desch–Schappacher perturbations of the  $C_0$ -semigroup  $(T(t))_{t\geq 0}$ and Desch–Schappacher perturbations of the left implemented semigroup  $(\mathcal{U}_L(t))_{t\geq 0}$ . Before doing so we need some auxiliary results exploiting the algebraic structure of the domain of Hille–Yosida operators on  $\mathscr{L}(E)$ .

**Lemma 6.3.** Let E be a Banach space and  $(\mathcal{A}, D(\mathcal{A}))$  a Hille–Yosida operator on  $\mathscr{L}(E)$ , *i.e.*, suppose there exists  $\omega \in \mathbb{R}$  and  $M \geq 1$  such that  $(\omega, \infty) \subseteq \rho(\mathcal{A})$  and

$$\|R(\lambda, \mathcal{A})^n\| \le \frac{M}{(\lambda - \omega)^n},$$

for each  $\lambda > \omega$  and  $n \in \mathbb{N}$ . The following are equivalent:

- (a)  $D(\mathcal{A})$  is a right ideal of the Banach algebra  $\mathscr{L}(E)$ , i.e.,  $CB \in D(\mathcal{A})$  whenever  $C \in D(\mathcal{A})$ ,  $B \in \mathscr{L}(E)$ , and  $\mathcal{A}$  is a right  $\mathscr{L}(E)$ -module homomorphism, i.e.,  $\mathcal{A}(CB) = \mathcal{A}(C)B$  for  $C \in D(\mathcal{A})$  and  $B \in \mathscr{L}(E)$ .
- (b) There exists a Hille-Yosida operator (A, D(A)) such that  $\mathcal{A}(C) = AC$ , where  $D(\mathcal{A}) = \mathscr{L}(E, D(A))$ .
- (c) There exists a Hille–Yosida operator (A, D(A)) such that  $\mathcal{A}(C) = A_{-1}C$ , where  $D(\mathcal{A}) = \{C \in \mathscr{L}(E) : A_{-1}C \in \mathscr{L}(E)\}.$

*Proof.* The implication (c)  $\Rightarrow$  (a) is just a checking of properties of an explicitly given operator. The implication (b)  $\Leftrightarrow$  (c) follows from the fact that the operator A and  $A_{-1}$  coincide on the domain D(A) of A.

(a)  $\Rightarrow$  (b) By definition one has  $R(\lambda, \mathcal{A}) \in \mathscr{L}(\mathscr{L}(E))$  whenever  $\lambda \in \rho(\mathcal{A})$ . Define for  $\lambda \in \rho(\mathcal{A})$ 

$$R(\lambda) := R(\lambda, \mathcal{A})(I).$$

Since  $R(\lambda, \mathcal{A}), \lambda \in \rho(\mathcal{A})$  satisfy the resolvent identity also  $R(\lambda), \lambda \in \rho(\mathcal{A})$  do:

$$R(\lambda) - R(\mu) = R(\lambda, \mathcal{A})(I) - R(\mu, \mathcal{A})(I) = (R(\lambda, \mathcal{A}) - R(\mu, \mathcal{A}))(I)$$
$$= ((\lambda - \mu)R(\lambda, \mathcal{A})R(\mu, \mathcal{A}))(I) = (\lambda - \mu)R(\lambda)R(\mu)$$

for each  $\lambda, \mu \in \rho(\mathcal{A})$ . Hence the family  $(R(\lambda))_{\lambda \in \rho(\mathcal{A})}$  is a pseudoresolvent. If  $R(\lambda)x = 0$  for some  $x \in E$ , then

$$0 = \lambda R(\lambda)x = \lambda R(\lambda, \mathcal{A})(I)x.$$

But since  $\lambda R(\lambda, \mathcal{A})(I) \to I$  as  $\lambda \to \infty$ , it follow that x = 0. Therefore  $R(\lambda)$  is injective, and hence there exists a closed operator  $(\mathcal{A}, \mathcal{D}(\mathcal{A}))$  such that  $R(\lambda) = R(\lambda, \mathcal{A})$ , i.e.,

$$R(\lambda, A) = R(\lambda, \mathcal{A})(I),$$

Let  $C \in D(\mathcal{A})$ , i.e.,  $C = R(\lambda, \mathcal{A})D$  for some  $D \in \mathscr{L}(E)$ . Then

$$\mathcal{A}(C) = \mathcal{A}(R(\lambda, \mathcal{A})D) = \lambda R(\lambda, \mathcal{A})D - D = (\lambda R(\lambda, \mathcal{A}) - I)D$$

$$= (\lambda R(\lambda, A) - (\lambda - A)R(\lambda, A))D = AR(\lambda, A)D = AC.$$

**Lemma 6.4.** Let E be a Banach space and  $(\mathcal{A}, D(\mathcal{A}))$  a generator of a bi-continuous semigroup  $(\mathcal{T}(t))_{t\geq 0}$  on  $\mathscr{L}(E)$  with respect to  $\tau_{sot}$ . The following are equivalent:

- (a)  $D(\mathcal{A})$  is a right-ideal of  $\mathscr{L}(E)$  and  $\mathcal{A}$  is a right  $\mathscr{L}(E)$ -module homomorphism.
- (b) The semigroup  $(\mathcal{T}(t))_{t\geq 0}$  is left implemented, i.e., there exists a  $C_0$ -semigroup  $(S(t))_{t\geq 0}$ such that  $\mathcal{T}(t)C = S(t)C$  for each  $t\geq 0$ .

Under these equivalent conditions, if (B, D(B)) is the generator of the  $C_0$ -semigroup  $(S(t))_{t\geq 0}$ , then  $\mathcal{A}(C) = B_{-1}C$  for each  $C \in \mathscr{L}(E, E_{-1}) = X_{-1}(\mathcal{A})$ .

*Proof.* (b)  $\Rightarrow$  (a) : If  $C \in D(\mathcal{A})$ , then the limit

$$(\mathcal{A}C)(x) := \lim_{t \searrow 0} \frac{\mathcal{T}(t)Cx - Cx}{t},$$

exists for each  $x \in X_0$ . Since  $(\mathcal{T}(t))_{t \geq 0}$  is left implemented we obtain for  $B \in \mathscr{L}(E)$ 

$$(\mathcal{A}(CB))(x) = \lim_{t \to 0} \frac{\mathcal{T}(t)(CB)x - (CB)x}{t} = \lim_{t \to 0} \frac{(\mathcal{T}(t)C)(Bx) - C(Bx)}{t},$$

and we conclude that  $CB \in D(\mathcal{A})$  and  $\mathcal{A}(CB) = \mathcal{A}(C)B$ .

(a)  $\Rightarrow$  (b) : For  $\lambda \in \rho(\mathcal{A}), C \in D(\mathcal{A}), B \in \mathscr{L}(E)$  one has

$$(\lambda - \mathcal{A})(CB) = \lambda CB - \mathcal{A}(C)B = (\lambda C - \mathcal{A}(C))B.$$

Since  $\lambda - A$  is a bijective map we conclude that

$$R(\lambda, \mathcal{A})(DB) = (R(\lambda, \mathcal{A})D)B$$

for each  $D \in \mathscr{L}(E)$ . By the Euler-Formula (see [28, Thm. 4.6] and [52, Chapter II, Sect. 3]) we obtain

$$\mathcal{T}(t)C = \tau_{\text{sot}} \lim_{n \to \infty} \left(\frac{n}{t} R\left(\frac{n}{t}, \mathcal{A}\right)\right)^n C$$

From this we deduce the equality

$$\mathcal{T}(t)(CB)(x) = \left(\tau \lim_{n \to \infty} \left(\frac{n}{t} R\left(\frac{n}{t}, \mathcal{A}\right)\right)^n C\right) B = (\mathcal{T}(t)C)B.$$

Set  $S(t) := \mathcal{T}(t)I$ , and we are done

$$\mathcal{T}(t)C = \mathcal{T}(t)(I \cdot C) = (\mathcal{T}(t)I)C = S(t)C.$$

Finally,  $\mathcal{A}$  is multiplication operator by the generator (B, D(B)) of the semigroup  $(S(t))_{t\geq 0}$  by Lemma 6.3.

**Proposition 6.5.** Let  $(T(t))_{t\geq 0}$  and  $(S(t))_{t\geq 0}$  be  $C_0$ -semigroups on the Banach space E, and let (A, D(A)) denote the generator of  $(T(t))_{t\geq 0}$ . Let  $(\mathcal{U}(t))_{t\geq 0}$  and  $(\mathcal{V}(t))_{t\geq 0}$  be the semigroups

left implemented by  $(T(t))_{t\geq 0}$  and  $(S(t))_{t\geq 0}$ , respectively. Let  $(\mathcal{G}, D(\mathcal{G}))$  be the generator of  $(\mathcal{U}(t))_{t\geq 0}$  and let  $\mathcal{K} : \mathscr{L}(E) \to \mathscr{L}(E, E_{-1}(A))$  be such that  $\mathcal{K} \in \mathcal{S}_{t_0}^{DS,\tau_{sot}}(\mathcal{U})$  and such that  $\mathcal{C} := (\mathcal{G}_{-1} + \mathcal{K})_{|\mathscr{L}(E)}$  (with maximal domain) is the generator of  $(\mathcal{V}(t))_{t\geq 0}$ . Then  $\mathcal{K}$  has the property that

$$\mathcal{K}(CD) = \mathcal{K}(C)D,$$

for each  $C, D \in \mathscr{L}(E)$ .

*Proof.* Since by assumption  $\mathcal{G}$  and  $\mathcal{C} = (\mathcal{G}_{-1} + \mathcal{K})_{|\mathscr{L}(E)}$  both generate implemented semigroups we conclude by Lemma 6.4 that  $\mathcal{G}$ , and hence  $\mathcal{G}_{-1}$ , and  $\mathcal{C}$  are all multiplication operators. One has  $\mathcal{G}_{-1}(C) = A_{-1}C$  for each  $C \in \mathscr{L}(E)$  and there exists an operator  $M : E \to E_{-1}(L)$ such that  $\mathcal{C}(C) = MC$  for each  $C \in D(\mathcal{C})$ . We conclude that

$$\mathcal{K}(C) = MC - A_{-1}C$$

for each  $C \in D(\mathcal{C})$ . Since  $(\mathcal{C}, D(\mathcal{C}))$  is bi-dense in  $\mathscr{L}(E)$ , for each  $C \in \mathscr{L}(E)$ , there exists a sequence of operators  $(C_n)_{n \in \mathbb{N}}$  in  $D(\mathcal{C})$  such that  $\sup_{n \in \mathbb{N}} ||C_n|| < \infty$  and

$$C_n x \to C x,$$

for each  $x \in E$ . The continuity of  $\mathcal{K}$  and  $\mathcal{G}_{-1}$  yields

$$\mathcal{K}(C_n) \stackrel{\tau_{\text{sot}}}{\to} \mathcal{K}(C),$$
$$\mathcal{G}(C_n) \stackrel{\tau_{\text{sot}}}{\to} \mathcal{G}(C)$$

with convergence in  $\mathscr{L}_{\text{sot}}(E, E_{-1}(A))$ . Therefore, for each  $x \in E$  the sequence  $(MC_n x)_{n \in \mathbb{N}}$  is Cauchy in  $E_{-1}(A)$  and we can define

$$Lx := \lim_{n \to \infty} MC_n x, \quad x \in E.$$

By construction we obtain  $L \in \mathscr{L}(E, E_{-1}(A))$  and  $\mathcal{C}(C) = LC$  for each  $C \in \mathscr{L}(E)$  and therefore

$$\mathcal{K}(C) = A_{-1}C + LC,$$

for  $C \in D(\mathcal{C})$ . Now we define  $B := L + A_{-1}$  as an operator in  $\mathscr{L}(E, E_{-1}(A))$  and conclude that

$$\mathcal{K}(C) = BC$$

for each  $C \in \mathscr{L}(E)$  and that was to be proven.

#### 6.3.2 A one-to-one correspondence

We are now prepared to relate Desch–Schappacher perturbations of the implemented semigroup with the perturbations of the underlying  $C_0$ -semigroup. To do so we have to use the class of Desch–Schappacher admissible operators  $S_{t_0}^{DS}$  for  $C_0$ -semigroups. Recall from [52, Chapter III, Section 3a] the following definitions for a strongly continuous semigroup  $(T(t))_{t\geq 0}$  on a Banach space E. We define

$$\mathcal{S}_{t_0}^{DS}(T) := \left\{ B \in \mathscr{L}(E, E_{-1}) : V_B \in \mathscr{L}\left(\mathcal{C}\left(\left[0, t_0\right], \mathscr{L}_{\text{sot}}(E)\right)\right), \|V_B\| < 1 \right\},\$$

where  $V_B$  denotes the corresponding Volterra operator on E defined by

$$(V_B F)(t) := \int_0^t T_{-1}(t-r)BF(r) \, \mathrm{d}r, \quad F \in \mathcal{C}([0,t_0], E), \ t \in [0,t_0].$$

The following result shows that Desch–Schappacher perturbations of a  $C_0$ -semigroup always give us Desch–Schappacher perturbations of the corresponding implemented semigroup.

**Theorem 6.6.** Let  $(\mathcal{U}(t))_{t\geq 0}$  be the semigroup on  $\mathscr{L}(E)$  left implemented by the  $C_0$ -semigroup  $(T(t))_{t\geq 0}$ . Suppose that  $B \in \mathcal{S}_{t_0}^{DS}$  and let  $(S(t))_{t\geq 0}$  be the perturbed  $C_0$ -semigroup. Define the operator  $\mathcal{K} : \mathscr{L}(E) \to \mathscr{L}(E, E_{-1})$  by

$$\mathcal{K}S := BS, \quad S \in \mathscr{L}(E).$$

Then  $\mathcal{K} \in \mathcal{S}_{t_0}^{DS,\tau_{\text{sot}}}$  and the perturbed semigroup  $(\mathcal{V}(t))_{t\geq 0}$  is left implemented by  $(S(t))_{t\geq 0}$ . Proof. First of all we show that  $V_{\mathcal{K}}F(t)C \in \mathscr{L}(E)$  for  $F \in \mathfrak{X}_{t_0}, t \in [0, t_0]$  and  $C \in \mathscr{L}(E)$ . Define  $f \in C([0, t_0], \mathscr{L}_{\text{sot}}(E))$  by f(r) := F(r)C and observe

$$(V_{\mathcal{K}}F)(t)Cx = \int_0^t \mathcal{U}_{-1}(t-r)\mathcal{K}F(r)Cx \, \mathrm{d}r = \int_0^t T_{-1}(t-r)Bf(r)x \, \mathrm{d}r.$$

Since by assumption  $B \in \mathcal{S}_{t_0}^{DS}$ , we obtain  $(V_{\mathcal{K}}F)(t)Cx \in E$ . The following estimate will be crucial for what follows

$$\|(V_{\mathcal{K}}F)(t)Cx\| = \left\|\int_{0}^{t} \mathcal{U}_{-1}(t-r)\mathcal{K}F(r)Cx \, \mathrm{d}r\right\| = \left\|\int_{0}^{t} T_{-1}(t-r)BF(r)Cx\right\|$$
$$= \|(V_{B}f)(t)x\| \le \|V_{B}\| \cdot \|f\| \cdot \|Cx\| \le \|V_{B}\| \cdot \|f\| \cdot \|C\| \cdot \|x\|.$$

This estimate shows that  $(V_{\mathcal{K}}F)(t)C \in \mathscr{L}(E)$ . Moreover, we directly see that  $\operatorname{Ran}(V_{\mathcal{K}}) \subseteq \mathfrak{X}_{t_0}$ , since  $\tau_{\operatorname{sot}}$ -strong continuity, norm boundedness and bi-equicontinuity of  $V_{\mathcal{K}}F$  follow also from the previous estimate. Also the fact that  $||V_{\mathcal{K}}|| < 1$  is immediate, due to the assumption that  $B \in \mathcal{S}_{t_0}^{DS}$ . Finally, we show that  $(\mathcal{G}_{-1} + \mathcal{K})_{|\mathscr{L}(E)}$  generates the semigroup left implemented by  $(S(t))_{t\geq 0}$ . For this notice that for sufficiently large  $\lambda > 0$  we have

$$R(\lambda, (A_{-1} + B)_E)Cx = \int_0^\infty e^{-\lambda t} S(t)Cx \, dt = \int_0^\infty e^{-\lambda t} \mathcal{V}(t)Cx \, dt$$
$$= R(\lambda, (\mathcal{G}_{-1} + \mathcal{K})_{|\mathscr{L}(E)})Cx,$$

for all  $x \in E$  and  $C \in \mathscr{L}(E)$ . Whence we conclude that  $(\mathcal{G}_{-1} + \mathcal{K})_{|\mathscr{L}(E)}$  generates the semigroup left implemented by  $(S(t))_{t\geq 0}$ .

The converse of Theorem 6.6 is also true.

**Theorem 6.7.** Let  $(\mathcal{U}(t))_{t\geq 0}$  and  $(\mathcal{V}(t))_{t\geq 0}$  be two semigroups on  $\mathscr{L}(E)$ , left implemented by the  $C_0$ -semigroups  $(T(t))_{t\geq 0}$  and  $(S(t))_{t\geq 0}$ , respectively. Let (A, D(A)) be the generator of  $(T(t))_{t\geq 0}$  and let  $\mathcal{K} \in \mathcal{S}_{t_0}^{DS,\tau_{sot}}(\mathcal{U})$  be such that  $(\mathcal{V}(t))_{t\geq 0}$  is the corresponding perturbed semigroup. Define  $B \in \mathscr{L}(E, E_{-1})$  by

$$Bx := (\mathcal{K}I)x, \quad x \in E.$$

Then  $B \in \mathcal{S}_{t_0}^{DS}(T)$  and  $(A_{-1} + B)_{|E}$  generates  $(S(t))_{t \geq 0}$ .

Proof. Let  $f \in C([0, t_0], \mathscr{L}_{sot}(E))$  and  $x \in E$ . We observe that by Lemma 6.3 one has that  $(\mathcal{K}I)f(r) = \mathcal{K}(If(r)) = \mathcal{K}f(r)$  for each  $r \in [0, t_0]$ . For  $f \in C([0, t_0], \mathscr{L}_{sot}(E))$  we define  $F \in \mathfrak{X}_{t_0}$  by  $F(r) := M_{f(r)}$ , the multiplication with f(r), i.e., F(r)C = f(r)C for each  $C \in \mathscr{L}(E)$ . The following computation is crucial for the proof:

$$V_B f(t)x = \int_0^t T_{-1}(t-r)Bf(r)x \, dr = \int_0^t T_{-1}(t-r)(\mathcal{K}I)f(r)x \, dr$$
  
=  $\int_0^t \mathcal{U}_{-1}(t-r)\mathcal{K}f(r)x \, dr = \int_0^t \mathcal{U}_{-1}(t-r)\mathcal{K}F(r)Ix \, dr$   
=  $(V_{\mathcal{K}}F)(t)Ix.$ 

From this and from the assumption that  $\mathcal{K} \in \mathcal{S}_{t_0}^{DS, \tau_{\text{sot}}}$ , we conclude that  $B \in \mathcal{S}_{t_0}^{DS}$ . Moreover we have

$$S(t)x = \mathcal{V}(t)Ix = \mathcal{U}(t)Ix + \int_0^t \mathcal{U}_{-1}(t-r)\mathcal{K}\mathcal{V}(r)Ix \, \mathrm{d}r = T(t)x + \int_0^t T_{-1}(t-r)BS(r)x \, \mathrm{d}r,$$

for each  $x \in E$ . This yields that  $(A_{-1} + B)_{|E}$  generates  $(S(t))_{t \ge 0}$ .

Summarizing Theorems 6.7 and 6.6 we can state the following.

**Corollary 6.8.** Let  $(\mathcal{U}(t))_{t\geq 0}$  and  $(\mathcal{V}(t))_{t\geq 0}$  be two semigroups on  $\mathscr{L}(E)$  left implemented by the  $C_0$ -semigroups  $(T(t))_{t\geq 0}$  and  $(S(t))_{t\geq 0}$  on E, respectively. Let us denote the generators of  $(\mathcal{U}(t))_{t\geq 0}$  and  $(T(t))_{t\geq 0}$  by  $(\mathcal{G}, D(\mathcal{G}))$  and (A, D(A)), respectively. The following are equivalent:

- (i) There exists  $\mathcal{K} \in \mathcal{S}_{t_0}^{DS,\tau}(\mathcal{U})$  such that  $(\mathcal{V}(t))_{t\geq 0}$  is generated by  $(\mathcal{G}_{-1}+\mathcal{K})_{|\mathscr{L}(E)}$ .
- (ii) There exists  $B \in \mathcal{S}_{t_0}^{DS}(T)$  such that  $(S(t))_{t \ge 0}$  is generated by  $(A_{-1} + B)_{|E}$ .

**Remark 6.9.** Notice that not every Desch–Schappacher perturbation of a implemented semigroup gives again a implemented semigroup. To see this let  $(\mathcal{G}, D(\mathcal{G}))$  be the generator of the left implemented semigroup  $(\mathcal{U}(t))_{t\geq 0}$  and  $\Phi \in (\mathscr{L}(E), \tau_{sot})'$ . Define, as above, an operator  $\mathcal{K}: \mathscr{L}(E) \to \mathscr{L}(E, E_{-1})$  by

$$\mathcal{K}(C) := \Phi(C)\mathcal{G}_{-1}(I), \quad C \in \mathscr{L}(E).$$

Such an operator  $\mathcal{K}$  is not multiplicative if  $\Phi \neq 0$ .

#### 6.3.3 Comparisons

Now we relate comparison properties of the implemented semigroup and properties of the underlying  $C_0$ -semigroup. First of all, for  $B \in \mathscr{L}(E)$  we define the multiplication operator  $M_B \in \mathscr{L}(\mathscr{L}(E), \mathscr{L}(E))$  by  $M_B S := BS$ . Then one has  $||M_B|| = ||B||$ . By taking B := T(t) - S(t) for t > 0 we directly obtain the following result.

**Lemma 6.10.** Let  $(\mathcal{U}(t))_{t\geq 0}$  and  $(\mathcal{V}(t))_{t\geq 0}$  be two semigroups on  $\mathscr{L}(E)$  left implemented by the  $C_0$ -semigroups  $(T(t))_{t\geq 0}$  and  $(S(t))_{t\geq 0}$ , respectively. Then the following are equivalent:

(a) There exists  $M \ge 0$  such that  $\|\mathcal{U}(t) - \mathcal{V}(t)\| \le Mt$  for each  $t \in [0, 1]$ .

(b) There exists  $M \ge 0$  such that  $||T(t) - S(t)|| \le Mt$  for each  $t \in [0, 1]$ .

Recall from [28, Prop. 6.1] that the Favard spaces of the implemented semigroup and of the underlying semigroup for  $\alpha \in [0, 1]$  are connected by

$$F_{\alpha}(\mathcal{U}) = \mathscr{L}(E, F_{\alpha}(T)). \tag{6.3.1}$$

This yields to the following result.

**Lemma 6.11.** For  $B \in \mathscr{L}(E, E_{-1})$  we define  $\mathcal{K} : \mathscr{L}(E) \to \mathscr{L}(E, E_{-1})$  by  $\mathcal{K}S := BS$ . Then  $\operatorname{Ran}(\mathcal{K}) \subseteq F_{\alpha}(\mathcal{U})$  if and only if  $\operatorname{Ran}(B) \subseteq F_{\alpha}(T)$ .

By [6] and [28] the extrapolated implemented semigroup is defined by

$$\mathcal{U}_{-1}(t)S = T_{-1}(t)S, \quad S \in \overline{\mathscr{L}(E)}^{\mathscr{L}_{\text{sot}}(E,E_{-1})} = \mathscr{L}(E,E_{-1}).$$

This gives

$$F_0(\mathcal{U}) = F_1(\mathcal{U}_{-1}) = \mathscr{L}(E, F_1(T_{-1})) = \mathscr{L}(E, F_0(T)).$$

**Proposition 6.12.** Let  $(\mathcal{U}(t))_{t\geq 0}$  and  $(\mathcal{V}(t))_{t\geq 0}$  be two semigroups on  $\mathscr{L}(E)$  left implemented by  $(T(t))_{t\geq 0}$  and  $(S(t))_{t\geq 0}$ , respectively. Furthermore let  $(\mathcal{G}, D(\mathcal{G}))$  denote the generator of  $(\mathcal{U}(t))_{t\geq 0}$ . Suppose that there exists  $M \geq 0$  such that

$$\|\mathcal{U}(t) - \mathcal{V}(t)\| \le Mt$$

for each  $t \in [0,1]$ . Then there exists  $\mathcal{K} \in \mathcal{S}_{t_0}^{DS,\tau}$  with  $\operatorname{Ran}(\mathcal{K}) \subseteq F_0(\mathcal{G})$ .

Proof. Since  $\|\mathcal{U}(t) - \mathcal{V}(t)\| \leq Mt$  for each  $t \in [0, 1]$  we can use Lemma 6.10 to conclude that  $\|T(t) - S(t)\| \leq Mt$  for each  $t \in [0, 1]$ . If (A, D(A)) denotes the generator of  $(T(t))_{t\geq 0}$ , then by [52, Chapter III, Thm. 3.9] we find  $B \in \mathscr{L}(E, E_{-1})$  such that  $B \in \mathcal{S}_{t_0}^{DS}$  and  $\operatorname{Ran}(B) \subseteq F_0(A)$ . As in Theorem 6.6 this gives rise to an multiplication operator  $\mathcal{K} : \mathscr{L}(E) \to \mathscr{L}(E, E_{-1})$  defined by

$$\mathcal{K}S := BS, \quad S \in \mathscr{L}(E).$$

By Lemma 6.11 we conclude that  $\operatorname{Ran}(\mathcal{K}) \subseteq F_0(\mathcal{G})$ . It remains to show that  $(\mathcal{G}_{-1} + \mathcal{K})_{|\mathscr{L}(E)}$ generates  $(\mathcal{V}(t))_{t\geq 0}$ . But, by [52, Chapter III, Thm. 3.9],  $(A_{-1}+B)_{|E}$  generates  $(S(t))_{t\geq 0}$ .  $\Box$ 

Combining Propositions 6.12, 4.5 and [52, Chapter III, Thm. 3.9] we obtain the following theorem.

**Theorem 6.13.** Let  $(\mathcal{U}(t))_{t\geq 0}$  and  $(\mathcal{V}(t))_{t\geq 0}$  be two semigroups on  $\mathscr{L}(E)$  left implemented by  $(T(t))_{t\geq 0}$  and  $(S(t))_{t\geq 0}$ , respectively. Denote by  $(\mathcal{G}, D(\mathcal{G}))$  the generator of  $(\mathcal{U}(t))_{t\geq 0}$  and by (A, D(A)) the generator of  $(T(t))_{t\geq 0}$ . If  $\mathcal{K} \in \mathcal{S}_{t_0}^{DS,\tau}(\mathcal{U})$  such that  $\operatorname{Ran}(\mathcal{K}) \subseteq F_0(\mathcal{G})$ , then there exists  $B \in \mathcal{S}_{t_0}^{DS}(T)$  with  $\operatorname{Ran}(B) \subseteq F_0(A)$  such that  $\mathcal{K}S = BS$  for each  $S \in \mathscr{L}(E)$ .

*Proof.* By Proposition 4.5 we find  $M \ge 0$  such that  $\|\mathcal{U}(t) - \mathcal{V}(t)\| \le Mt$  for each  $t \in [0, 1]$ . Following the proof of Proposition 6.12 there exists  $B \in \mathcal{S}_{t_0}^{DS}$  such that  $\operatorname{Ran}(B) \subseteq F_0(A)$ .  $\Box$ 

From this we can deduce the following equivalence.

**Theorem 6.14.** Let  $(\mathcal{U}(t))_{t\geq 0}$  and  $(\mathcal{V}(t))_{t\geq 0}$  be two semigroups on  $\mathscr{L}(E)$  left implemented by the  $C_0$ -semigroups  $(T(t))_{t\geq 0}$  and  $(S(t))_{t\geq 0}$  on E, respectively. Let us denote the generators of  $(\mathcal{U}(t))_{t\geq 0}$  and  $(T(t))_{t\geq 0}$  by  $(\mathcal{G}, D(\mathcal{G}))$  and (A, D(A)), respectively. The following are equivalent:

- (a) There exists  $\mathcal{K} \in \mathcal{S}_{t_0}^{DS,\tau}(\mathcal{U})$  such that  $\operatorname{Ran}(\mathcal{K}) \subseteq F_0(\mathcal{G})$  and such that  $(\mathcal{V}(t))_{t \ge 0}$  is generated by  $(\mathcal{G}_{-1} + \mathcal{K})_{|\mathscr{L}(E)}$ .
- (b) There exists  $B \in \mathcal{S}_{t_0}^{DS}(T)$  such that  $\operatorname{Ran}(B) \subseteq F_0(A)$  and such that  $(S(t))_{t \ge 0}$  is generated by  $(A_{-1} + B)_{|E}$ .

## § 6.4 Notes

The results mentioned in this chapter are taken from [28] and [27]. We remark, that we only considered the left implemented case. The right implemented semigroup as well as the two-sided implemented semigroup are not studied in such detail. Here we refer for extrapolation results to the work of J. Alber [5, 6]. The question of implementation, as it was already mentioned above by Ludwig's Theorem and the Wigner symmetry, is still interesting. Especially, the results from Section 6.3.2 and Section 6.3.3 assume that the perturbed semigroups are implemented as well. The question is then, under which assumptions on the perturbation operator the perturbed implemented semigroup is again implemented. One indication could in point of fact be the aspects of quantum mechanics.

## Chapter 7

## Flows on Networks

## Introduction

Consider a very large network, whose actual size may not be known but some of its structural properties are understood well. One way to model this situation is to consider an infinite graph with combinatorially reasonable assumptions based on the a priori knowledge about the structure of the network. Along the edges of the network some transport processes take place that are coupled in the vertices in which the edges meet. This means that we consider each edge as an interval and describe functions on it, that is, we consider a *metric graph*. Systems of partial differential equations on a metric graph are also known as *quantum graphs*. The transport processes (or *flows*) on the edges are given by partial differential equations of the form  $\frac{\partial}{\partial t}u_j(t,x) = c_j\frac{\partial}{\partial x}u_j(t,x)$  and are interlinked in the common nodes via some prescribed transmission conditions.

Such a problem was considered by Dorn et al. [41, 43, 42] on the state space  $L^1([0, 1], \ell^1)$  applying the theory of strongly continuous operator semigroups. A semigroup approach to flows in finite metric graphs was first presented by Kramar and Sikolya [76] and further used in [51, 42, 22, 15, 21] while transport processes in infinite networks were also studied in [16, 18]. However, all these results were obtained in the L<sup>1</sup>-setting. By considering problems in infinite graphs, the flow problem in the L<sup> $\infty$ </sup>-setting is interesting for applications as well. J. von Below and J.A. Lubary [120, 121], for example, study eigenvalues of the Laplacian on infinite metric graphs with an L<sup> $\infty$ </sup>-setting. To the best of our knowledge, transport equations on infinite metric graphs with an L<sup> $\infty$ </sup>-state space have not yet been studied. We consider this problem on the state space L<sup> $\infty$ </sup> ([0, 1],  $\ell^1$ ) where the obtained operator semigroup is not strongly continuous. To tackle this we make use of the theory of bi-continuous semigroups.

This chapter is organised as follows. Section 7.1 is a preliminary section where we introduce some notions for networks and metric graphs. In Section 7.2 we present the flow problem for an infinite metric graph. We first prove the well-posedness in the case when all flow velocities  $c_j$  equal 1. Next, we generalise this result to the case with rationally dependent velocities satisfying a finiteness condition. Finally, we show that the general problem is well-posed. Actually this will be done only on a finite metric graph due to technical assumptions.

## § 7.1 Preliminaries

In order to talk about finite and infinite networks we use the notation used in [76] and [41]. A network is modelled with an *infinite directed graph* G = (V, E) with a set of vertices  $V = \{v_i \mid i \in I\}$  and a set of directed edges  $E = \{e_j \mid j \in J\} \subseteq V \times V$  for some countable sets  $I, J \subseteq \mathbb{N}$ . For a directed edge  $e = (v_i, v_k)$  we call  $v_i$  the *tail* and  $v_k$  the *head* of e. Further, the edge e is an outgoing edge of the vertex  $v_i$  and an *incoming edge* for the vertex  $v_k$ . We assume that the graph G is simple, i.e., there are no loops or multiple edges. This means in particular, that there are no edges of the form  $e = (v_i, v_i), i \in I$  (i.e., the tail and the head of the edge coincide and so an edge connects a vertex with itself) and no several edges connecting two vertices in the same direction



Figure 7.1: A loop (a) and a graph with multiple edges (b), cf. [40, Fig. 2.2]

Moreover, G is assumed to be *locally finite*, i.e., each vertex has only finitely many outgoing edges. Furthermore, the graph G is *weighted*, that is equipped with some weights  $0 \le w_{ij} \le 1$  such that

$$\sum_{j \in I} w_{ij} = 1 \text{ for all } i \in I.$$
(7.1.1)

The structure of a graph can be described by its incidence or its adjacency matrix. The outgoing incidence matrix  $\Phi^- = (\Phi_{ij}^-)$  is defined by

$$\Phi_{ij}^{-} := \begin{cases} 1 & \text{if } \mathbf{v}_i \xrightarrow{\mathbf{e}_j}, \\ 0 & \text{otherwise,} \end{cases}$$
(7.1.2)

whereas the weighted outgoing incidence matrix  $\Phi_{\omega}^{-} = (\Phi_{\omega,ij}^{-})$  is defined by

$$\Phi_{\omega,ij}^{-} := \begin{cases} w_{ij} & \text{if } \mathbf{v}_i \xrightarrow{\mathbf{e}_j}, \\ 0 & \text{otherwise.} \end{cases}$$
(7.1.3)

By  $v_i \xrightarrow{e_j}$  we mean that the vertex  $v_i$  is the tail of the edge  $e_j$ . The *incoming incidence* matrix  $\Phi^+ = (\Phi_{ij}^+)$  is defined by

$$\Phi_{ij}^{+} := \begin{cases} 1 & \text{if } \xrightarrow{e_{j}} \mathbf{v}_{i}, \\ 0 & \text{otherwise.} \end{cases}$$
(7.1.4)

Here  $\xrightarrow{e_j} v_i$  means that the vertex  $v_i$  is the head of the edge  $e_j$ . The *incidence matrix*  $\Phi$  of the directed graph G, describing the structure of the network completely, is then defined by  $\Phi := \Phi^+ - \Phi^-$ . There are two other important matrices associated to a general graph and who are needed in what follows. The *transposed adjacency matrix* of the weighted graph G is defined by

$$\mathbb{A}:=\Phi^+\left(\Phi^-_\omega
ight)^+$$
 .

The nonzero entries of A correspond exactly to the edges of the graph, keeping track also of the appropriate weights, cf. [21, p. 280]. In fact, A can be described explicit as

$$\mathbb{A}_{ij} := \begin{cases} w_{jk} & \text{if } \mathbf{v}_j \xrightarrow{\mathbf{e}_k} \mathbf{v}_i, \\ 0 & \text{otherwise.} \end{cases}$$
(7.1.5)

Last but not least we use the so-called weighted (transposed) adjacency matrix of the line graph  $\mathbb{B} = (\mathbb{B}_{ij})$  defined by  $\mathbb{B} := (\Phi_{\omega}^{-})^{\top} \Phi^{+}$ . One can also give an explicit entrywise description as

$$\mathbb{B}_{ij} := \begin{cases} w_{ki} & \text{if } \xrightarrow{\mathbf{e}_j} \mathbf{v}_k \xrightarrow{\mathbf{e}_i}, \\ 0 & \text{otherwise.} \end{cases}$$
(7.1.6)

By (7.1.1), the matrix  $\mathbb{B}$  is column stochastic, i.e., the sum of entries of each column is 1, and defines a bounded positive operator on  $\ell^1 := \ell^1(J)$  with  $r(\mathbb{B}) = ||\mathbb{B}|| = 1$ .

We identify every edge of our graph with the unit interval,  $e_j \equiv [0,1]$  for each  $j \in J$ , and parametrise it contrary to the direction of the flow, if  $c_j > 0$ ,  $j \in J$ , so that it is assumed to have its tail at the endpoint 1 and its head at the endpoint 0. For simplicity we use the notation  $e_j(1)$  and  $e_j(0)$  for the tail and the head, respectively. In this way we obtain a *metric* graph.

## § 7.2 Transport problems on (in)finite metric graphs

We now consider a transport process (or a flow) along the edges of an infinite network, modelled by a metric graph G. The distribution of material along edge  $e_j$  at time  $t \ge 0$  is described by function a  $u_j(x,t)$  for  $x \in [0,1]$ . The material is transported along the edge  $e_j$ with constant velocity  $c_j > 0$ ,  $j \in J$ . We assume that

$$0 < c_{\min} \le c_j \le c_{\max} < \infty \tag{7.2.1}$$

for all  $j \in J$ . Let  $C := \text{diag}(c_j)_{j \in J}$  be a diagonal velocity matrix and define another weighted adjacency matrix of the line graph by

$$\mathbb{B}^C := C^{-1} \mathbb{B} C.$$

In the vertices the material gets redistributed according to some prescribed rules. This is modelled in the boundary conditions by using the adjacency matrix  $\mathbb{B}_C$ . The flow process on G is thus described by the following infinite system of equations

$$\begin{cases} \frac{\partial}{\partial t} u_j(x,t) = c_j \frac{\partial}{\partial x} u_j(x,t), & x \in (0,1), \ t \ge 0, \\ u_j(1,t) = \sum_{k \in J} \mathbb{B}_{jk}^C u_k(0,t), & t \ge 0, \ \text{(boundary conditions)} \\ u_j(x,0) = f_j(x), & x \in (0,1), \end{cases}$$
(7.2.2)

for every  $j \in J$ , where  $f_j(x)$  are the initial distributions along the edges. One can give different interpretations to the weights  $w_{ij}$ , i.e., entries of the matrix  $\mathbb{B}$ , resulting in different transport problems. The two most obvious are the following.

- 1.  $w_{ij}$  is the proportion of the material arriving from edge  $e_j$  leaving on edge  $e_i$ .
- 2.  $w_{ij}$  is the proportion of the material arriving in vertex  $e_i(0) = e_i(1)$  leaving on edge  $e_i$ .

Note, that in both situations (7.1.1) represents a conservation of mass and the assumption on local finiteness of the graph guarantees that all the sums are finite. While the latter situation is the most common one (see e.g. [41, 76, 21]) the first one was considered for finite networks in [22, Sect. 5]. Here, we will not give any particular interpretation and will treat all the cases simultaneously.

**Remark 7.1.** By replacing in (7.2.2) the graph matrix  $\mathbb{B}^C$  with some other matrix, one obtains a more general initial-value problem that does not necessarily consider a process in a physical network. Such a problem from population dynamics was for example studied in [16]. Furthermore, the question when such a general problem can be identified with a corresponding problem on a metric graph was raised in [15].

#### 7.2.1 The simple case: constant velocities

First we assume that all the velocities are the same:  $c_j = 1$  for each  $j \in J$ . In what follows we abbreviate the space  $\ell^1(J)$  by  $\ell^1$ . As the state space we set  $X_0 := L^{\infty}([0,1], \ell^1)$  equipped with the norm

$$\|f\|_{X_0} := \mathop{\mathrm{ess \, sup}}_{s \in [0,1]} \|f(s)\|_{\ell^1}.$$

On the Banach space  $X_0$  we define the operator (A, D(A)) by

$$A := \operatorname{diag}\left(\frac{\mathrm{d}}{\mathrm{d}x}\right),$$

$$D(A) := \left\{ v \in \mathrm{W}^{1,\infty}\left([0,1],\ell^1\right) \mid v(1) = \mathbb{B}v(0) \right\}.$$
(7.2.3)

Now we consider the corresponding abstract Cauchy problem on  $X_0$  corresponding to the operator (A, D(A)) defined in (7.2.3)

$$\begin{cases} \dot{u}(t) = Au(t), & t \ge 0, \\ u(0) = (f_j)_{j \in J}. \end{cases}$$
(7.2.4)

For the sake of completeness we prove that (7.2.4) is equivalent to the flow problem (7.2.2) in case when all the velocities equal 1.

To see this we show that the domain condition in (7.2.3) is equivalent to the boundary condition in (7.2.2). For  $g \in D(A)$  we obtain for its *j*-th component the following equality

$$g_j(1) = \mathbb{B}_j g(0) = \left(\Phi_\omega^-\right)_j^\top \Phi^+ g(0)$$

By the structure of  $(\Phi_{\omega}^{-})^{\top}$  one obtains

$$\Phi_{ij}^{-}g_{j}(1) = w_{ij} \sum_{k \in J}^{m} \Phi_{ik}^{+}g_{k}(0)$$

For the converse we start by using the boundary condition in (7.2.2). Moreover, we recall from [21, Prop. 18.2] that the *j*-th row of  $(\Phi_{\omega}^{-})^{\top}$  corresponds to the edge  $e_j$  and has exactly one nonzero element, say  $w_{ij}$ , corresponding to the starting vertex  $v_i$  for this edge. That is to say that  $\Phi_{ij}^{-} = 1$ . Hence we obtain

$$\begin{aligned} (\mathbb{B}g(0))_j &= \left(\Phi_{\omega}^{-}\right)_j^{\top} \Phi^+ g(0) \\ &= w_{ij} \sum_{k \in J}^m \Phi_{ik}^+ g_k(0) \\ &= \Phi_{ij}^- g_j(1) = g_j(1) \end{aligned}$$

This shows indeed that the two problems (7.2.2) and (7.2.4) are equivalent.

On the state space  $L^1([0,1], \ell^1)$  this was considered by B. Dorn [41] where an explicit formula for the solution semigroup in terms of a shift and matrix  $\mathbb{B}$  was derived. Dorn made use of the left-translation semigroup  $(T_l(t))_{t\geq 0}$ , which is not strongly continuous on  $X_0$ . However, by using duality arguments we will show that it is a bi-continuous semigroup on  $X_0$ . First note that the left-translation semigroup on  $X_0 = L^1([0,1], c_0)' = L^{\infty}([0,1], \ell^1)$  is the adjoint semigroup of the right-translation semigroup on  $L^1([0,1], c_0)$ , see [52, Chapter I, Example 5.14]. Here  $c_0$  is the space of all sequences converging to 0, which is in fact a Banach space equipped with the supremum-norm.

**Lemma 7.2.** The right-translation semigroup  $(T_r(t))_{t>0}$ , defined by

$$T_{\mathbf{r}}(t)f(s) := \begin{cases} f(s-t), & s-t \ge 0, \\ 0, & s-t < 0, \end{cases}$$

for  $f \in L^{1}([0,1], c_{0}), t \geq 0$  and  $s \in [0,1]$ , is strongly continuous on  $L^{1}([0,1], c_{0})$ .

*Proof.* Let  $\mathbf{x} = (x_n)_{n \in \mathbb{N}} \in c_0$  and let  $f := \mathbf{x} \cdot \mathbf{1}_{\Omega}$  for a measurable subset  $\Omega \subseteq [0, 1]$ . We first show that  $T_r(t)f \to f$  with respect to the norm on  $L^1([0, 1], c_0)$  as  $t \to 0$ :

$$\begin{split} &\int_0^1 \|T_{\mathbf{r}}(t)f(s) - f(s)\|_{\mathbf{c}_0} \,\mathrm{d}s = \int_0^1 \left\|f(s-t)\mathbf{1}_{[s-t\geq 0]} - f(s)\right\|_{\mathbf{c}_0} \,\mathrm{d}s \\ &= \|\mathbf{x}\|_{\mathbf{c}_0} \cdot \int_0^1 |\mathbf{1}_{\Omega}(s-t) - \mathbf{1}_{\Omega}(s)| \,\mathrm{d}s = \|\mathbf{x}\|_{\mathbf{c}_0} \cdot \int_0^1 \left|\mathbf{1}_{(\Omega+t)\bigtriangleup\Omega}(s)\right| \,\mathrm{d}s \\ &= \|\mathbf{x}\|_{\mathbf{c}_0} \cdot \lambda^1 \left((\Omega+t)\bigtriangleup\Omega\right) \to 0 \text{ as } t \to 0, \end{split}$$

where  $\lambda^1$  is the one-dimensional Lebesgue measure on [0, 1] and  $\Delta$  the symmetric difference of

sets defined by  $A\Delta B := (A \cup B) \setminus (A \cap B)$ ,  $A, B \subseteq [0, 1]$ . Since every function  $f \in L^1([0, 1], c_0)$ is an increasing limit of linear combinations of functions of the form  $\mathbf{x} \cdot \mathbf{1}_{\Omega}$  for some  $\mathbf{x} \in c_0$  and measurable set  $\Omega \subseteq [0, 1]$ , the vector-valued version of Beppo–Levi's monotone convergence theorem yields the result.

- **Remark 7.3.** (i) The Beppo-Levi theorem we used in the preceding proof is by [117, Prop. 2.6] only applicable in our case owing to the fact that  $c_0$  is a Banach lattice with  $\sigma$ -order continuous norm meaning that if  $(x_n)_{n\in\mathbb{N}}$  a decreasing sequence with respect to the ordering on  $c_0$  such that the infimum is 0, then  $(x_n)_{n\in\mathbb{N}}$  norm-converges to 0. For general Banach spaces X there is no monotone convergence theorem on  $L^1([0,1], X)$ .
  - (ii) Observe that the classical proof of strong continuity, as it can for example be found in [21, Ex. 9.11], uses a density argument instead of the monotone convergence theorem. Especially, the strong continuity does not depend on the space  $c_0$ , hence one could generalise this result to strongly continuous right-translation semigroups on  $L^1([0, 1], Y)$  and even to  $L^p([0, 1], Y)$  for an arbitrary Banach space Y and  $1 \le p < \infty$ .

**Lemma 7.4.** The left-translation semigroup  $(T_{l}(t))_{t\geq 0}$ , defined by

$$T_{\mathbf{l}}(t)f(s) := \begin{cases} f(s+t), & s+t \le 1, \\ 0, & s+t > 1, \end{cases}$$

for  $t \ge 0$  and  $s \in [0,1]$ , is bi-continuous on  $L^{\infty}([0,1], \ell^1)$  with respect to the weak\*-topology.

*Proof.* This result follows directly from the results in Chapter 1 since  $(T_l(t))_{t\geq 0}$  is the adjoint semigroup of the strongly continuous semigroup  $(T_r(t))_{t\geq 0}$ , cf. [52, Chapter II, Sect. 3.6, Exam. (i)], hence we can apply the result from Section 1.2.2, i.e., the adjoint semigroup is bi-continuous with respect to the weak\*-topology.

We now use the formula for the semigroup which was derived by B. Dorn for infinite networks [41] and show that it yields a bi-continuous semigroup on  $L^{\infty}([0,1], \ell^1)$ . For that we have to check all the assertions from Definition 1.3 which we do in several steps.

**Lemma 7.5.** The semigroup  $(T(t))_{t\geq 0}$  on  $X_0 = L^{\infty}([0,1], \ell^1)$ , defined by

$$(T(t)f)(s) = \mathbb{B}^n f(t+s-n), \quad n \le t+s < n+1, \ f \in X_0, \ n \in \mathbb{N}_0,$$
(7.2.5)

is strongly continuous with respect to the weak\*-topology.

*Proof.* The semigroup property is easy to verify, cf. [40, Prop. 1.2.1]. Observe that for any  $f \in X_0, g \in L^1([0,1], c_0)$ , and  $t \in (0,1]$  we have

$$\begin{aligned} |\langle T(t)f - f, g\rangle| &= \left| \int_0^1 \langle T(t)f(s) - f(s), g(s)\rangle \,\mathrm{d}s \right| \\ &\leq \left| \int_0^{1-t} \langle f(s+t) - f(s), g(s)\rangle \,\mathrm{d}s \right| + \int_{1-t}^1 |\langle \mathbb{B}f(s+t-1) - f(s), g(s)\rangle| \,\mathrm{d}s \\ &= \left| \int_0^1 \langle T_l(t)f(s) - f(s), g(s)\rangle \,\mathrm{d}s \right| + \int_{1-t}^1 |\langle \mathbb{B}f(s+t-1) - f(s), g(s)\rangle| \,\mathrm{d}s. \end{aligned}$$

Now, notice that the second summand vanishes since  $\lambda^1([1-t,1]) \to 0$  as  $t \to 0$ . Here,  $\lambda^1$  is again the one-dimensional Lebesgue measure on the unit interval [0,1]. By Lemma 7.4, the left-translation semigroup is bi-continuous on  $X_0$ , which means, in particular, that it is strongly continuous with respect to the weak\*-topology and hence the first summand also vanishes as  $t \to 0$ .

**Lemma 7.6.** The semigroup  $(T(t))_{t\geq 0}$ , defined by (7.2.5), is a contraction semigroup on  $X_0$ .

*Proof.* Let  $f \in X_0$  and  $t \ge 0$ . Then there exists  $n \in \mathbb{N}_0$  such that  $n \le t < n+1$ . This means that for  $s \in [0, 1]$  one has  $n \le s+t < n+2$ . By (7.2.5), we can make the following estimate.

$$\begin{aligned} \|T(t)f\|_{X_0} &= \mathop{\mathrm{ess\ sup}}_{s\in[0,1]} \|T(t)f(s)\|_{\ell^1} \\ &\leq \max\left\{ \mathop{\mathrm{ess\ sup}}_{s\in[0,n+1-t)} \|\mathbb{B}^n f(s+t-n)\|_{\ell^1}, \; \mathop{\mathrm{ess\ sup}}_{s\in[n+1-t,1)} \left\|\mathbb{B}^{n+1} f(s+t-n-1)\right\|_{\ell^1} \right\}. \end{aligned}$$

Since  $\|\mathbb{B}^n\| = \|\mathbb{B}\|^n = 1$ , we have

$$\|\mathbb{B}^n f(s+t-n)\|_{\ell^1} \le \|\mathbb{B}^n\| \cdot \|f\|_{X_0} = \|f\|_{X_0}$$

and hence,  $||T(t)f||_{X_0} \le ||f||_{X_0}$ .

**Lemma 7.7.** The semigroup  $(T(t))_{t\geq 0}$ , defined by (7.2.5), is locally bi-equicontinuous with respect to the weak\*-topology on  $X_0 = L^{\infty}([0,1], \ell^1)$ .

Proof. Let  $(f_n)_{n\in\mathbb{N}}$  be a sequence of functions in  $X_0$  that is  $\|\cdot\|_{X_0}$ -bounded and converges to 0 with respect to the weak\*-topology. By Definition 1.3 we need to show that  $(T(t)f_n)_{n\in\mathbb{N}}$  also converges to 0 with respect to the weak\*-topology uniformly for  $t \in [0, t_0]$ . To this end, fix  $t_0 > 0$  and let  $m := \lfloor t_0 \rfloor$ . Moreover, for fixed  $t \in [0, t_0]$  let  $k := \lfloor t \rfloor$ . Then, for this  $t \in [0, t_0]$  we obtain

$$\begin{aligned} |\langle T(t)f_n,g\rangle| &\leq \int_0^1 \left| \langle T(t)f_n(s),g(s)\rangle_{X_0} \right| \mathrm{d}s \\ &\leq \int_0^1 \left| \left\langle \mathbb{B}^k f_n(s+t-k),g(s)\right\rangle_{X_0} \right| \mathrm{d}s \\ &\leq \int_0^1 \left| \left\langle T_l(t-k)f_n(s),\left(\mathbb{B}^k\right)'g(s)\right\rangle_{X_0} \right| \mathrm{d}s. \end{aligned}$$

By taking the maximum for  $0 \le t \le t_0$  and the corresponding maximum for  $0 \le k \le m$  we obtain for  $0 \le s + t \le m + 1$ ,  $s \in [0, 1]$  by (7.2.5) the following:

$$\begin{aligned} |\langle T(t)f_n,g\rangle| &\leq \int_0^1 \left| \langle T(t)f_n(s),g(s)\rangle_{X_0} \right| \mathrm{d}s \\ &\leq \int_0^1 \max_{0\leq k\leq m} \left| \left\langle \mathbb{B}^k f_n(s+t-k),g(s)\right\rangle_{X_0} \right| \mathrm{d}s \\ &\leq \int_0^1 \max_{0\leq k\leq m} \left| \left\langle T_l(t-k)f_n(s),\left(\mathbb{B}^k\right)'g(s)\right\rangle_{X_0} \right| \mathrm{d}s, \end{aligned}$$

for each  $g \in L^1([0,1],c_0)$ . Since, by Lemma 7.4, the left-translation semigroup  $(T_l(t))_{t\geq 0}$  is bi-continuous, hence locally bi-equicontinuous,  $|\langle T(t)f_n,g\rangle|$  tends to 0 uniformly on  $[0,t_0]$ . This finishes the proof.

Let us recall here the explicit expression of the resolvent of operator (A, D(A)) defined by (7.2.3) which was obtained in [41, Theorem 18]. This result does not rely on the Banach space and remains the same if we take  $X_0 = L^{\infty}([0,1], \ell^1)$  instead of  $L^1([0,1], \ell^1)$ . To be self-contained we reproduce the proof.

**Proposition 7.8.** For  $\operatorname{Re}(\lambda) > 0$  the resolvent  $R(\lambda, A)$  of the operator (A, D(A)) defined by (7.2.3) is given by

$$(R(\lambda, A)f)(s) := \sum_{k=0}^{\infty} e^{-\lambda k} \int_0^1 e^{-\lambda(t+1-s)} \mathbb{B}^{k+1} f(t) \, \mathrm{d}t + \int_s^1 e^{\lambda(s-t)} f(t) \, \mathrm{d}t, \quad f \in X_0, \ s \in [0,1].$$

*Proof.* Let  $f \in X_0$  and  $g \in D(A)$  such that  $(\lambda - A)g = f$ , or in other words,  $\lambda g - g' = f$ . To solve this differential equation we use the variation of constants formula to obtain

$$g(s) = e^{\lambda s} \cdot d + \int_{s}^{1} e^{\lambda(s-t)} f(t) \, dt, \quad s \in [0,1], \qquad (7.2.6)$$

for some  $d \in \ell^1$ . By the assumption that  $g \in D(A)$  we have  $g(1) = \mathbb{B}g(0)$  which yields,

$$d \cdot e^{\lambda} = \mathbb{B}d + \mathbb{B}\int_0^1 e^{-\lambda t} f(t) dt,$$

whence

$$(I - e^{-\lambda} \mathbb{B})d = \mathbb{B} \int_0^1 e^{-\lambda(t+1)} f(t) dt.$$

For  $\lambda \in \mathbb{C}$  with  $\operatorname{Re}(\lambda) > 0$  we obtain from the fact that  $r(\mathbb{B}) = 1$  that  $r\left(e^{-\lambda}\mathbb{B}\right) < 1$ . By applying the Neumann series one gets

$$d = \sum_{k=1}^{\infty} \left( e^{-\lambda} \mathbb{B} \right)^k \mathbb{B} \int_0^1 e^{-\lambda(t+1)} f(t) \, \mathrm{d}t.$$

Combining this with (7.2.6) yields the desired formula.

We are now in the state prove the first generation theorem.

**Theorem 7.9.** The operator (A, D(A)), defined in (7.2.3), generates a bi-continuous contraction semigroup  $(T(t))_{t\geq 0}$  on  $X_0$  with respect to the weak<sup>\*</sup>-topology. This semigroup is given by (7.2.5), i.e.,

$$(T(t)f)(s) = \mathbb{B}^n f(t+s-n), \quad n \le t+s < n+1, \ f \in X_0, \ n \in \mathbb{N}_0.$$

*Proof.* By Lemmas 7.5, 7.6, and 7.7, semigroup  $(T(t))_{t\geq 0}$  defined by (7.2.5) is a bi-continuous semigroup with respect to the weak\*-topology. It remains to show that (A, D(A)), given in (7.2.3), is the generator of this semigroup. Let (B, D(B)) be the generator of  $(T(t))_{t\geq 0}$ . By (1.3.2), the resolvent of B is the Laplace transform of the semigroup  $(T(t))_{t\geq 0}$ , that is, for

 $\lambda > \omega_0(T)$  we have

$$(R(\lambda, B)f)(s) = \int_0^\infty \left( e^{-\lambda t} T(t) f \right)(s) dt$$
  
=  $\int_0^{1-s} e^{-\lambda t} f(t+s) dt + \sum_{n=1}^\infty \int_{n-s}^{n-s+1} e^{-\lambda t} \mathbb{B}^n f(t+s-n) dt$   
=  $\int_s^1 e^{-\lambda(\xi-s)} f(\xi) d\xi + \sum_{n=1}^\infty \int_0^1 e^{-\lambda(\xi-s+n)} \mathbb{B}^n f(\xi) d\xi.$ 

To conclude the proof we observe that D(A) is bi-dense in  $X_0$ . To see this, we refer to Theorem 7.15 below, where we show this for the more general case (even if we there consider finite networks, the idea of the proof is the same). Now, by Proposition 7.8, the resolvent operators  $R(\lambda, A)$  and  $R(\lambda, C)$ , coincide on the bi-dense set D(A), so we conclude that C = A.

**Corollary 7.10.** If all  $c_j = 1$ ,  $j \in J$ , the flow problem (7.2.2) is well-posed on  $X_0 = L^{\infty}([0,1], \ell^1)$ .

**Remark 7.11.** All the obtained results also hold for finite networks. If G = (V, E) is a finite network with  $|E| = m < \infty$ , we have  $\ell^1(\{1, \ldots, m\}) \cong \mathbb{C}^m$ , hence we consider the semigroups on the space  $X_0 = L^{\infty}([0, 1], \mathbb{C}^m)$ .

#### 7.2.2 The rationally dependent case

We now consider the case when the velocities  $c_j$  appearing in (7.2.2) are not all equal to 1 and define on  $X_0 := L^{\infty}([0,1], \ell^1)$  the operator

$$A_C := \operatorname{diag}\left(c_j \cdot \frac{\mathrm{d}}{\mathrm{d}x}\right),$$
  
$$D(A_C) := \left\{f \in \mathrm{W}^{1,\infty}\left([0,1],\ell^1\right) \mid f(1) = \mathbb{B}^C f(0)\right\}.$$
  
(7.2.7)

We assume, however, that the velocities are linearly dependent over  $\mathbb{Q}$ :  $\frac{c_i}{c_j} \in \mathbb{Q}$  for all  $i, j \in J$ , with a finite common multiplier, that is,

there exists 
$$0 < c \in \mathbb{R}$$
 such that  $\ell_j := \frac{c}{c_j} \in \mathbb{N}$  for all  $j \in J$ . (7.2.8)

This enables us to use the procedure that was introduced in the proof of [76, Thm. 4.5] and carried out in detail in [17, Sect. 3]. We construct a new directed graph  $\tilde{G}$  by adding  $\ell_j - 1$ vertices on edge  $e_j$  for all  $j \in J$ . The newly obtained edges inherit the direction of the original edge and are parametrised by unit intervals [0,1]. We can thus consider a new problem on  $\tilde{G}$  with corresponding functions  $\tilde{u}_j$  and velocities  $\tilde{c}_j := c$  for each  $j \in \tilde{J}$ . After appropriately correcting the initial and boundary conditions the new problem is equivalent to the original one. Since all the velocities on the edges of the new graph are equal, we can treat this case by rescaling the velocities to 1 and use the results from Subsection 7.2.1. Moreover, since (7.2.1) and (7.2.8) hold, the procedure described by J. Banasiak and P. Namayanja in [17, Sect. 3] for the finite case can be as applied in the infinite case as well. Hence, we even obtain an isomorphism between the corresponding semigroups. **Proposition 7.12.** Let the assumptions (7.2.1) and (7.2.8) on the velocities  $c_j$  hold. Then operator  $(A_C, D(A_C))$ , defined in (7.2.7), generates a contraction bi-continuous semigroup  $(T_C(t))_{t\geq 0}$  on  $X_0$  with respect to the weak<sup>\*</sup>-topology. Moreover, there exists an isomorphism  $S: X_0 \to X_0$  such that

$$T_C(ct)f = ST(t)S^{-1}f, (7.2.9)$$

where the semigroup  $(T(t))_{t>0}$  is given by (7.2.5)

**Corollary 7.13.** If the assumptions (7.2.1) and (7.2.8) on the velocities  $c_j$  hold, the flow problem (7.2.2) is well-posed on  $X_0 = L^{\infty}([0,1], \ell^1)$ .

#### 7.2.3 The general case for finite networks

We finally consider the case of general  $c_j \in \mathbb{R}$  but restrict ourselves to finite graphs, i.e., we work on the Banach space  $X_0 = L^{\infty}([0,1], \mathbb{C}^m)$ , where *m* denotes the number of edges in the graph. In [21, Cor. 18.15] the Lumer–Phillips generation theorem for positive strongly continuous semigroups is applied to show that the transport problem with general  $c_j \in \mathbb{R}$ is well-posed on  $X_0 = L^1([0,1], \mathbb{C}^m)$ . Since an appropriate variant of a Lumer–Phillips generation theorem for the bi-continuous situation is not known, we proceed differently and use the bi-continuous version of the Trotter–Kato approximation theorem, cf. Theorem 1.22. Let

$$E_{\lambda}(s) := \operatorname{diag}\left(e^{(\lambda/c_j)s}\right), \quad s \in [0,1], \quad \text{and} \quad \mathbb{B}_{\lambda}^C := E_{\lambda}(-1)\mathbb{B}^C, \operatorname{Re}(\lambda) > 0.$$

By using this notation one can write an explicit expression for the resolvent of operator  $A_C$  defined in (7.2.7), cf. [21, Prop. 18.12].

**Lemma 7.14.** For  $\operatorname{Re}(\lambda) > 0$  the resolvent  $R(\lambda, A_C)$  of operator  $A_C$  given in (7.2.7) equals

$$R(\lambda, A_C) = \left( I_{X_0} + E_{\lambda}(\cdot) \left( 1 - \mathbb{B}_{\lambda}^C \right)^{-1} \mathbb{B}_{\lambda}^C \otimes \delta_0 \right) R_{\lambda},$$

where  $\delta_0: L^{\infty}([0,1], \mathbb{C}^m) \to \mathbb{C}^m$  denotes the point evaluation at 0 and

$$(R_{\lambda}f)(s) = \int_{s}^{1} E_{\lambda}(s-t)C^{-1}f(t) \, \mathrm{d}t, \quad s \in [0,1], \ f \in \mathrm{L}^{\infty}\left([0,1], \mathbb{C}^{m}\right).$$

*Proof.* Let  $f \in X_0$  and  $g \in D(A_C)$  such that  $(\lambda - A_C)g = f$ . As a result of the variation of constants formula we obtain

$$g(s) = E_{\lambda}(s)d + \int_{s}^{1} E_{\lambda}(s-t)C^{-1}f(t) \, \mathrm{d}t, \quad s \in [0,1], \qquad (7.2.10)$$

for some  $d \in \mathbb{C}^m$ . The boundary condition of  $D(A_C)$  yields that  $g(0) = \mathbb{B}^C g(1)$  meaning that

$$E_{\lambda}(1)d = \mathbb{B}^{C}d + \mathbb{B}^{C}\int_{0}^{1}E_{\lambda}(-t)C^{-1}f(t) \,\mathrm{d}t.$$

By multiplying the previous equation by  $E_{\lambda}(-1)$  we get

$$\left(I - \mathbb{B}_{\lambda}^{C}\right) d = \mathbb{B}_{\lambda}^{C}(R_{\lambda}f)(0)$$

We observe that  $r\left(\mathbb{B}_{\lambda}^{C}\right) \leq \left\|\mathbb{B}_{\lambda}^{C}\right\| < 1$  (see [21, Sect. 18.3]), hence  $(I - \mathbb{B}_{\lambda}^{C})^{-1}$  exists for  $\operatorname{Re}(\lambda) > 0$  and

$$d = \left(I - \mathbb{B}_{\lambda}^{C}\right)^{-1} \mathbb{B}_{\lambda}^{C}(R_{\lambda}f)(0)$$

Together with (7.2.10) we get the desired expression for the resolvent.

**Theorem 7.15.** The operator  $(A_C, D(A_C))$ , defined in (7.2.7), generates a bi-continuous semigroup  $(T_C(t))_{t\geq 0}$  on  $X_0 = L^{\infty}([0, 1], \mathbb{C}^m)$ .

*Proof.* We first show that operator  $A_C$  is bi-densely defined. Take any  $f \in L^{\infty}([0,1], \mathbb{C}^m)$ . For  $n \in \mathbb{N}$  let  $\Omega_n := \left[\frac{1}{n}, 1 - \frac{1}{n}\right] \subseteq [0,1]$  and define  $f_n : [0,1] \to \mathbb{C}^m$  by a linear truncation of f outside  $\Omega_n$ , i.e.,

$$f_n(x) := \begin{cases} nf\left(\frac{1}{n}\right)x, & x \in \left[0, \frac{1}{n}\right], \\ f(x), & x \in \left[\frac{1}{n}, 1 - \frac{1}{n}\right], \\ nf\left(1 - \frac{1}{n}\right)(1 - x), & x \in \left[1 - \frac{1}{n}, 1\right]. \end{cases}$$

Observe that  $f_n$  is Lipschitz for each  $n \in \mathbb{N}$  and hence  $f_n \in W^{1,\infty}([0,1], \mathbb{C}^m)$ . Moreover  $f_n(1) = f_n(0) = 0$  for each  $n \in \mathbb{N}$  implying that  $f_n(0) = \mathbb{B}^C f_n(0)$ , hence  $f_n \in D(A_C)$ . Furthermore one has that  $\sup_{n \in \mathbb{N}} ||f_n||_{\infty} \leq ||f||_{\infty} < \infty$  and  $f_n \to f$  as  $n \to \infty$  with respect to the weak\*-topology, since

$$\left| \int_{0}^{1} \left\langle \left( f_{n}(x) - f(x) \right), g(x) \right\rangle \, \mathrm{d}x \right| \leq 2 \, \|f\|_{\infty} \, \lambda^{1} \left( \left[ 0, \frac{1}{n} \right] \cup \left[ 1 - \frac{1}{n}, 1 \right] \right) \|g\|_{1} = \frac{4}{n} \, \|f\|_{\infty} \, \|g\|_{1}$$

for each  $g \in L^1([0,1], \mathbb{C}^m)$ . We now define a sequence of operators  $A_n$  approximating  $A_C$ in the following way. For each  $c_j \in \mathbb{R}$  there exists a sequence  $(c_j^{(n)})_{n \in \mathbb{N}}$  in  $\mathbb{Q}$  such that  $\lim_{n\to\infty} c_j^{(n)} = c_j$ . Since the network is finite, for each  $n \in \mathbb{N}$  the velocities  $c_j^{(n)}, j \in J$ , satisfy condition (7.2.8) and, by Proposition 7.12 we obtain a bi-continuous contraction semigroup  $(T_n(t))_{t\geq 0}$  generated by

$$A_n := \operatorname{diag}\left(c_j^{(n)} \cdot \frac{\mathrm{d}}{\mathrm{d}x}\right),$$
  
$$D(A_n) := \left\{ f \in \mathrm{W}^{1,\infty}\left([0,1], \mathbb{C}^m\right) \mid f(1) = \mathbb{B}^{C_n} f(0) \right\},$$
  
(7.2.11)

where  $C_n := \operatorname{diag}\left(c_j^{(n)}\right)$ . Moreover, all semigroups  $(T_n(t))_{t\geq 0}$ ,  $n \in \mathbb{N}$ , are similar and thus uniformly bi-continuous of type 0, cf. Definition 1.20. Observe, that the general assumptions of Theorem 1.22 are satisfied. Let us now check the assumptions of assertion (b) of Theorem 1.22. Let  $R := R(\lambda, A_C)$  and observe that  $R : L^{\infty}([0,1], \mathbb{C}^m) \to D(A_C)$  is a bijection. By the above,  $\operatorname{Ran}(R)$  is bi-dense in  $L^{\infty}([0,1], \mathbb{C}^m)$ . For every  $n \in \mathbb{N}$ , replacing  $c_j$  by  $c_j^{(n)}$  for all  $j \in J$ , Lemma 7.14 yields an explicit expression for  $R(\lambda, A_n)$ . It is easy to see that  $R(\lambda, A_n)f \xrightarrow{\|\cdot\|} Rf$  for  $f \in D(A_C)$  as  $n \to \infty$ . Applying Theorem 1.22 gives us a bi-continuous semigroup  $(T_C(t))_{t\geq 0}$  and an operator (B, D(B)) generating this semigroup. Note that, since in our case  $R = R(\lambda, A_C)$  is a resolvent, by Remark 1.23 we have  $R = R(\lambda, A_C) = R(\lambda, B)$ for  $\lambda \in \rho(A_C)$  and by the uniqueness of the Laplace transform we conclude that (B, D(B)) = $(A_C, D(A_C))$ .

**Corollary 7.16.** The flow problem (7.2.2) for finite networks is well-posed on  $L^{\infty}([0,1], \mathbb{C}^m)$ .

**Remark 7.17.** Observe that in the same manner, by using the original strongly continuous version of the Trotter–Kato Theorem (see [52, Chapter III, Sect.4b]), one can deduce the well-posedness of the problem on  $X_0 = L^1([0, 1], \mathbb{C}^m)$ .

## § 7.3 Notes

The content of this chapter is joint work with M. Kramar Fijavž [29]. In fact, we treat the case of the state space  $L^{\infty}([0,1], \ell^1)$  which gives rise to bi-continuous semigroups. We chose this state space since on the one hand one has the Radon–Nikodym property of  $\ell^1$  and on the other hand a preadjoint semigroup which can be controlled to obtain our results. To mention is that some of the results become achievable after a fruitful discussion with B. Farkas.

The research question actually appeared as a part of a suggested research proposal for a *PPP project* of the *DAAD* driven forth by B. Farkas and M. Kramar Fijavž. Although this project was not supported we started a collaboration. To be more precise, in April 2018 the author visited the *University of Ljubljana* and was financially supported by the *ERASMUS+* program in the context of a staff training (STT).

M. Kramar Fijavž and the author first met during the final workshop of the  $20^{\text{th}}$  Internetseminar on Linear Parabolic Equations at the University of Salerno. As a matter of fact the author was part of Project F: Non-autonomous diffusion in networks conducted by Kramar Fijavž. This project awaken interest of the author in networks. Afterwards we stayed in contact and this get our collaboration off the ground.

We remark that there is an open question: can Theorem 7.15 be handled without the assumption of finiteness of the network as mentioned in Section 7.2.3? In fact, this assumption is caused by the use of the Trotter–Kato approximation theorem. In [21, Sect. 18.3] the strongly continuous case for finite networks on the phase space  $L^1([0,1], \mathbb{C}^m)$  is discussed. The author thinks that it should be no problem to generalize this to infinite networks. As a matter of fact, one uses the Lumer–Phillips generation theorem, cf. [21, Thm. 11.10] and [52, Chapter II, Thm. 3.15]. Such a generation theorem is not known so far for the bi-continuous case. Even if we also have the work of B. Dorn [41], the state spaces  $L^{\infty}([0,1], \ell^{\infty})$  and  $L^1([0,1], \ell^{\infty})$  are not yet considered and are left open.

# Bibliography

- [1] P. Acquistapace. Evolution operators and strong solutions of abstract linear parabolic equations. *Differential Integral Equations*, 1(4):433–457, 1988.
- [2] P. Acquistapace and B. Terreni. A unified approach to abstract linear nonautonomous parabolic equations. *Rend. Sem. Mat. Univ. Padova*, 78:47–107, 1987.
- [3] M. Adler, M. Bombieri, and K.-J. Engel. On perturbations of generators of C<sub>0</sub>semigroups. Abstr. Appl. Anal., pages Art. ID 213020, 13, 2014.
- [4] A. A. Albanese and E. Mangino. Trotter-Kato theorems for bi-continuous semigroups and applications to Feller semigroups. *Journal of Mathematical Analysis and Applications*, 289(2):477 – 492, 2004.
- [5] J. Alber. Implementierte Halbgruppen. Diplomarbeit, Eberhard-Karls-Universität Tübingen, 1999.
- [6] J. Alber. On implemented semigroups. Semigroup Forum, 63(3):371–386, Oct 2001.
- [7] H. Amann. Parabolic evolution equations in interpolation and extrapolation spaces. J. Funct. Anal., 78(2):233-270, 1988.
- [8] W. Arendt. Resolvent positive operators. Proc. London Math. Soc. (3), 54(2):321–349, 1987.
- [9] W. Arendt, O. El-Mennaoui, and V. Kéyantuo. Local integrated semigroups: evolution with jumps of regularity. J. Math. Anal. Appl., 186(2):572–595, 1994.
- [10] W. Arendt, A. Grabosch, G. Greiner, U. Groh, H. P. Lotz, U. Moustakas, R. Nagel, F. Neubrander, and U. Schlotterbeck. *One-parameter semigroups of positive operators*, volume 1184 of *Lecture Notes in Mathematics*. Springer-Verlag, Berlin, 1986.
- [11] W. Arendt, A. Grabosch, G. Greiner, U. Groh, H. P. Lotz, U. Moustakas, R. Nagel, F. Neubrander, and U. Schlotterbeck. *One-parameter semigroups of positive operators*, volume 1184 of *Lecture Notes in Mathematics*. Springer-Verlag, Berlin, 1986.
- [12] W. Arendt, F. Räbiger, and A. Sourour. Spectral properties of the operator equation AX + XB = Y. Quart. J. Math. Oxford Ser. (2), 45(178):133–149, 1994.
- [13] W. Arendt and A. Rhandi. Perturbation of positive semigroups. Arch. Math. (Basel), 56(2):107–119, 1991.

- [14] J. Banasiak and L. Arlotti. Perturbations of Positive Semigroups with Applications. Springer Monographs in Mathematics. Springer London, 2006.
- [15] J. Banasiak and A. Falkiewicz. Some transport and diffusion processes on networks and their graph realizability. *Appl. Math. Lett.*, 45:25–30, 2015.
- [16] J. Banasiak and A. Falkiewicz. A singular limit for an age structured mutation problem. Math. Biosci. Eng., 14(1):17–30, 2017.
- [17] J. Banasiak and P. Namayanja. Asymptotic behaviour of flows on reducible networks. *Netw. Heterog. Media*, 9(2):197–216, 2014.
- [18] J. Banasiak and A. Puchalska. Generalized network transport and Euler-Hille formula. Discrete Contin. Dyn. Syst., Ser. B, 23(5):1873–1893, 2018.
- [19] M. Baroun and B. Jacob. Admissibility and observability of observation operators for semilinear problems. *Integral Equations Operator Theory*, 64(1):1–20, 2009.
- [20] A. Bátkai, B. Jacob, J. Voigt, and J. Wintermayr. Perturbations of positive semigroups on AM-spaces. Semigroup Forum, 96(2):333–347, 2018.
- [21] A. Bátkai, M. Kramar Fijavž, and A. Rhandi. Positive operator semigroups. From finite to infinite dimensions., volume 257. Basel: Birkhäuser/Springer, 2017.
- [22] F. Bayazit, B. Dorn, and M. Kramar Fijavž. Asymptotic periodicity of flows in timedepending networks. *Netw. Heterog. Media*, 8(4):843–855, 2013.
- [23] V. Bogachev. Measure Theory. Number Vol. 1 in Measure Theory. Springer Berlin Heidelberg, 2007.
- [24] V. Bogachev. *Differentiable Measures and the Malliavin Calculus*. Mathematical surveys and monographs. American Mathematical Soc., 2010.
- [25] V. Bogachev. Gaussian Measures. Mathematical Surveys and Monographs. American Mathematical Society, 2015.
- [26] O. Bratteli and D. W. Robinson. Operator algebras and quantum statistical mechanics. Vol. 1. Springer-Verlag, New York-Heidelberg, 1979. C\*- and W\*-algebras, algebras, symmetry groups, decomposition of states, Texts and Monographs in Physics.
- [27] C. Budde and B. Farkas. A Desch–Schappacher perturbation theorem for bi-continuous semigroups. to appear in *Mathematische Nachrichten*, ArXiv:1811.08455, Nov. 2018.
- [28] C. Budde and B. Farkas. Intermediate and extrapolated spaces for bi-continuous operator semigroups. *Journal of Evolution Equations*, 19(2):321–359, Jun 2019.
- [29] C. Budde and M. Kramar Fijavž. Bi-continuous semigroups for flows in infinite networks. ArXiv:1901.10292, Jan. 2019.
- [30] P. Busch and S. P. Gudder. Effects as functions on projective Hilbert space. Lett. Math. Phys., 47(4):329–337, 1999.

- [31] V. Casarino, K.-J. Engel, R. Nagel, and G. Nickel. A semigroup approach to boundary feedback systems. *Integral Equations Operator Theory*, 47(3):289–306, 2003.
- [32] S. Cerrai. A Hille-Yosida theorem for weakly continuous semigroups. Semigroup Forum, 49(3):349–367, 1994.
- [33] J. B. Conway. A course in abstract analysis, volume 141 of Graduate Studies in Mathematics. American Mathematical Society, Providence, RI, 2012.
- [34] J. B. Cooper. Saks spaces and applications to functional analysis, volume 139 of North-Holland Mathematics Studies. North-Holland Publishing Co., Amsterdam, second edition, 1987. Notas de Matemática [Mathematical Notes], 116.
- [35] R. Cristescu. Vector seminorms, spaces with vector norm, and regular operators. *Rev. Roumaine Math. Pures Appl.*, 53(5-6):407–418, 2008.
- [36] G. Da Prato and P. Grisvard. On extrapolation spaces. Atti Accad. Naz. Lincei Rend. Cl. Sci. Fis. Mat. Natur. (8), 72(6):330–332 (1983), 1982.
- [37] G. Da Prato and P. Grisvard. Maximal regularity for evolution equations by interpolation and extrapolation. *Journal of Functional Analysis*, 58(2):107 – 124, 1984.
- [38] W. Desch and W. Schappacher. On relatively bounded perturbations of linear C<sub>0</sub>semigroups. Ann. Sc. Norm. Super. Pisa, Cl. Sci., IV. Ser., 11:327–341, 1984.
- [39] W. Desch and W. Schappacher. Some perturbation results for analytic semigroups. Mathematische Annalen, 281(1):157–162, 1988.
- [40] B. Dorn. Semigroups for flows on infinite networks. Diplomarbeit, Eberhard-Karls-Universität Tübingen, 2005.
- [41] B. Dorn. Semigroups for flows in infinite networks. Semigroup Forum, 76(2):341–356, Mar 2008.
- [42] B. Dorn, V. Keicher, and E. Sikolya. Asymptotic periodicity of recurrent flows in infinite networks. *Math. Z.*, 263(1):69–87, 2009.
- [43] B. Dorn, M. Kramar Fijavž, R. Nagel, and A. Radl. The semigroup approach to transport processes in networks. *Physica D: Nonlinear Phenomena*, 239(15):1416 – 1421, 2010. Evolution Equations in Pure and Applied Sciences.
- [44] N. Dunford and J. Schwartz. *Linear Operators: General theory*. Pure and applied mathematics. Interscience Publishers, 1958.
- [45] T. Eisner. Stability of operators and operator semigroups, volume 209 of Operator Theory: Advances and Applications. Birkhäuser Verlag, Basel, 2010.
- [46] K.-J. Engel. Spectral theory and generator property for one-sided coupled operator matrices. Semigroup Forum, 58(2):267–295, 1999.
- [47] K.-J. Engel and M. Kramar Fijavž. Exact and positive controllability of boundary control systems. *Netw. Heterog. Media*, 12(2):319–337, 2017.

- [48] K.-J. Engel and M. Kramar Fijavž. Waves and diffusion on metric graphs with general vertex conditions. Evol. Equ. Control Theory, 8(3):633–661, 2019.
- [49] K.-J. Engel, M. Kramar Fijavž, B. Klöss, R. Nagel, and E. Sikolya. Maximal controllability for boundary control problems. *Appl. Math. Optim.*, 62(2):205–227, 2010.
- [50] K.-J. Engel, M. Kramar Fijavž, R. Nagel, and E. Sikolya. Vertex control of flows in networks. *Netw. Heterog. Media*, 3(4):709–722, 2008.
- [51] K.-J. Engel, M. Kramar Fijavž, R. Nagel, and E. Sikolya. Vertex control of flows in networks. *Networks & Heterogeneous Media*, 3:709, 2008.
- [52] K.-J. Engel and R. Nagel. One-parameter semigroups for linear evolution equations, volume 194 of Graduate Texts in Mathematics. Springer-Verlag, New York, 2000. With contributions by S. Brendle, M. Campiti, T. Hahn, G. Metafune, G. Nickel, D. Pallara, C. Perazzoli, A. Rhandi, S. Romanelli and R. Schnaubelt.
- [53] J. Esterle. Mittag-Leffler methods in the theory of Banach algebras and a new approach to Michael's problem. In Proceedings of the conference on Banach algebras and several complex variables (New Haven, Conn., 1983), volume 32 of Contemp. Math., pages 107–129. Amer. Math. Soc., Providence, RI, 1984.
- [54] B. Farkas. Perturbations of Bi-Continuous Semigroups. PhD thesis, Eötvös Loránd University, 2003.
- [55] B. Farkas. Perturbations of bi-continuous semigroups. Studia Math., 161(2):147–161, 2004.
- [56] B. Farkas. Perturbations of bi-continuous semigroups with applications to transition semigroups on  $C_b(H)$ . Semigroup Forum, 68(1):87–107, 2004.
- [57] B. Farkas. Adjoint bi-continuous semigroups and semigroups on the space of measures. *Czechoslovak Mathematical Journal*, 61(2):309–322, Jun 2011.
- [58] S. V. Fomin. Differentiable measures in linear spaces. Uspehi Mat. Nauk, 23(1 (139)):221-222, 1968.
- [59] T. Graser. Operator multipliers generating strongly continuous semigroups. Semigroup Forum, 55(1):68, 1997.
- [60] G. Greiner. Perturbing the boundary conditions of a generator. *Houston J. Math.*, 13(2):213–229, 1987.
- [61] U. Groh and F. Neubrander. Stabilität starkstetiger, positiver Operatorhalbgruppen auf C\*-Algebren. Math. Ann., 256(4):509–516, 1981.
- [62] M. Haase. Operator-valued H<sup>∞</sup>-calculus in inter- and extrapolation spaces. Integral Equations and Operator Theory, 56(2):197–228, Oct 2006.
- [63] S. Hadd, R. Manzo, and A. Rhandi. Unbounded perturbations of the generator domain. Discrete Contin. Dyn. Syst., 35(2):703–723, 2015.

- [64] S. Hadd, R. Manzo, and A. Rhandi. Unbounded perturbations of the generator domain. Discrete Contin. Dyn. Syst., 35(2):703–723, 2015.
- [65] R. Heymann. Multiplication operators on Bochner spaces and Banach fibre spaces. PhD thesis, Eberhard-Karls-Universität Tübingen, 2015.
- [66] E. Hille. Functional Analysis and Semi-Groups. American Mathematical Society Colloquium Publications, vol. 31. American Mathematical Society, New York, 1948.
- [67] B. Jacob, R. Nabiullin, J. Partington, and F. Schwenninger. On input-to-state-stability and integral input-to-state-stability for parabolic boundary control systems. In 2016 IEEE 55th Conference on Decision and Control (CDC), pages 2265–2269, Dec 2016.
- [68] B. Jacob, R. Nabiullin, J. R. Partington, and F. L. Schwenninger. Infinite-dimensional input-to-state stability and Orlicz spaces. SIAM J. Control Optim., 56(2):868–889, 2018.
- [69] B. Jacob, F. L. Schwenninger, and H. Zwart. On continuity of solutions for parabolic control systems and input-to-state stability. *Journal of Differential Equations*, page 23 pages, 2018.
- [70] B. Jacob, S.-A. Wegner, and J. Wintermayr. Desch-Schappacher perturbation of oneparameter semigroups on locally convex spaces. *Math. Nachr.*, 288(8-9):925–934, 2015.
- [71] T. Kato. Remarks on pseudo-resolvents and infinitesimal generators of semi-groups. Proc. Japan Acad., 35(8):467–468, 1959.
- [72] T. Kato. Abstract evolution equations of parabolic type in Banach and Hilbert spaces. Nagoya Math. J., 19:93–125, 1961.
- [73] T. Kato and H. Tanabe. On the abstract evolution equation. Osaka Math. J., 14:107– 133, 1962.
- [74] J. Kisyński. Sur les opérateurs de Green des problèmes de Cauchy abstraits. Studia Mathematica, 23(3):285–328, 1964.
- [75] R. Kraaij. Strongly continuous and locally equi-continuous semigroups on locally convex spaces. Semigroup Forum, 92(1):158–185, Feb 2016.
- [76] M. Kramar and E. Sikolya. Spectral properties and asymptotic periodicity of flows in networks. *Mathematische Zeitschrift*, 249(1):139–162, Jan 2005.
- [77] M. Kramar Fijavž, D. Mugnolo, and E. Sikolya. Variational and semigroup methods for waves and diffusion in networks. *Applied Mathematics and Optimization*, 55(2):219–240, Mar 2007.
- [78] F. Kühnemund. Bi-Continuous Semigroups on Spaces with Two Topologies: Theory and Applications. PhD thesis, Eberhard-Karls-Universität Tübingen, 2001.
- [79] F. Kühnemund. A Hille-Yosida theorem for bi-continuous semigroups. Semigroup Forum, 67(2):205–225, 2003.

- [80] M. Kunze. Continuity and equicontinuity of semigroups on norming dual pairs. Semigroup Forum, 79(3):540, Aug 2009.
- [81] K. Landsman. Foundations of quantum theory, volume 188 of Fundamental Theories of Physics. Springer, Cham, 2017. From classical concepts to operator algebras.
- [82] L. Lorenzi and M. Bertoldi. Analytical methods for Markov semigroups, volume 283 of Pure and Applied Mathematics (Boca Raton). Chapman & Hall/CRC, Boca Raton, FL, 2007.
- [83] H. P. Lotz. Uniform convergence of operators on  $L^{\infty}$  and similar spaces. Math. Z., 190(2):207–220, 1985.
- [84] G. Lumer and L. Weis, editors. Evolution equations and their applications in physical and life sciences, volume 215 of Lecture Notes in Pure and Applied Mathematics. Marcel Dekker, Inc., New York, 2001.
- [85] A. Lunardi. Interpolation theory. Appunti. Scuola Normale Superiore di Pisa (Nuova Serie). [Lecture Notes. Scuola Normale Superiore di Pisa (New Series)]. Edizioni della Normale, Pisa, second edition, 2009.
- [86] P. Magal and S. Ruan. Theory and applications of abstract semilinear Cauchy problems, volume 201 of Applied Mathematical Sciences. Springer, Cham, 2018. With a foreword by Glenn Webb.
- [87] I. V. Melnikova and A. Filinkov. Abstract Cauchy problems: three approaches, volume 120 of Chapman & Hall/CRC Monographs and Surveys in Pure and Applied Mathematics. Chapman & Hall/CRC, Boca Raton, FL, 2001.
- [88] P. Meyer-Nieberg. Banach lattices. Berlin etc.: Springer-Verlag, 1991.
- [89] I. Miyadera. On perturbation theory for semi-groups of operators. Tôhoku Math. J. (2), 18:299–310, 1966.
- [90] G. Murphy. C<sup>\*</sup>-algebras and Operator Theory. Academic Press, 1990.
- [91] R. Nagel. Sobolev spaces and semigroups. Semesterbericht Funktionalanalysis, pages 1–19, 1983.
- [92] R. Nagel. Extrapolation spaces for semigroups. RIMS Kôkyûroku, 1009:181–191, 1997.
- [93] R. Nagel, G. Nickel, and S. Romanelli. Identification of extrapolation spaces for unbounded operators. *Quaest. Math.*, 19(1-2):83–100, 1996.
- [94] R. Nagel and E. Sinestrari. Inhomogeneous Volterra integrodifferential equations for Hille-Yosida operators. In *Functional analysis (Essen, 1991)*, volume 150 of *Lecture Notes in Pure and Appl. Math.*, pages 51–70. Dekker, New York, 1994.
- [95] R. Nagel and E. Sinestrari. Extrapolation spaces and minimal regularity for evolution equations. J. Evol. Equ., 6(2):287–303, 2006.
- [96] G. Nickel. A new look at boundary perturbations of generators. Electron. J. Differential Equations, pages No. 95, 14, 2004.

- [97] V. u. Q. Phóng. The operator equation AX XB = C with unbounded operators A and B and related abstract Cauchy problems. Math. Z., 208(4):567–588, 1991.
- [98] V. u. Q. Phóng. On the exponential stability and dichotomy of C<sub>0</sub>-semigroups. Studia Math., 132(2):141–149, 1999.
- [99] V. u. Q. Phóng and E. Schüler. The operator equation AX XB = C, admissibility, and asymptotic behavior of differential equations. Journal of Differential Equations, 145:394–419, 05 1998.
- [100] E. Priola. On a class of Markov type semigroups in spaces of uniformly continuous and bounded functions. *Studia Math.*, 136(3):271–295, 1999.
- [101] W. Rudin. Functional analysis. International Series in Pure and Applied Mathematics. McGraw-Hill, Inc., New York, second edition, 1991.
- [102] H. H. Schaefer. Banach lattices and positive operators. Springer-Verlag, New York-Heidelberg, 1974. Die Grundlehren der mathematischen Wissenschaften, Band 215.
- [103] A. K. Scirrat. Evolution Semigroups for Well-Posed, NonAutonomous Evolution Families. PhD thesis, Louisiana State University and Agricultural and Mechanical College, 2016.
- [104] E. Sinestrari. Interpolation and extrapolation spaces in evolution equations. In J. Cea, D. Chenais, G. Geymonat, and J. L. Lions, editors, *Partial Differential Equations and Functional Analysis: In Memory of Pierre Grisvard*, pages 235–254. Birkhäuser Boston, Boston, MA, 1996.
- [105] A. V. Skorohod. On the differentiability of measures which correspond to stochastic processes. I. Processes with independent increments. *Teor. Veroyatnost. i Primenen*, 2:417–443, 1957.
- [106] A. V. Skorohod. Integration in Hilbert space. Springer-Verlag, New York-Heidelberg, 1974. Translated from the Russian by Kenneth Wickwire, Ergebnisse der Mathematik und ihrer Grenzgebiete, Band 79.
- [107] O. Staffans. Well-posed linear systems, volume 103 of Encyclopedia of Mathematics and its Applications. Cambridge University Press, Cambridge, 2005.
- [108] O. J. Staffans and G. Weiss. Transfer functions of regular linear systems. III. Inversions and duality. *Integral Equations Operator Theory*, 49(4):517–558, 2004.
- [109] O. J. Staffans and G. Weiss. Transfer functions of regular linear systems part III: Inversions and duality. *Integral Equations and Operator Theory*, 49(4):517–558, Aug 2004.
- [110] V. N. Sudakov. Linear sets with quasi-invariant measure. Dokl. Akad. Nauk SSSR, 127:524–525, 1959.
- [111] H. Tanabe. On the equations of evolution in a Banach space. Osaka Math. J., 12:363– 376, 1960.

- [112] H. Tanabe. Evolutional equations of parabolic type. Proc. Japan Acad., 37:610–613, 1961.
- [113] S. Thomaschewski. From methods for autonomous and non-autonomous Cauchy problems. PhD thesis, Universität Ulm, 2003.
- [114] H. F. Trotter. Approximation of semi-groups of operators. Pacific J. Math., 8(4):887– 919, 1958.
- [115] Y. Umemura. Measures on infinite dimensional vector spaces. Publ. Res. Inst. Math. Sci. Ser. A, 1:1–47, 1965.
- [116] J. van Neerven. The adjoint of a semigroup of linear operators, volume 1529 of Lecture Notes in Mathematics. Springer-Verlag, Berlin, 1992.
- [117] W. van Zuijlen. Integration of functions with values in a Riesz space. Master's thesis, Radboud Universiteit Nijmegen, 2012.
- [118] J. Voigt. On the perturbation theory for strongly continuous semigroups. Math. Ann., 229(2):163–171, 1977.
- [119] J. Voigt. On resolvent positive operators and positive  $C_0$ -semigroups on AL-spaces. Semigroup forum, 38(2):263-266, 1989.
- [120] J. von Below and J. A. Lubary. The eigenvalues of the Laplacian on locally finite networks. *Results Math.*, 47(3-4):199–225, 2005.
- [121] J. von Below and J. A. Lubary. The eigenvalues of the Laplacian on locally finite networks under generalized node transition. *Results Math.*, 54(1-2):15–39, 2009.
- [122] T. Walter. Störungstheorie von Generatoren und Favardklassen. Semesterbericht Funktionalanalysis, pages 207–220, 1986.
- [123] S.-A. Wegner. Universal extrapolation spaces for  $C_0$ -semigroups. Annali dell 'Universita' di Ferrara, 60(2):447–463, Nov 2014.
- [124] K. Yosida. On the differentiability and the representation of one-parameter semi-group of linear operators. J. Math. Soc. Japan, 1:15–21, 1948.
- [125] A. C. Zaanen. Archimedean Spaces and Convergence, pages 39–53. Springer Berlin Heidelberg, Berlin, Heidelberg, 1997.